

This is a repository copy of Determining the optimal model for role substitution in NHS dental services in the UK: a mixed-methods study.

White Rose Research Online URL for this paper: https://eprints.whiterose.ac.uk/188467/

Version: Published Version

#### Article:

Brocklehurst, P.B., Birch, S.B., McDonald, R.M. et al. (4 more authors) (2016) Determining the optimal model for role substitution in NHS dental services in the UK: a mixed-methods study. Health Services and Delivery Research, 4 (22). ISSN 2050-4349

https://doi.org/10.3310/hsdr04220

© Queen's Printer and Controller of HMSO 2016. This work was produced by Brocklehurst et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

#### Reuse

Items deposited in White Rose Research Online are protected by copyright, with all rights reserved unless indicated otherwise. They may be downloaded and/or printed for private study, or other acts as permitted by national copyright laws. The publisher or other rights holders may allow further reproduction and re-use of the full text version. This is indicated by the licence information on the White Rose Research Online record for the item.

#### **Takedown**

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



### **HEALTH SERVICES AND DELIVERY RESEARCH**

VOLUME 4 ISSUE 22 JULY 2016 ISSN 2050-4349

Determining the optimal model for role substitution in NHS dental services in the UK: a mixed-methods study

Paul Brocklehurst, Stephen Birch, Ruth McDonald, Harry Hill, Lucy O'Malley, Richard Macey and Martin Tickle



# Determining the optimal model for role substitution in NHS dental services in the UK: a mixed-methods study

Paul Brocklehurst, 1\* Stephen Birch, 2 Ruth McDonald, 3 Harry Hill, 4 Lucy O'Malley, 4 Richard Macey 4 and Martin Tickle 4

<sup>1</sup>School of Healthcare Sciences, Bangor University, Bangor, UK
<sup>2</sup>Faculty of Health Sciences, McMaster University, Hamilton, ON, Canada
<sup>3</sup>Manchester Business School, University of Manchester, Manchester, UK
<sup>4</sup>School of Dentistry, University of Manchester, Manchester, UK

Declared competing interests of authors: none

Published July 2016 DOI: 10.3310/hsdr04220

This report should be referenced as follows:

Brocklehurst P, Birch S, McDonald R, Hill H, O'Malley L, Macey R, *et al.* Determining the optimal model for role substitution in NHS dental services in the UK: a mixed-methods study. *Health Serv Deliv Res* 2016;**4**(22).

<sup>\*</sup>Corresponding author

### **Health Services and Delivery Research**

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: nihredit@southampton.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme or programmes which preceded the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **HS&DR** programme

The Health Services and Delivery Research (HS&DR) programme, part of the National Institute for Health Research (NIHR), was established to fund a broad range of research. It combines the strengths and contributions of two previous NIHR research programmes: the Health Services Research (HSR) programme and the Service Delivery and Organisation (SDO) programme, which were merged in January 2012.

The HS&DR programme aims to produce rigorous and relevant evidence on the quality, access and organisation of health services including costs and outcomes, as well as research on implementation. The programme will enhance the strategic focus on research that matters to the NHS and is keen to support ambitious evaluative research to improve health services.

For more information about the HS&DR programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hsdr

#### This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 11/1025/04. The contractual start date was in February 2013. The final report began editorial review in August 2015 and was accepted for publication in January 2016. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health.

© Queen's Printer and Controller of HMSO 2016. This work was produced by Brocklehurst et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### Health Services and Delivery Research Editor-in-Chief

Professor Jo Rycroft-Malone Professor of Health Services and Implementation Research, Bangor University, UK

### **NIHR Journals Library Editor-in-Chief**

Professor Tom Walley Director, NIHR Evaluation, Trials and Studies and Director of the EME Programme, UK

#### **NIHR Journals Library Editors**

**Professor Ken Stein** Chair of HTA Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andree Le May Chair of NIHR Journals Library Editorial Group (EME, HS&DR, PGfAR, PHR journals)

Dr Martin Ashton-Key Consultant in Public Health Medicine/Consultant Advisor, NETSCC, UK

**Professor Matthias Beck** Chair in Public Sector Management and Subject Leader (Management Group), Queen's University Management School, Queen's University Belfast, UK

**Professor Aileen Clarke** Professor of Public Health and Health Services Research, Warwick Medical School, University of Warwick, UK

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

**Professor Elaine McColl** Director, Newcastle Clinical Trials Unit, Institute of Health and Society, Newcastle University, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

**Professor Geoffrey Meads** Professor of Health Sciences Research, Health and Wellbeing Research and Development Group, University of Winchester, UK

Professor John Norrie Health Services Research Unit, University of Aberdeen, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

**Professor Martin Underwood** Director, Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of members of the NIHR Journals Library Board: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: nihredit@southampton.ac.uk

### **Abstract**

### Determining the optimal model for role substitution in NHS dental services in the UK: a mixed-methods study

Paul Brocklehurst,<sup>1</sup>\* Stephen Birch,<sup>2</sup> Ruth McDonald,<sup>3</sup> Harry Hill,<sup>4</sup> Lucy O'Malley,<sup>4</sup> Richard Macey<sup>4</sup> and Martin Tickle<sup>4</sup>

<sup>1</sup>School of Healthcare Sciences, Bangor University, Bangor, UK

**Background:** Maximising health gain for a given level and mix of resources is an ethical imperative for health-service planners. Approximately half of all patients who attend a regular NHS dental check-up do not require any further treatment, whereas many in the population do not regularly attend. Thus, the most expensive resource (the dentist) is seeing healthy patients at a time when many of those with disease do not access care. Role substitution in NHS dentistry, where other members of the dental team undertake the clinical tasks previously provided by dentists, has the potential to increase efficiency and the capacity to care and lower costs. However, no studies have empirically investigated the efficiency of NHS dental provision that makes use of role substitution.

Research questions: This programme of research sought to address three research questions: (1) what is the efficiency of NHS dental teams that make use of role substitution?; (2) what are the barriers to, and facilitators of, role substitution in NHS dental practices?; and (3) how do incentives in the remuneration systems influence the organisation of these inputs and production of outputs in the NHS?

**Design:** Data envelopment analysis was used to develop a productive efficiency frontier for participating NHS practices, which were then compared on a relative basis, after controlling for patient and practice characteristics. External validity was tested using stochastic frontier modelling, while semistructured interviews explored the views of participating dental teams and their patients to role substitution.

Setting: NHS 'high-street' general dental practices.

Participants: 121 practices across the north of England.

**Interventions:** No active interventions were undertaken.

**Main outcome measures:** Relative efficiency of participating NHS practices, alongside a detailed narrative of their views about role substitution dentistry. Social acceptability for patients.

<sup>&</sup>lt;sup>2</sup>Faculty of Health Sciences, McMaster University, Hamilton, ON, Canada

<sup>&</sup>lt;sup>3</sup>Manchester Business School, University of Manchester, Manchester, UK

<sup>&</sup>lt;sup>4</sup>School of Dentistry, University of Manchester, Manchester, UK

<sup>\*</sup>Corresponding author p.brocklehurst@bangor.ac.uk

Results: The utilisation of non-dentist roles in NHS practices was relatively low, the most common role type being the dental hygienist. Increasing the number of non-dentist team members reduced efficiency. However, it was not possible to determine the relative efficiency of individual team members, as the NHS contracts only with dentists. Financial incentives in the NHS dental contract and the views of practice principals (i.e. senior staff members) were equally important. Bespoke payment and referral systems were required to make role substitution economically viable. Many non-dentist team members were not being used to their full scope of practice and constraints on their ability to prescribe reduced efficiency further. Many non-dentist team members experienced a precarious existence, commonly being employed at multiple practices. Patients had a low level of awareness of the different non-dentist roles in a dental team. Many exhibited an inherent trust in the professional 'system', but prior experience of role substitution was important for social acceptability.

**Conclusions:** Better alignment between the financial incentives within the NHS dental contract and the use of role substitution is required, although professional acceptability remains critical.

**Study limitations:** Output data collected did not reflect the quality of care provided by the dental team and the input data were self-reported.

**Future work:** Further work is required to improve the evidence base for the use of role substitution in NHS dentistry, exploring the effects and costs of provision.

Funding: The National Institute for Health Research Health Services and Delivery Research programme.

### **Contents**

LIST OT TABLES	XI
List of figures	xv
Glossary	xvii
List of abbreviations	xix
Plain English summary	xxi
Scientific summary	xxiii
Chapter 1 Overview of the report	1
Outline of the report	1
Introduction	1
Aims and objectives	2
Research design and project overview	2
Structure of the report	3
Chapter 2 Organisation of NHS dentistry in primary care	5
Introduction	5
Organisation of NHS dentistry in primary care in England and Wales	5
Organisation of NHS dentistry in primary care in Northern Ireland and Scotland	7
Role substitution in NHS dentistry in primary care	8
Financial incentives in NHS dentistry in primary care	10
Concluding remarks	11
Chapter 3 Preliminary screening of NHS dental care professionals	13
Introduction	13
Methods	13
Results	14
Discussion	16
Concluding remarks	16
Chapter 4 Technical efficiency of NHS teams that use role substitution	17
Introduction	17
Aims and objectives	18
Methods	18
Data collection	18
Data envelopment analysis and the analysis of efficiency scores	21
Sensitivity analysis	22
Stochastic frontier model approach to estimating efficiency	22
Sample size	23

Results	23
Sample of NHS dental practices in England	23
Data envelopment analysis	24
Sensitivity checks on the estimation of efficiency scores	29
Stochastic frontier modelling	30
Impact of different remuneration systems on the technical efficiency of role	
substitution models	30
Discussion	31
Main findings	31
Study limitations	32
Concluding remarks	33
Chapter 5 Attitudes of patients and the NHS dental team to role substitution	35
Introduction	35
Methods	35
Recruitment	35
Interview structure and context	35
Data analysis	36
Results	36
Overview of sample	36
Results: practice principals	37
Results: dental care practitioners	42
Results: patients	51
Discussion	56
Concluding remarks	57
Chapter 6 The role of financial incentives on NHS dental activity	59
Introduction	59
Methods	60
Data collection and matching approach	60
Variable description	60
Analysis	61
Results	62
Descriptive statistics	62
Main findings	63
Patient selection	65
Receipt of care among registrants	65
Mix of treatments	65
Financial viability	65
Discussion	65
Concluding remarks	67
Chapter 7 Conclusions and recommendations	69
Considerations for future policy	71
Strengths and limitations of the research	7
Recommendations for further research	72
Concluding remarks	73
Acknowledgements	75
References	77

Appendix 1 Analysis of additional sites	87
Appendix 2 Dental team questionnaire	99
Appendix 3 Analysis of Oasis data	105
Appendix 4 Topic guide and coding frame	109
Appendix 5 Patient and public involvement	117

### **List of tables**

TABLE 1 Demographics of the DCP sample	14
TABLE 2 Common models of role substitution in sampled NHS practices in the UK	16
TABLE 3 Input and output variables for the oral health-care production	19
TABLE 4 Practice characteristics recorded (via questionnaire)	20
<b>TABLE 5</b> Patient population and service mix characteristics (via Business Services Authority)	20
TABLE 6 Sample practices in England compared with national data	24
TABLE 7 Descriptive statistics of practices that did and did not use DCPs	25
TABLE 8 Sample statistics of NHS dental health-care output measures	25
TABLE 9 Sample statistics of staffing variables	26
TABLE 10 Sample statistics of practice/site variables	26
TABLE 11 Mean practice efficiency score when health-care output is measured by UDAs, treatment plans and patients seen	27
TABLE 12 Association with efficiency scores of any use of DCPs with NHS patients for different health-care output measures	27
TABLE 13 Association with efficiency scores of the level of use of DCPs with NHS patients for different health-care output measures	28
TABLE 14 Associations with efficiency scores of the proportion of clinical time provided for different health-care output measures	28
TABLE 15 Mediating associations with efficiency scores between the level of DCP use and patient case-mix variables	29
TABLE 16 Associations between the numbers of routine services delivered in the NHS and the level of DCP use on NHS patients	29
TABLE 17 Alternative specifications of inputs into the productive efficiency frontier	30
TABLE 18 Stochastic frontier modelling approach to estimating efficiency scores	31
TABLE 19 Numbers of participants by type, country, mode of interview and recruitment	36
TABLE 20 Sample comparison of GDS control practices with the Oasis group for practice characteristics and patient selection variables	62

receipt of care among registrants, mix of treatments and financial viability variables	63
TABLE 22 Separate regressions showing the difference in the outcome variable between Oasis and comparison practices for each group of RQs	64
TABLE 23 Sample statistics of NHS dental health-care output measures	87
TABLE 24 Sample statistics of staffing measures	88
TABLE 25 Sample statistics of practice/site measures	89
TABLE 26 Summary of efficiency scores estimated by input orientation	89
TABLE 27 Summary of efficiency scores estimated by output orientation	89
TABLE 28 Association with efficiency scores of any use of DCPs with NHS patients when the health-care output was UDAs/gross NHS income	90
TABLE 29 Association with efficiency scores of any use of DCPs with NHS patients when the health-care output was treatment plans (claims)	91
TABLE 30 Association with efficiency scores of any use of DCPs with NHS patients when the health-care output was patients seen	91
TABLE 31 Associations with efficiency scores of the utilisation of DCPs with either NHS or private patients	92
TABLE 32 Association with efficiency scores of the level of use of DCPs with NHS patients when the health-care output was UDAs/gross NHS income	93
TABLE 33 Association with efficiency scores of the level of use of DCPs with NHS patients when the health-care output was treatment plans (claims)	94
TABLE 34 Association with efficiency scores of the level of use of DCPs with NHS patients when the health-care output was patients seen	94
TABLE 35 Association with efficiency scores of the proportion of clinical time provided by DCPs when the health-care output was UDAs/gross income	95
TABLE 36 Association with efficiency scores of the proportion of clinical time provided by DCPs when the health-care output was treatment courses (claims)	95
TABLE 37 Association with efficiency scores of the proportion of clinical time provided by DCPs when the health-care output was patients seen	96
TABLE 38 Mediating associations with efficiency scores between the level of DCP use and patient case-mix variables	97
TABLE 39 Associations between the numbers of routine services delivered in the NHS and the level of DCP use on NHS patients	98

TABLE 40 Associations between routine services delivered in the NHS and the proportion of DCPs' clinical time relative to GDPs' clinical time working with	00
NHS patients	98
TABLE 41 Sample comparison of Oasis practices with control practices for practice characteristics and patient selection variables	105
TABLE 42 Sample comparison of Oasis practices with control practices for receipt of care among registrants, mix of treatments and financial viability variables	106
TABLE 43 Separate regressions showing the differences in the outcome variables between Oasis and control practices for each group of RQs when estimated using OLS with robust standard errors, random effects and random effects with the	
Mundlak correction	108
TABLE 44 Coding frame: GDPs	112
TABLE 45 Coding frame: DCPs	113
TABLE 46 Coding frame: patients	115

### **List of figures**

FIGURE 1 Distribution of DHs, DTs and DHTs in NHS practices across the UK

15

### **Glossary**

**Data envelopment analysis and stochastic frontier modelling** The two principal methods in health economics of measuring technical efficiency.

**High-street dentists** Also known as general dental practitioners, the only clinicians who can contract directly with the NHS. In England, general dental practitioners are provided with a target for their clinical activity, known as the Annual Contract Value. Three bands of treatment activity exist in England (band 1, band 2 and band 3) to reflect differing degrees of treatment complexity. Band 1 relates to examinations and preventative treatments, while bands 2 and 3 relate to invasive and irreversible dental procedures. Band 1 attracts one Unit of Dental Activity, whereas bands 2 and 3 attract three and 12 Units of Dental Activity, respectively. The Annual Contract Value is the annual target of Units of Dental Activity that a dental team must provide.

**Non-dentist members of the dental team** Team members that can substitute the role of the general dental practitioner, such as the dental hygienist, the dental hygiene therapist and the dental therapist, collectively known as dental care professionals. Although, technically, dental nurses and dental technicians are also classed as dental care professionals, their activity supplements the role of the general dental practitioner. As such, the term dental care professional in this report refers to those non-dentist members of the dental team who can engage in role substitution.

### **List of abbreviations**

ACV BSDHT	Annual Contract Value  British Society of Dental Hygiene	NIHR	National Institute for Health Research
וווטכט	and Therapy	OLS	ordinary least squares
DCP	dental care professional	PCT	primary care trust
DEA	data envelopment analysis	PDS	personal dental service
DH	dental hygienist	PP	practice principal
DHT	dental hygiene therapist	PPI	patient and public involvement
DT	dental therapist	RESET	Ramsey regression equation
FFS	fee for service		specification error test
GDC	General Dental Council	RQ	research question
GDP	general dental practitioner	SFM	stochastic frontier modelling
GDS	General Dental Services	UDA	Unit of Dental Activity

### **Plain English summary**

mproving productivity in the NHS is a key consideration for policy-makers and is becoming increasingly important, given budgetary pressures on public expenditure. Across England, approximately half of the patients who attend for a regular NHS dental check-up do not require any further treatment, whereas half of the population does not regularly attend the dentist. In other words, the most expensive resource (the dentist) is seeing healthy patients at a time when many of those with the most dental health problems do not access care. One way to address this issue is to improve the productivity of the dental team. Role substitution, when non-dentist members of the dental team take on tasks previously limited to dentists, has the potential to do this. The aims of this study were to examine how productive dental teams are that make use of non-dentists and to explore the barriers to, and enablers of, change.

A multimethod approach was adopted using questionnaires, health-economic models specifically designed to measure productivity and interviews with dentists, their teams and patients. The extent of role substitution in NHS dentistry was limited; the most common non-dentist team member was the dental hygienist. As the number of non-dentist team members increased, the productivity of the NHS practice dropped. Financial incentives played a major role in this, but the views of practice principals were also important. Patients had a low level of awareness of the different roles of the dental team, but many had been seen and were happy to be treated by non-dentists.

### **Scientific summary**

### **Background**

Maximising health gain for a given level and mix of resources is an ethical imperative for health-service planners. The independent review of NHS dentistry in 2009 (Department of Health. NHS Dental Services In England: An Independent Review Led by Professor Jimmy Steele. London: Department of Health; 2009) concluded that there was an overwhelming need to make the best use of the whole dental workforce. Across England, approximately 55% of patients who attend for a regular dental check-up do not require any further treatment. By contrast, half of the population does not attend a general dental practitioner (GDP) on a regular basis and this group tends to be the most disadvantaged and experiences the majority of dental disease. As a result, patients with the least need are being seen and treated by the most expensive resource, the GDP, whereas patients with high levels of need can have problems accessing NHS dental services.

Role substitution occurs when appropriately qualified non-dentist team members undertake the clinical activity previously provided by GDPs. In primary dental care, role substitution is provided principally by dental hygienists (DHs), dental hygiene therapists (DHTs) and dental therapists (DTs). This contrasts with dental nurses, who augment the activity of the GDP in the form of role supplementation. Unlike role supplementation, role substitution has the potential to improve efficiency and the capacity to care and lower costs. In addition, it has the potential to reduce inequalities in service delivery as resources can be diverted to the population with the greatest need. Technical efficiency is defined as the production of the maximum amount of output from a given amount of input so that a service operates at the production frontier. Academic research into technical efficiency is being used increasingly in health-care evaluation, although no studies have empirically investigated the efficiency of NHS dentistry or role substitution in dental practices. By contrast, NHS dental service provision has developed historically, with levels of future service provision being determined by extrapolating trends from past activity into the future, at a time when disease levels are actually reducing.

#### Aims and objectives

This programme of research sought to address three research questions (RQs):

- 1. What is the technical efficiency of NHS dental teams that make use of role substitution?
- 2. What are the barriers to, and facilitators of, role substitution in NHS dental practices?
- 3. How do incentives in the remuneration systems influence the organisation of these inputs and production of outputs in the NHS?

The following were the objectives of the programme:

- to conduct a cross-sectional study to determine the current working patterns of non-dentist team members capable of role substitution in NHS dental practices (DHs, DHTs and DTs)
- to collect input data (NHS hours worked) and output data (clinical activity) from participating practices
- to identify the most technically efficient provider mix among the different provider configurations observed
- to assess the external validity of the methods used for efficiency analysis
- to explore barriers/enablers to increased use of role substitution
- to examine how the technical efficiency of the different provider models varied across different remuneration systems (England, Northern Ireland and Scotland)
- to examine how financial incentives in remuneration systems influence the organisation of inputs and the production of outputs in NHS dentistry.

#### **Methods**

A multimethod approach was adopted, integrating both qualitative and quantitative methodologies. Following NHS ethical approval (12/WA/0403), the membership of the British Society of Dental Hygiene and Therapy (BSDHT) were contacted and asked to complete a screening questionnaire in 2013 to determine the current working patterns of DHs, DHTs and DTs in NHS dental practices. Questionnaires were distributed to all of the 3100 members of the BSDHT (saturation sampling) after being reviewed by the research team's patient and public involvement (PPI) group.

Based on the results of this initial screening questionnaire, NHS practices that utilised role substitution were identified in England, Scotland and Northern Ireland. Once these were identified, a second detailed questionnaire was distributed to capture input data (number of NHS hours worked by the whole dental team and number of surgeries) and important practice demographics. Practices were also asked to consent to its output data being obtained (measures of clinical activity recorded by their relevant NHS payment agency). The input and output data were then linked and anonymised by a third party.

Data envelopment analysis (DEA) was then used to identify the most technically efficient provider mix in participating practices and stochastic frontier modelling (SFM) followed to assess the external validity of the DEA approach. Prior to DEA, the linked data sets were compared with the national average, using data provided by the relevant NHS payment agency. A production possibility frontier was then created using DEA and the relative distance of each practice from the frontier was measured to determine their technical efficiency. A sensitivity analysis was conducted with alternative specifications for different input and output variables to explore if the findings were sensitive to the choice of these variables. Efficiency scores were then regressed onto a range of practice variables to identify correlates of inefficiency. As DEA provides only a relative measure of efficiency, SFM was used in parallel with estimate the frontier, using regression (Cobb–Douglas function) to test the robustness of the DEA findings.

For the qualitative workstream, practices were recruited using the information generated from the screening questionnaire. Technically efficient and inefficient practices were also identified from the quantitative workstream. This was augmented by invitations to DHs, DHTs and DTs via social networks. Semistructured interviews were then conducted to explore the potential barriers to, and facilitators of, role substitution in NHS practices, using interview schedules that had been developed iteratively with the help of the research team's PPI group.

Data collection and analysis were run concurrently between October 2013 and May 2015, using a framework informed by the literature and the PPI group. This was then developed further as the findings emerged. All interviews were audiorecorded and transcribed verbatim. Thematic analysis was carried out by group, so practice principals (PPs), non-dentist team members and patients were coded separately. Interpretations were then pooled and edited in the presence of all three researchers to produce the final version of the coding frame, with disputes being resolved using a majority voting system. Data analysis was undertaken concomitantly with the interviews, which continued until saturation.

To examine the third RQ, data relating to a natural experiment in NHS service provision in Northern Ireland were collected and analysed. Thirteen practices that were remunerated via capitation were compared with 57 matched controls. The latter were selected according to practice location and were paid on a fee-for-service (FFS) basis within the NHS. The data from claim forms were pooled by the Business Service Organisation over 43 months (April 2011 to October 2014) and underwent ordinary least squares regression, using a long panel form. The Ramsey regression equation specification error test was then used to test for model misspecification.

#### **Results**

Two hundred and eighty-seven of the 1859 screening questionnaires (15.4%) were returned from 432 NHS practices. The most common non-dentist team member capable of role substitution was the DH (64.1%), followed by DHTs (33.8%) and DTs (1.0%). Their mean age was 42.1 years and the mean time since qualification was 16.5 years. Almost all were female and the mean time in post was 8.4 years. More than half worked in one practice, but a substantial majority worked in more than one practice. The mean number of additional non-dentist team members that the respondents worked with was 1.7.

One hundred and twenty-one practices provided input data that could be linked to the Business Services Authority in England. When Units of Dental Activity (UDAs) were used as the output measure in England, NHS dental practices operated at a mean level of efficiency of 64%. This changed very little when the outputs were measured in terms of number of patients seen or the number of treatment plans generated. NHS dental practices that did not use any form of role substitution had a higher mean level of efficiency (68%; n = 39). Any use of non-dentist team members was found to be associated with statistically significant lower efficiency scores (14% lower for UDAs, 11% lower for treatment plans or patients seen) than practices that used no role substitution.

No significant interactions with patient population characteristics were found. Correlations between efficiency scores estimated by SFM and DEA were > 0.6, which supported the robustness of estimated DEA efficiency scores and the internal validity of the approach. Efficiency in NHS dental practices in Northern Ireland and Scotland was consistently higher (94% and 80%, respectively), although it was difficult to make robust comparisons across the two countries because of their smaller sample sizes (n = 20 and n = 29, respectively).

Sixteen PPs and 17 non-dentist team members were interviewed. The use of role substitution in the NHS across the practices was highly varied. Some practices used DHs, DHTs and DTs to their full scope of practice and were enabling them to undergo further training; whereas in other practices, they were limited in the duties they could carry out and were used primarily to complete routine scale and polishing.

The PPs interviewed felt that the greater use of role substitution did not result in financial gain. Instead, the current NHS dental contract appeared to disincentivise its use. In addition, the attitudes and beliefs of PPs towards non-dentist team members appeared to be highly influential in terms of how the practice was organised and whether or not the practice culture supported role substitution. Practices with low levels of role substitution appeared to be led by PPs who were concerned about the abilities of DHs, DHTs and DTs, in particular how much longer it might take to carry out treatment when compared with a GDP. PPs who used non-dentist team members tended to have confidence in the ability of their own staff, but were less confident in the use of role substitution more generally.

Among non-dentist team members themselves, a number of day-to-day barriers were found to reduce the efficiency of working in the practice environment. These included not being able to prescribe fluoride varnish and the inability to prescribe local analgesia or radiographs. Notably, a substantial proportion of DHs, DHTs and DTs worked across a number of practices (between two and eight).

The majority of patients had a low level of awareness of the roles of different members of the dental team. Most patients interviewed were familiar with GDPs and DHs, but they were not aware that DHTs and DTs could undertake restorations or extractions. Restorations, crowns and bridges were identified as activities undertaken by GDPs, whereas periodontal treatment ('hygiene work') fell within the expertise of DHs. Most patients interviewed were not interested in the debate within the profession about who could treat them. Instead, they reported that they just wanted to know that any treatment being carried out was undertaken by a suitably qualified clinician. They all appeared to place a great deal of trust in the NHS, the regulatory system and in the opinion of their GDP.

The results from the natural experiment in Northern Ireland showed that the mean number of monthly treatment items delivered by FFS practices was higher than the number remunerated by capitation (922 and 811 treatment items, respectively). The mean monthly number of treatment plans delivered to patients was higher in the capitation practices than in the FFS practices (463 and 392, respectively). The mean monetary value of the treatment plans delivered was lower in capitation practices (by £10.83). These differences were statistically significant at a 5% level.

There was no evidence of differences in patient selection between the two types of practices, although practices paid by capitation provided, on average, 9.7 fewer examinations, 17.2 fewer fillings and 11.5 fewer scale and polish services per 100 unique patients per month (p < 0.00). They also provided an average of 6.3 more extractions per 100 unique patients per month (p < 0.00). The volume of fluoride varnish applications per patient and per 100 child registrations was not significantly different between practice groups.

#### **Conclusions**

The extent of role substitution in NHS dental practices appears to be relatively low and mainly limited to the use of DHs for routine periodontal treatment. NHS dental practices that utilised fewer non-dentist team members were associated with higher levels of technical efficiency, that is as role substitution in NHS practices increased, their relative efficiency dropped. The efficiency of role substitution in NHS dental practices appeared to be heavily influenced by the remuneration system. The inability of non-dentist team members to contract directly with the NHS meant that it was not possible to determine the technical efficiency of individual DHs, DHTs and DTs. They were also subject to the prevailing organisational culture within each practice, restrictions on their ability to prescribe and the views of the individual PPs that employed them.

The scope for addressing the recommendations of the Independent Review of NHS dentistry (Department of Health 2009) to make best use of the whole dental workforce appears limited, as the traditional model of care using GDPs predominates. Aligning financial incentives to encourage the greater use of role substitution would be an important intermediate step, although the development of a NHS contract with DHs, DHTs and DTs would offer greater flexibility for commissioners of services. As oral health continues to improve, this is important if maximising health gain for a given level and mix of resources remains an ethical imperative for the NHS. Failure to do so will increasingly mean that the most expensive resource treats those with the least need.

Further research is needed to:

- 1. explore new models of NHS dental care that are efficient and make best use of increasingly scarce resources (identifying and assessing different approaches to organising the delivery of dental health care in the NHS)
- 2. determine the cost-effectiveness of new models of dental care based on role substitution in NHS dentistry
- 3. examine the quality of the care provided by non-dentist team members and understand how they compare with GDPs
- 4. develop a needs-based workforce plan for NHS dentistry that makes best use of the whole dental team
- 5. explore the impact of traditional and new models of care on the health inequalities of NHS dental service provision, including equity, service coverage and access to care.

#### **Funding**

Funding for this study was provided by the Health Services and Delivery Research programme of the National Institute for Health Research.

### **Chapter 1** Overview of the report

### **Outline of the report**

This report is composed of seven chapters. Chapter 1 provides an introduction to the project and sets the scene.

#### **Introduction**

Maximising health gain for a given level and mix of resources is an ethical imperative for health-service planners.<sup>1</sup> An independent review of NHS dentistry in 2009<sup>2</sup> concluded that there was an overwhelming need to make best use of the whole dental workforce. Across England, approximately 55% of patients who attend for a regular dental check-up do not require any further treatment.<sup>3</sup> This costs the NHS over £1B per annum and represents around one-quarter of the total NHS expenditure on dental care.<sup>3</sup> By contrast, half of the population does not attend a general dental practitioner (GDP) on a regular basis, and this group tends to be the most disadvantaged and experiences the majority of dental disease.<sup>4,5</sup> As a result, patients with the least need are being seen and treated by the most expensive resource, the GDP, whereas patients with high levels of need can have problems accessing NHS dental services.<sup>5</sup>

Role substitution occurs when appropriately qualified non-dentist team members undertake the clinical activity previously provided by GDPs.<sup>6</sup> In primary dental care, role substitution is provided principally by dental hygienists (DHs), dental hygiene therapists (DHTs) and dental therapists (DTs). For the remainder of this report, non-dentist team members capable of role substitution are referred to as dental care professionals (DCPs). This is in contrast to dental nurses, who augment the activity of the GDP in the form of role supplementation. Unlike role supplementation, role substitution has the potential to improve efficiency and the capacity to care and to lower costs. In addition, it has the potential to reduce inequalities in service delivery as resources can be diverted to the population with the greatest need. DCPs cover many different categories of dental professions. The scope of this programme of research was to examine DCP roles that could be used in role substitution, specifically DT, DHT and DH. For concision, this subcategory of roles (where role substitution may be expected to take place) is referred to as DCPs in the remainder of this report.

Technical efficiency is defined as the production of the maximum amount of output from a given amount of input so that a service operates at the production frontier.<sup>7</sup> Academic research into technical efficiency is being used increasingly in health-care evaluation,<sup>8</sup> although no studies have investigated the efficiency of NHS dentistry or examined the impact of role substitution in dental practices. By contrast, NHS dental service provision has developed historically, with levels of future service provision being determined by past levels of provision, at a time when disease levels are actually reducing.<sup>9–11</sup>

General dental practitioners are acutely sensitive to financial incentives within the dental contract,<sup>12–14</sup> although intrinsic motivation and professional standards can also be important moderators.<sup>15</sup> Retrospective fee-for-service (FFS) remuneration systems can lead to overtreatment in order to maximise profit.<sup>13,16,17</sup> Prospective per capita remuneration systems reduce the financial risk for the third-party payer, at the potential cost of patient selection (skimping and dumping) and undertreatment.<sup>18–20</sup>

Empirical research in medical settings suggests that appropriately trained nurses can deliver high-quality care that matches doctors' performance in preventative health care, routine follow-up of patients with long-term conditions and as the first contact for people with minor illness.<sup>21,22</sup> However, efficiency gains are possible only if the doctor stops carrying out the tasks delegated to nurses.<sup>23</sup> In addition, role substitution may challenge the professional identity of both professional groups, leading to opposition as clinicians struggle to maintain traditional boundaries.<sup>24,25</sup>

Given the hitherto ad hoc approach to service design, it is important to determine the most technically efficient model for role substitution a priori. As outlined above, in dentistry this has the potential to increase efficiency and effectiveness in service provision.<sup>26</sup> It also has the potential to release resources and increase the capacity to care.<sup>27–30</sup> Harris and Sun<sup>31,32</sup> found that role substitution may be effective in improving efficiency, but this may be limited to particular situations when conditions are conducive. Therefore, not only is it critical to determine the most technically efficient role-substitutive models, but it is equally important to explore the values of policy-makers and clinicians to determine the factors affecting the implementation of such innovative designs and how patients would view such a change in service design.

This programme of research sought to address three research questions (RQs):

- 1. What is the technical efficiency of NHS dental teams that make use of GDPs and DCPs, that is, what is the relationship between clinical inputs and outputs across different role-substitutive models used in the UK?
- 2. What are the barriers to, and facilitators of, role substitution in NHS dental practices?
- 3. How do incentives in the remuneration systems in the UK influence the organisation of these inputs and production of outputs in the NHS?

### **Aims and objectives**

The aim of this programme of research was to undertake a mixed-methods study to determine the technical efficiency of role substitution between GDPs and DCPs in NHS dental practices and ascertain if this is influenced by the incentives within the NHS remuneration system. Role substitution is defined as DCPs undertaking the clinical tasks previously provided by GDPs. Barriers of, and enablers to, the greater use of role substitution in a NHS dental practice were also examined, from the perspective of policy-makers, GDPs, DCPs and patients.

The objectives of the programme of research were to:

- conduct a cross-sectional study to determine the current working patterns of DCPs in NHS dental practices
- collect input data (NHS hours worked) and output data (clinical activity) from participating practices
- identify the most technically efficient provider mix among the different provider configurations observed in the study practices
- assess the external validity of the methods used for efficiency analysis
- explore barriers/enablers to the greater use of role substitution; and develop an understanding of the configuration of the dental team, collect the input data (NHS hours worked) and gain consent to collect the output data
- examine how the technical efficiency of the different provider models varies across different retrospective and prospective payment systems for adults in the UK
- explore how financial incentives in remuneration systems influence the organisation of inputs and the production of outputs in NHS dentistry.

### Research design and project overview

A multimethod approach was adopted, integrating both qualitative and quantitative methodologies.

Data envelopment analysis (DEA) and stochastic frontier modelling (SFM) were used to determine the technical efficiency of different role substitution models.<sup>33</sup> Technical efficiency is the effectiveness with which a given set of inputs is used to produce an output. A dental practice is said to be technically efficient if it is producing the maximum output from the minimum quantity of inputs, such as labour, capital and technology.<sup>33</sup>

Data envelopment analysis accommodates multiple inputs and multiple outputs in a single measure of technical efficiency and has become the dominant approach to efficiency measurement in health care and other sectors of the economy.<sup>34</sup> The most efficient practices/sites form a frontier of productive efficiency and those that lie below this frontier are allocated proportionally smaller efficiency values, which correspond to their distance from the frontier. Optimal service design is then relative to service design observed in other dental practices, after differences between practices in workplace and environmental characteristics are controlled for.

Stochastic frontier modelling is a regression approach to estimating the frontier of productive efficiency. The level of variation in health-care output that is not explained by variation in the level of inputs (i.e. the regression residuals) forms a measure of relative efficiency compared with average practices. The method divides the regression residuals into a 'noise' or 'measurement error' component and an efficiency component. The last component is used to make an assessment of how far each role substitutive model differs from the most efficient use of role substitution.

Estimates of differences between practices in efficiency of production were analysed and interpreted using DEA and validated using SFM. Factors that influenced the efficiency of different models of role substitution were explored and sensitivity analyses were also undertaken to determine the extent to which efficiency was affected by changes in other variables. Both DEA and SFM investigate units of production and do not account for the quality of dental health care delivered by practices.

Efficiency is understood in terms of the level of inputs used to achieve the outputs produced ('technical efficiency'). Technical efficiency, the effectiveness with which a given set of inputs is used to produce an output, was determined using two principal measures of activity: total clinical activity and patient throughput. In England, the former is constrained as a result of the capped nature of the NHS dental contract. Dental practices face penalties if the contract value falls below 4% of its Annual Contract Value (ACV) or exceeds its ACV by 2% or more. This means that outputs are constrained in the DEA computation for dental practices from England.

The qualitative component of the programme was undertaken in parallel with the quantitative element. Interviews were undertaken with policy-makers, GDPs who made use of role substitution, GDPs who did not, DCPs and patients. The final number of interviews was determined by saturation (i.e. no new themes were emerging). The semistructured interviews were recorded and transcribed verbatim. Field notes were also undertaken as appropriate. The transcripts then underwent thematic analysis using the constant comparative method.<sup>35</sup> Initial coding focused primarily on the barriers to, and facilitators of, to role substitution in line with our RQ. In addition, inductive coding of data alongside the field notes was undertaken to produce a 'thick description' of the social processes involved.<sup>36</sup>

### **Structure of the report**

This report is arranged as follows.

Chapter 2 describes the historical context of NHS dentistry, the use of role substitution in NHS dentistry and the role of financial incentives in how services are organised.

Chapter 3 presents a descriptive study that was undertaken across the UK to understand the current levels of DCP utilisation.

Chapter 4 describes an economic analysis of the technical efficiency of DCP utilisation in England, using DEA and SFM. Supplementary analyses were performed for Northern Ireland, Scotland and the community dental services. These were based on much smaller samples of practices and could not be combined with the English analysis because of differences in the way these services are managed and funded. The results of these supplementary analyses are provided in *Appendix 1*.

Chapter 5 examines role substitution using the qualitative paradigm, exploring the barriers to, and facilitators of, role substitution in NHS dental practices (thereby addressing RQ2).

Chapter 6 then uses data from a large, 3-year project in Northern Ireland to look at how incentives in the remuneration systems in the UK influence the organisation of inputs and production of outputs in the NHS.

Finally, *Chapter 7* integrates the findings from the empirical work and provides an overview and assessment of the contribution of role substitution and financial incentives to the technical efficiency of practices in NHS dentistry.

## **Chapter 2** Organisation of NHS dentistry in primary care

#### Introduction

This chapter provides the context for the programme of research undertaken. It describes the organisation of NHS dentistry in primary care in England, Northern Ireland and Scotland. It then goes on to describe role substitution in NHS dentistry in primary care and describes the role of financial incentives in NHS dentistry.

### Organisation of NHS dentistry in primary care in England and Wales

NHS dentistry began in 1948. The National Health Service Act (1946)<sup>37</sup> had three key principles: (1) no one would ever have to fear not getting care they needed because they could not afford it; (2) care would be free at the point of delivery; and (3) care would be based on clinical need. As a result, NHS dental services were provided free of charge for the entire population. GDPs were considered to be independent contractors under the NHS [General Dental Services (GDS) contracts] regulations and so could establish their practice anywhere in the UK. GDPs' GDS remuneration was based on a FFS basis (i.e. the volume and type of work undertaken) and, although they were self-employed, dentists were eligible for a NHS pension. By 1950, the government became concerned about the affordability of the new service, given the volume of work being undertaken by NHS GDPs (mostly extractions and fillings). This resulted in the introduction of the Patient Charge Regulations (1952),<sup>38</sup> which required patients to contribute to the bulk of the cost of treatment according to a Statement of Dental Remuneration (a list of available treatments under the NHS).

Other than minor changes to the Statement of Dental Remuneration, the shape of GDS provision remained relatively static until the introduction of a new contract in 1990.<sup>39</sup> For the first time, the NHS dental contract contained an element of prospective payment. The remuneration arrangements for adults and children were split, with adults still treated under FFS arrangements (plus additional continuing care payments) but children treated on a wholly capitation basis. As a result, approximately 20% of the income for NHS GDPs was based on the number of dental patients registered (on a per capita basis), rather than simply being based on the volume of activity on a FFS basis. The policy intention of the 1990 contract was to encourage registration and promote prevention and continuity of care, moving service provision away from treating disease to maintaining oral health.<sup>39</sup> However, higher-than-expected expenditure in the following year led to the government making substantive cuts to NHS fees in 1991. Those in the dental profession felt that they were unfairly penalised by this 'claw-back' and many GDPs felt unhappy with the new NHS system of payment. This triggered a progressive shift towards the provision of privately funded dentistry within the profession, with a reduction in the availability of NHS services. By the mid-1990s, access to dental services for NHS patients was becoming a political issue and it was increasingly recognised that reform of the NHS contract was necessary. As highlighted by the Bloomfield report<sup>40</sup> the 'system of remuneration for dentists seems to have an inherent leaning towards instability which threatens to undermine the commitment of dental practitioners to the NHS'.

The NHS (Primary Care) Act 1997<sup>41</sup> enabled the voluntary establishment of personal dental services (PDSs) pilot schemes to explore alternative ways of delivering NHS dental services. A key feature of these new contracts was how they were tied to local issues around need and access to care through contracting arrangements with primary care trusts (PCTs). For NHS GDPs, the new PDS contracts offered greater flexibility for the provision of services, with less emphasis on the volume of activity provided. Instead, NHS GDPs were paid on a per capita basis and rewarded for improving access to care and maintaining oral

health for those patients who were already registered. The net effect of paying the PDS schemes was a dramatic reduction in NHS service activity, a fall in the provision of complex treatments and loss of Patient Charge Revenue.<sup>42</sup> The last was particularly worrying for the government as patient charges had become an important component of funding for NHS dentistry. However, many viewed the PDS pilots as a success, given the focus on local needs.<sup>42,43</sup>

In 2000, *Modernising NHS Dentistry – Implementing the NHS Plan*<sup>44</sup> was introduced in England. Again, improving access was considered to be the most important policy objective for the government and the new proposals gave PCTs powerful new commissioning tools to improve access to NHS dentistry, the provision of preventative services and to monitor the performance of GDPs. This was further emphasised in *Options for Change* in 2002.<sup>43</sup> However, a question that remained for the government was how to concomitantly improve access to care and the provision of prevention, while maintaining Patient Charge Revenue (i.e. the income generated from patient charges, which accounts for approximately 25% of the payment to the NHS GDP).

In 2006, a new NHS dental contract was introduced in England,<sup>45</sup> organised around a local commissioning model. NHS GDPs were paid according to activity categorised in one of three bands of treatment, rather than the 400 individual treatment items in the Statement of Dental Remuneration. Band 1 attracted one Unit of Dental Activity (UDA) and included an examination, radiographs as appropriate, and a simple scale and polish. Band 2 attracted three UDAs and included restorations, extractions and root canal treatments, while more complex crowns, bridges and dentures attracted 12 UDAs as a band 3 treatment. Patient Charge Revenues were also simplified so that there were only three levels of payment tied to each band. To reflect individual variation among NHS GDPs, the value of the UDA was based on individual earnings in a reference period, although for most, this meant that a UDA was worth between £20 and £25. One feature of the new NHS contract was that NHS GDPs would receive three UDAs for one band 2 treatment (i.e. they would be paid the same amount), whether they did one restoration in a single visit or multiple restorations over a number of visits. In a similar manner, NHS GDPs were also allocated 12 UDAs for one band 3 treatment irrespective of whether they provided one crown or multiple crowns within one course of treatment.

A key feature of the 2006 contract was cost containment, that is, GDPs' NHS activity and revenue was capped at an agreed number of UDAs per year, known as an ACV. NHS GDPs were then paid one-twelfth of their ACV on a monthly basis, less than the anticipated Patient Charge Revenue. As a result, NHS GDPs' outputs under the new contract in England were constrained and they were penalised if they underperformed (< 96% of their ACV) or overperformed (> 102% of their ACV). Patient registration in the 2006 contract also ceased to exist, along with the contractual responsibility for NHS GDPs to provide emergency care for their patients. Instead, this responsibility was devolved to PCTs along with the planning and securing of NHS dental services for their locality.

The effect of this NHS contract change in England in 2006 was investigated by an earlier National Institute for Health Research (NIHR) Service Delivery and Organisation programme funded study (reference number 08/1618/158).<sup>13,14</sup> Large and abrupt changes in the provision of a number of treatments coincided with the introduction of the 2006 contract.<sup>13</sup> The number of complex treatments provided (root canal treatments, crowns and bridges) fell dramatically, whereas the number of extractions rose. This appeared to reflect an increase in those activities that were easier and less costly to perform at the expense of those treatments that were more time intensive or that incurred laboratory costs for crowns and bridges. The authors concluded that 'the change in treatment patterns suggests that significant numbers of dentists [were] attempting to hit their UDA contract targets in the most efficient way possible, i.e. shifting towards treatments where rewards are high relative to costs, as opposed to selecting on the basis of clinical factors'. <sup>13</sup> In addition, McDonald *et al.* <sup>14</sup> remarked 'it is the interests of dentists, as opposed to patients, which [were] being prioritised'.

Given this change to activity following the 2006 contract and mounting dissatisfaction among NHS GDPs and patients, the House of Commons Health Select Committee undertook a review in 2008.<sup>46</sup> The review identified key problems with the NHS contract, including failure to improve patient access, higher patient

charges, problems caused by removing the patient registration process, lack of incentives to promote prevention, incentives to provide treatment which may not best align with a patient's needs (e.g. extraction over a root canal treatment) and a highly unpopular UDA payment system. This was followed by an independent review of NHS service provision in England by Steele in 2009.<sup>2</sup> Again, this found that the incentives for NHS GDPs were not aligned with the goal of securing oral health. The independent review highlighted a number of substantive inefficiencies in the system and recommended that a new NHS contract that had clearer incentives for improving health, access and quality be developed in England.<sup>2</sup>

In 2011, 70 NHS dental pilots were established in England to explore how NHS dentistry could shift its focus from primarily treatment and repair to prevention and oral health.<sup>47</sup> This process was built around a new oral health assessment that aimed to assess the risk of the patient, enabling them to be assigned to new clinical care pathways to create a healthy oral environment based on their level of need. This was supported by three different types of remuneration models to test different ways of remunerating NHS GDPs. All three models were based on a prospective payment system (per capita), in which dental practices would be remunerated based on the number of patients they cared for and their quality of care (not FFS).

Four years on, the information generated from these pilots remains unclear. Although the dental pilots presented an ideal opportunity to address the issues raised by the Steele review,<sup>2</sup> the findings reported were mainly descriptive in nature. The evaluation of the oral health assessment (and associated risk algorithms) was also limited to professional and social acceptability and its predictive validity remains untested.

In 2015, new dental 'prototypes' were introduced, which appeared to build on the limited information generated from the pilots.<sup>48</sup> The principles of the 'prototypes' were again based on an initial assessment and a care pathway approach to prevention and care. In aligning to the *Building the NHS of the Five Year Forward View*,<sup>49</sup> emphasis was placed on prevention, empowering patients to take control of their own oral health and making the most efficient use of NHS resources. Although they still included capitation payments and remuneration for quality, an important difference with the 2011 pilots was an emphasis on activity, through a blend of both capitation and FFS.<sup>48</sup>

In terms of the dental workforce, the latest information from the Health and Social Care Information Centre reveals that 23,723 NHS GDPs were undertaking activity within the NHS during the period April 2013 to April 2014.<sup>3</sup> This is an increase of 17.7% (n = 3563) compared with 2006, when the existing contract in England started. Approximately half of the workforce was female (46.1%). A total of 29.9 million patients were seen in the 24-month period ending June 2014 (55.9% of the total population). In 2014, 54.5% of all NHS activity was related to band 1 payments, 29.7% related to band 2 and 5.6% related to band 3. The remaining UDAs (10.2%) were allocated to emergency and other payments.

### Organisation of NHS dentistry in primary care in Northern Ireland and Scotland

The provision of NHS dentistry in Northern Ireland and Scotland followed a similar trajectory to NHS dentistry in England up until the 2006 contract. Instead of moving towards a banded system of retrospective payment (the UDA system), both Northern Ireland and Scotland remained with the existing Statement of Dental Remuneration and FFS payment system. As such, although a few minor changes were made at a national level, the shape of the NHS contract in both countries is very similar to the 1990 contract in England and is primarily based on FFS for adults and capitation for children.

In Northern Ireland, approximately 1100 NHS GDPs work in 380 dental practices. Unlike England's current NHS contract (the UDA system), there is no local commissioning model and so outputs remain unconstrained. This means that there is no limit on the amount of FFS NHS activity that can be produced. This has led to costs for NHS dentistry doubling in the last 5 years. In 2013–14, the total cost was estimated to be £124M, of

which £20M came from Patient Charge Revenue. There are also no limitations on where new NHS practices can be established or the number of new NHS GDPs that can enter the market. As highlighted above, this is in contrast to England, where, under the local commissioning model, new NHS GDPs have to obtain agreement before setting up any new service. The number of NHS GDPs in Northern Ireland in 2014 was twice the level recorded in 1989.

As a result of this dramatic rise in activity, costs and the number of NHS GDPs, Northern Ireland began piloting a new dental contract in 2015. This is based on per capita payments and is the subject of a new NIHR Health Services and Delivery Research programme study by the authors (reference number 14/19/12). The technical efficiency of NHS GDPs working under a capitation arrangement in an earlier 3-year pilot aimed at improving access to care in 2010 is reported in *Chapter 6*.

In Scotland, the number of NHS GDPs rose from 2848 in 2011 to 3035 in 2014.<sup>50</sup> Over 4.6 million people were registered with a NHS GDP in 2014 (87% of the population) and over 74% had seen their NHS GDP in the previous 2 years.<sup>3</sup> The proportion of both children and adults registered with a NHS GDP continued to increase during the last 2 years, with more than 91% of children and 83% of adults registered at the end of March 2014.<sup>3</sup>

# Role substitution in NHS dentistry in primary care

Designing the most appropriate dental workforce for NHS dentistry is critical to ensure 'the right number of people with the right skills are in the right place at the right time to provide the right services to the right people'. 50.51 At its inception in 1948, the only type of clinician working within NHS dentistry was the GDP. In 1921, the Dentists Act 1921<sup>52</sup> created the UK Dental Board, whose role was to be the professional body for dentistry and oversee its practice, regulating who could practise as a GDP. In 1946, the Teviot Report<sup>53</sup> argued that the dental profession had become sufficiently mature to self-govern and, by 1956, the new Dentists Act 1956<sup>54</sup> had enshrined this principle in law. To discharge this Act, the functions of the board were taken over by the General Dental Council (GDC).

The Dentists Act 1956<sup>54</sup> also allowed the training of 'dental auxiliaries' for the first time and, in 1960, the first 'auxiliary' school was opened at New Cross Hospital in London, UK. The newly formed GDC subsequently developed the regulated titles of 'DH' and 'DT', but restricted the types of clinical activities that they could legally undertake. The duties of the former role related to the provision of prevention and periodontal treatment, while the latter role was permitted to provide a range of direct restorative procedures and extract deciduous teeth. DHs were allowed to practice in the NHS, but it was not until 2002 that a new Dentists Act 2002<sup>55</sup> allowed DTs to work in NHS dental practices (their role before this had been limited to the provision of care in NHS community dental services settings only). Since 1983, UK dental schools also offered 'dual' integrated training over a period of 2 years. The new 'DHT' was able to offer the full range of clinical activities that both DHs and DTs could legally undertake.<sup>56</sup>

Common to all these roles, however, was the inability of patients to directly access their services. All patients had to be seen by a GDP first and then referred on to a DH, DT or DHT. A further factor in the provision of care within the NHS that had kept GDPs as the 'gate-keeper' was the inability of DCPs to directly contract with the NHS. Under both the current GDS and PDS agreements of 2005, only GDPs can legally establish a NHS contract and be entered onto a Performers List to provide NHS care. In 2013, a contested landmark decision by the GDC<sup>57</sup> allowed patients for the first time to access the care of DHs, DTs and DHTs without a referral from a GDP. This was known as 'direct access' and led to the expansion of DCPs' scope of practice.<sup>58</sup> However, DHs, DTs and DHTs remain unable to contract directly with the NHS and so the majority are employed by NHS GDPs on a sessional, salary or activity basis.

The Nuffield inquiry on the education and training of personnel auxiliary to dentistry was published in the UK in 1993.<sup>59</sup> It focused on a redefinition of the dental team through which dental care would be delivered and also examined if, and how far, the role of auxiliaries could be expanded. Since the publication of the Nuffield report,<sup>59</sup> much has been written in the dental literature about the need to adopt a model of care in the NHS that makes greater use of 'skill mix'.<sup>6</sup> Skill mix is a term that is used to describe a model of care in which the whole of the clinical team is utilised in delivering service activity.<sup>60</sup> The use of skill mix in medicine is relatively developed, but dentistry has always appeared to lag behind.<sup>6</sup> Skill mix can be further subdivided into role substitution and role supplementation. The former is when DCPs undertake clinical tasks instead of a GDP, whereas the latter is when other team members augment the activity of a GDP (e.g. dental nurses or dental technicians). It is argued that both should play a substantive role in shifting dentistry from a 'cure' to a 'care' culture in the future.<sup>61</sup>

Population health needs are changing. From the most recent epidemiological survey undertaken in the UK, 90% of young adults are expected to have more than 20 teeth in 10 years' time and the levels of dental caries and periodontal disease have fallen dramatically. However, social inequalities have remained relatively static, disease in young children remains intransigent (although geographical variations are evident) and there is a burgeoning group of older people who are now keeping their teeth for longer. In addition, approximately 45% of the adult population in England do not use NHS dental services routinely and much of the activity that is undertaken by NHS GDPs relates to band 1 (primarily check-ups without treatment) activity.

As population needs continue to change, there are increasing calls for the development of a NHS dental workforce that can best meet the emerging needs of these strata: maintaining the health of routine attenders, while freeing up resources to increase the capacity to care and reduce social inequalities in oral health.<sup>65</sup> These calls reflect a growing international debate on the subject.<sup>27,28,60</sup> In a review of the international literature, Nash *et al.*<sup>66</sup> conclude that 'access to basic dental care will not be available without the utilisation of dental therapists in the workforce', while Johnson<sup>67</sup> argues for a paradigm shift using DCPs to shift the culture 'from treatment to prevention, wellness and self-care'.

Role substitution by DHs, DTs and DHTs is long established in a number of countries and there are varying models of independent practice. <sup>66,68</sup> Sweden and the Netherlands were among the first countries to allow DHs to practise independently, and this was legalised in 1964 and 1978, respectively. <sup>69</sup> Finland, Denmark and Norway have allowed independent practice since 1994, 1996 and 2001, respectively. Similar practices are found in Switzerland, which started in 1997, and in Italy, where DHs have been able to work as independent practitioners since 1999. <sup>69</sup> In the USA and Canada, DHs are part of a growing profession; they can practise with varying degrees of independence in a number of US states, such as California, Colorado, Montana, Nebraska, New Mexico, Oregon and Washington, and Canadian provinces such as British Columbia, Alberta, Saskatchewan and Manitoba, although some restrictions remain in terms of settings. <sup>70</sup> Tasmania has a liberal regulatory model where DHs and DTs practise independently and can own private practices. <sup>69</sup> DTs are also considered to be independent in New Zealand, although they are not able to treat adults. In Samoa and Singapore, DTs must work under the supervision of a GDP. Fiji has allowed DTs to assume independent responsibility for managing clinics since 1985 and DTs have been allowed to practise independently in South Africa since 1994. <sup>69</sup>

The research evidence for the greater use of DHs, DTs and DHTs is emerging. In Galloway *et al.*'s<sup>71</sup> systematic review, DCPs were found to be better than GDPs at oral health promotion, and in the most recent Cochrane systematic review<sup>72</sup> survival rates of resin fissure sealants were similar over various time periods. Previous studies have found that DCPs are able to screen for oral diseases.<sup>73–79</sup> Treatment outcomes for atraumatic restorative techniques were also similar across a number of criteria for single-surface (small) and multisurface (large) restorations for DCPs and GDPs.<sup>72</sup> In a review of the literature on direct access,<sup>80</sup> DCPs were again found to perform to a similar standard as GDPs, although the evidence base found by Dyer *et al.*<sup>72</sup> and Brocklehurst *et al.*<sup>80</sup> was mainly descriptive with only a few experimental studies.

Although DHs appear to be well-accepted members of the dental team in the UK, 81,82 financial considerations appear to play a significant part in the decision to use a DT in the NHS. 56,83-90 Of the few studies that have examined DTs' profitability, patient charges generated did not cover the cost associated with their use. 91 As a result, many DTs have been employed by the NHS as DHs in England, limiting their ability to use the full range of their skills. 92 Although the social acceptability of DTs appears to be positive, 93-95 public awareness of DTs as a professional group is not widespread. 93 Despite this, it does appear that adults are willing to receive treatment from DCPs under the NHS and there is evidence of increased patient satisfaction. 95 However, some patients expect to pay less for treatment provided by a DCP and are reticent about care provided by a DCP for young children or when anxious. 93

The NHS GDPs run their practices as businesses to offset the cost of the capital risk of the premises and the equipment that they own, while ensuring liquidity to cover their overheads. <sup>46</sup> This means that the use of role substitution must be profitable and is sensitive to the remuneration system. <sup>96</sup> In medicine, transaction costs can be offset by economies of scale, which enable a broader range of services to be made available. <sup>97</sup> However, historically the size of dental practices has been limited to a small number of surgeries, with approximately one-third run single-handedly. <sup>98</sup> It also requires all GDPs in the dental team to refer patients to DCPs and there is evidence that not all do this because of the GDP effectively losing income to the DCP. <sup>31</sup>

One study that examined the efficiency of using DTs in the NHS found that this was described by NHS GDPs as the optimum use of surgery time to manage patient flow, provide patient satisfaction and maintain activity levels.<sup>32</sup> However, some efficiency gains were noted in the nine practices examined. Harris and Sun<sup>99</sup> concluded that role substitution 'may be limited to particular situations where conditions are conducive'. For example, the type of patient care was found to be relevant to whether or not referrals to DTs were deemed efficient. Patients with high levels of disease (many restorations) were thought less suitable for referral because GDPs would undertake care in fewer visits and there was a risk of complications necessitating referral back to a GDP.

Management of change is also a potential problem with role substitution, as professionals seek to protect their clinical roles and maintain traditional boundaries. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resource skills. Managing a transition to role substitution takes time and good human resourc

# Financial incentives in NHS dentistry in primary care

As highlighted above (see *Organisation of NHS Dentistry in primary care in England and Wales*), NHS GDPs are part of an altruistic profession, yet much of their activity in primary care is driven by the 'profit principle' to maintain the viability of their practices. NHS GDPs are acutely sensitive to incentives within the remuneration system. <sup>13,15,16,104</sup> Retrospective payment systems like FFS commonly lead to overtreatment, <sup>17,18,104</sup> whereas prospective systems or 'per capita' payments often lead to undertreatment. <sup>18</sup> When incentives promote preventative activity, NHS GDPs tend to actively engage and shape their dental team accordingly, greatly increasing the use of role supplementation and substitution. <sup>105</sup> Little is known about how the efficiency of NHS dental services is affected by role substitution. The only study in the UK (at the time of writing) used only nine practices. <sup>99</sup>

Oral health follows a social gradient, with the poorest experiencing the majority of the disease. <sup>106</sup> However, prevention and treatment of oral diseases are determined by an individual's ability to pay for services. <sup>107</sup> As a result, access to services tends to be greatest in those groups with the fewest needs. <sup>108,109</sup> Public funding of health and social care provides a means of overcoming the divergence between the ability to pay for care and need for care. It offers the opportunity for improving both efficiency (increasing health gains produced from available health-care resources) and equity (removing barriers to access to services associated with individuals' income or wealth). In an evaluation of the previous contract reform in England, Whittaker and Birch<sup>110</sup> found that the dental reforms impacted on access. Copayments via Patient Charge Revenue have also been a fundamental component of the NHS dental contract since the 1950s. In the USA, Manning *et al.*<sup>111</sup> found that dental service use increased as copayments decreased in a randomised trial of alternative insurance plans, whereas Parkin and Yule<sup>112</sup> found a negative relationship between price and dental care use in Scotland. Little is known about the impact of remuneration systems on the efficiency of NHS practices and how they might influence outputs such as access.

# **Concluding remarks**

The preceding sections have highlighted a range of issues relating to the NHS dental contract in England, Northern Ireland and Wales. It has also highlighted the factors that can impact on the use of role substitution in NHS dentistry and how in turn, this may impact on key areas of quality, including access, efficiency, efficacy, inequalities and social acceptability. *Chapter 3* explores the distribution of DCPs across the UK, if role substitution is associated with efficiency in NHS dentistry in England, the types of barriers to and enablers of the greater use of role substitution in NHS practices across the UK and the role that financial incentives have on the organisation and efficiency of NHS dental teams.

# **Chapter 3** Preliminary screening of NHS dental care professionals

#### Introduction

The publication of the Nuffield Report led many in the dental profession to articulate the importance and potential of using skill mix in NHS dentistry.<sup>59</sup> Although the GDC provided oversight on the range of clinical duties that DCPs could perform, it was not until 2008 that it required DHS, DTs and DHTs to be registered with the dental regulator. Prior to this point, the 'stock and flow' of DCPs in the dental marketplace could be estimated only from the number of training places available in the UK dental schools. By 2008 there were over 5000 DHs and DHTs registered with the GDC and approximately 250 dually qualified DHTs were being trained each year across 10 dental schools.<sup>113</sup>

Despite the potential for using DCPs as part of the NHS dental team, there is evidence that some DCPs are not being fully utilised in the NHS. 92.114–118 The oldest studies found that one-fifth of DH students who graduated from Liverpool in 1989 and one-tenth in 2000 did not stay in the dental profession. 117,118 In 2002, a survey undertaken in the south of England found that only one-fifth of DHs worked within the NHS, spending their remaining time undertaking private activity. 116 In the north of England, this proportion was slightly higher, at one-third, and in Scotland just over half (54%) of all DH activity was undertaken on NHS patients (in those NHS practices that utilised DCPs). 115 The most recent survey across the UK found again that just over one-fifth of practising DHs undertook activity on NHS patients. 114 The proportion of DCPs who work with NHS and private patients was not examined in these surveys, although they are capable of treating both.

There is also evidence from Godson *et al.*<sup>92</sup> that many DTs are not working to their full scope of practice (i.e. the range of clinical tasks that they are trained to perform) in the NHS in England.<sup>92</sup> Many DTs surveyed, who went on to record their views via free-text box (46.6%), stated that they practised exclusively or nearly exclusively as a DH. Approximately one-third (36%) worked wholly under NHS regulations, whereas a further 32% spent the majority of their time (50-99%) on NHS work.

Given this range of utilisation, the objective for this part of the research programme was to conduct a cross-sectional study to determine the current working patterns of DCPs in NHS dental practices across the UK. This was then used as part of the sampling strategy for the studies reported in *Chapters 4* and 5.

# **Methods**

Following NHS ethical approval (12/WA/0403), the membership of the British Society of Dental Hygiene and Therapy (BSDHT) was contacted and asked to complete a questionnaire in 2013. Questionnaires were distributed to all of the 3100 members of the BSDHT (saturation sampling) using a letter of invitation that was cowritten by the research team and the president of the BSDHT. The questionnaires were mailed out using the BSDHT database that was used to distribute their newsletter. The BSDHT is the predominant professional organisation for DHs, DTs and DHTs.

The questionnaire recorded the type of DCP and the address of the practice where they worked. It also asked participants to record the time in their current NHS post, the number of clinically active hours worked per week on NHS patients, the proportion of time working with children and adult NHS patients and the number and type of DCPs that they worked alongside.

The distribution process was managed by the BSDHT. Reminder letters and repeat questionnaires were mailed out twice, so that each of the 3100 members of the BSDHT was approached a maximum of three times.

The results of these questionnaires were then compiled and the addresses were used to geographically locate the practices where they worked, using ArcGIS (Esri, Redlands, CA, USA). Maps were created to record the distribution of NHS provision (see *Figure 1*).

# **Results**

A total of 1859 responses were returned (60.0%), of which 287 (15.4%) were from DHs, DTs and DHTs working in 432 NHS practices. The remaining responses either related to DCPs working in private practice or had been left blank or were unusable.

Based on the responses, the most common DCP type working in the NHS was the DH (64.1%) (*Table 1*) followed by DHTs (33.8%) and DTs (1.0%).

**TABLE 1** Demographics of the DCP sample

Variable	Detail
Type of DCP	
DH	64.1%
DHT	33.8%
DT	1.0%
Other	1.1%
Mean (SD) hours worked	12.6 (11.4)
Mean (SD) age (years)	42.1 (10.9)
Years (SD) qualified	16.5 (12.4)
Proportion (%) female	97.9
Mean (SD) time in post (years)	8.4 (8.7)
Number of NHS practices DCP works across	
1	59.9%
2	27.5%
3	12.5%
≥4	0.1%
Mean (SD) number of additional DCPs at the practice	1.7 (1.8)
SD, standard deviation.	

The mean age of the DCPs was 42.1 years and the mean number of years qualified was 16.5. Almost all of the DCPs were female and the mean time in post was 8.4 years. More than half of the DCPs worked in one practice, but a substantial majority worked in more than one practice. The mean number of additional DCPs that the respondents worked with was 1.7.

The geographical spread of DCPs that work in the NHS is provided in *Figure 1* and *Table 2* highlights the distribution of the 10 most common models in use in the NHS across the UK.



FIGURE 1 Distribution of DHs, DTs and DHTs in NHS practices across the UK.

TABLE 2 Common models of role substitution in sampled NHS practices in the UK

Model	Detail of the model used	England	Wales	Scotland	Northern Ireland	Total
1	DH with no additional DCP	93	11	7	3	114
2	DH with one other DH	39	1	6	0	46
3	DH with one DHT	23	1	2	0	26
4	DH with two other DHs	17	3	7	0	27
5	DH with one DH and one DHT	12	0	2	0	14
6	DHT with no additional DCP	48	2	5	0	55
7	DHT with one DH	22	0	3	0	25
8	DHT with one other DHT	22	2	1	0	25
9	DHT with two DHs	11	0	1	0	12
10	DT with one DH	11	0	0	0	11
Total		298	20	34	3	355

### **Discussion**

The most common type of DCP seeing NHS patients in the group investigated was the DH (64.1%), suggesting that role substitution for periodontal treatment was more common than for DCPs who undertake direct restorations (DHTs and DTs). These results are consistent with Godson *et al.*, <sup>92</sup> who found that, even where DHTs are employed, they are commonly used to undertake periodontal treatment. This may reflect a difference in the availability of work within the NHS or it may equally reflect a reluctance in GDPs to allow DHs, DTs and DHTs to encroach over traditional role boundaries. The role of DT in NHS practices was permitted only in 2002 and there appears to be evidence that both GDPs and dental students are either not aware of the expanded scope of practice for DTs and DHTs, or that they choose not to support this expansion of their role.<sup>113</sup>

Over half of the NHS practices employed DCPs, with the most common additional DCP type being the DH. The mean number of additional DCPs employed at a practice was 1.7 (see *Table 1*). This suggests that the practice culture and ethos is an important determining factor, that is, if skill mix is perceived to be important for the practice, it is common for multiple DCP roles to be employed.

The mean number of hours worked per week by NHS DHs, DTs and DHTs was 12.6, which means that most only worked for three or four sessions at any one practice. This is consistent with the finding that over one-third of these DCP roles worked in more than one practice. Eaton *et al.*<sup>114</sup> found that the mean number of clinical hours worked per week was 24.6 hours. This propensity for part-time working among DCPs is consistent with findings from other studies.<sup>92,119,120</sup> Eaton *et al.*<sup>114</sup> reported that 62.7% of DCPs worked in more than one practice. However, they did not distinguish between NHS activity and private activity, so this may explain some of the differences with this study.

# **Concluding remarks**

The objective of this component of the research programme was to provide a snapshot of DCP utilisation in the NHS and contribute towards the development of a sampling frame for the studies highlighted in *Chapters 4* and *5*. It is clear from the results that not all NHS high-street dental practices use DHs, DTs and DHTs and those that do are not employing them full time. Instead, it appears that it is common for DCPs to work across multiple practices.

# **Chapter 4** Technical efficiency of NHS teams that use role substitution

#### **Introduction**

The independent review of NHS dentistry in 2009<sup>2</sup> concluded that there was an overwhelming need to make best use of the whole dental workforce. The greater use of DCPs and role substitution provides a potential strategy for increasing the efficiency of NHS dentistry. Technical efficiency is defined as the production of the maximum amount of output from a given amount of input so that a service operates at the production frontier.<sup>7</sup> In NHS dentistry, this could be achieved by either increasing the amount of NHS activity produced (including the capacity to care) with the same level of input or producing the same level of activity (output) with a reduced input.

Role substitution is already established in medicine.<sup>21,22</sup> However, practice efficiency is improved only when (1) physician time is released from the tasks delegated to nurses (i.e. nurse time replaces physician time in these activities, as opposed to supplementing it) and (2) that released physician time is used for tasks that only physicians can perform.<sup>23</sup>

Harris and Sun<sup>32</sup> found that role substitution may improve efficiency in NHS dental service provision, but that this may be limited to particular situations or contexts. Therefore, it is important to determine (1) whether or not the way role substitution is used is associated with higher levels of technical efficiency and (2) whether or not the potential efficiency gains of role substitution are constrained by structural constraints. There is little evidence on the economic consequences of role substitution in NHS dental practices,<sup>2</sup> even though such evidence is important for commissioning NHS services across the UK.

In the UK, GDPs who work in the NHS are either practice principals (PPs) or associate GDPs. PPs own the NHS dental practice and are responsible for paying estate, staff and consumables associated with the delivery of a NHS dental contract. Associate GDPs are self-employed NHS GDPs who have their own subcontract with the NHS and who rent facilities and support staff time from the PP.

As highlighted in *Chapter 2*, PPs and associate GDPs in England do not receive any additional remuneration for NHS care delivered in excess of 102% of their ACV and are penalised through a pro rata 'claw-back' of remuneration if their activity falls below 96% of their ACV. As such, practices in England are output constrained and PPs organise their practices to operate within these constraints.

Dental care professionals are a heterogeneous group including DHs, DTs or DHTs. DCPs also include dental nurses and dental technicians, but these roles are predominantly supplementary in nature and so the term DCPs in this report collectively describes those non-GDP clinicians able to perform a range of clinical tasks as defined by their legislated scope of practice (regulated by the GDC).<sup>58</sup> One method of making efficiency improvements in NHS GDP practices is to substitute lower cost time of these DCPs for higher cost GDP time, as any output is constrained by the ACV threshold. However, as DHs, DTs and DHTs cannot contract with the NHS, the predominant model in NHS dentistry in England is to add the number of UDAs that these DCPs produce to the PP's NHS contract. As a result, there is an incentive for PPs to encourage role substitution and a disincentive for associate GDPs to support role substitution.

# Aims and objectives

The aim of this chapter is to determine if the use of these DCPs in NHS practices is associated with higher levels of efficiency of production in service delivery. In particular, the following RQs were addressed:

- 1. Are practices that employ DCPs significantly more technically efficient than those that do not employ DCPs after controlling for other practice/site and patient population characteristics?
- 2. Among practices that employ DCPs, was the level of technical efficiency significantly associated with the level of DCP use after controlling for other practice/site and patient population characteristics?

Although the first two RQs evaluate the efficiency implications of the presence and level of use of DCPs, they do not identify role substitution (i.e. DCPs contributing a higher proportion of total provider time in a practice). To explore the notion of role substitution a third RQ was:

3. Among practices that employ DCPs, is the level of technical efficiency significantly associated with the proportion of total clinical time provided by DCPs?

In order to consider whether or not the efficiency of using DCPs is specific to particular practice circumstances and settings (in particular the patient mix served by the practice), a fourth RQ to be addressed was:

4. Among practices that employ DCPs, was the association between technical efficiency and DCP use dependent on (1) the proportion of adult (non-elderly) patients treated (subgroup of patients with relatively low need compared with children or elderly patients) and (2) the proportion of exempt patients (subgroup of patients with relatively high need)?

Finally, in order to test whether or not role substitution was associated with the mix of services delivered, two further RQs to be addressed were:

- 5. Among practices that employ DCPs, was the proportion of UDAs received for provision of routine services significantly associated with the level of DCP use after controlling for other practice/site and patient population characteristics?
- 6. Among practices that employ DCPs, was the proportion of UDAs received for provision of routine services significantly associated with the proportion of clinical time provided by DCPs after controlling for other practice/site and patient population characteristics?

# **Methods**

# Data collection

This was a multicentre study of NHS dental practices based in the north of England. Contact details of NHS GDPs practising in the north of England were collected from a range of sources. These included the screening questionnaire in *Chapter 3*, Health Education England and the NIHR Clinical Research Networks. Two hundred and one NHS practices in the north-west of England and 143 NHS practices in the north-east were invited to participate. Further data were collected from sites in Northern Ireland and Scotland via the Health and Social Care Board and the Scottish Practice Based Research Network, respectively. The results of analyses using data from Northern Ireland and Scotland are provided in *Appendix 1*.

Each NHS practice was asked to complete a short research questionnaire about the inputs used in the practice's dental health-care production. As a rule of thumb, Dyson *et al.*<sup>121</sup> argues that the selected inputs and outputs for an efficiency analysis should cover the full range of resources used and outputs produced. The selection of inputs and outputs for the questionnaire was decided following a review of the existing economic literature on productive efficiency in dental health care and discussions with local NHS GDPs (Greater Manchester Local Professional Network). The questionnaire was then piloted and further refined after post-pilot interviews with local PPs from Greater Manchester.

The final questionnaire (see *Appendix 2*) was distributed to the identified NHS practices by a third party, who held the database for Health Education England and who anonymised the data. All the PPs were asked to provide consent for the research team to link the information in the questionnaire to data recording the extent of NHS clinical activity undertaken by the practice held by the Business Services Authority (the Information Services Division in Scotland and the Business Services Organisation in Northern Ireland provided the corresponding data for those jurisdictions). The types of input and output data collected in England are provided in *Table 3*. Data were provided for the period 2013–14 and excluded any orthodontic activity (some variables in the questionnaires for Scotland and Northern Ireland were amended to reflect the difference in their remuneration systems; see *Appendix 2*).

Tables 4 and 5 highlight the variables captured by the questionnaire or via the Business Services Authority that were used to capture patient population characteristics and measures of the service mix delivered to patients.

To test the representativeness of the sample from England, data were also collected from the Business Services Authority for the following variables:

- annual number of UDAs
- annual number of patients seen
- annual number of treatment plans (claims)
- percentage of UDAs from adults
- percentage of UDAs from patients aged 40 years and over
- percentage of UDAs from patients exempt from payment charges
- percentage of claims that are band 1
- percentage of claims that are band 2
- percentage of claims that are band 3
- percentage of claims that are urgent
- active NHS GDPs at the practice.

TABLE 3 Input and output variables for the oral health-care production

Variable name	Variable description	Data source
UDAs	Units of dental activity	Business Services Authority
Income	Revenue accrued	Business Services Authority
Patients seen	Number of adults and children seen	Business Services Authority
Treatment plans	Number of individual claims made	Business Services Authority
Weekly DCP sessions with NHS patients	Number of sessions (half days) worked by DHs, DHTs, DTs seeing NHS patients in a typical week	Questionnaire
Weekly GDP sessions with NHS patients	The number of sessions (half days) worked by GDPs (PPs and associate GDPs) seeing NHS patients in a typical week	Questionnaire
Weekly nurse sessions	Number of sessions (half days) worked by dental nurses in a typical week	Questionnaire
Weekly administration staff sessions	The number of sessions (half days) worked by practice managers and administrative staff in a typical week	Questionnaire
Number of surgeries	The number of surgeries that are typically in operation (for at least 3 days per week)	Questionnaire

**TABLE 4** Practice characteristics recorded (via questionnaire)

Variable name	Variable description
Corporation	Practice is a dental body corporate
Good Practice Award	Practice has been awarded the British Dental Association's Good Practice Award
Material costs	The approximate percentage of expenditure from total practice revenue spent on consumables
Laboratory costs	The approximate percentage of expenditure from total practice revenue spent on laboratory fees
Treatment time	How long the practice tends to book for regular and new patient examinations (minutes)
Proportion of DCP to GDP sessions seeing NHS patients	The percentage of DCP sessions seeing NHS patients to GDP sessions seeing NHS patients
Weekly PP sessions with NHS patients	The number of sessions worked by PPs seeing NHS patients in a typical week
Weekly associate GDP sessions with NHS patients	The number of sessions worked by associate GDPs seeing NHS patients in a typical week
Weekly DH sessions with NHS patients	The number of sessions worked by DHs seeing NHS patients in a typical week
Weekly DHT sessions with NHS patients	The number of sessions worked by DHTs seeing NHS patients in a typical week
Weekly DT sessions with NHS patients	The number of sessions worked by DTs seeing NHS patients in a typical week
Any use of DCPs with NHS patients	Practice/site employs DCPs for use on NHS patients
Any use of DCPs (with either NHS or private patients)	Practice/site employs DCPs
Number of FTE DCPs employed (to work with NHS or private patients)	FTE number of DCPs the practice employs
Weekly support staff sessions	The weekly number of administration staff sessions for GDS practices
FTE, full-time equivalent.	

TABLE 5 Patient population and service mix characteristics (via Business Services Authority)

Variable name	Variable description
Percentage of child patients	Percentage of patients treated that are children (aged 0–18 years)
Proportion of adult non-elderly patients	Proportion of total patients treated aged 19–59 years
Proportion of exempt patients	The percentage of patients treated when a patient was exempt
Proportion of patients with non-age related exemption from NHS charges status	The proportion of patients treated when a patient was exempt from NHS charges status for the following reasons: expectant or nursing mothers, income support, job seeker's allowance, working family tax credits and certificates for full or partial help with health-care costs
Proportion of band 1 treatments	The proportion of annual NHS claims for routine treatment (band 1)
Proportion of band 2 treatments	The proportion of annual NHS claims for routine treatment (band 2)
Proportion of band 3 treatments	The proportion of annual NHS claims for complex treatment (band 3)
Proportion of urgent treatments	The proportion of annual NHS claims for urgent treatment
Proportion of preventative treatments	The proportion of annual NHS claims that include preventative treatments
Proportion of examinations	The proportion of annual NHS claims for examinations
Proportion of fillings and restorations	The proportion of annual NHS claims for routine restorations

# Data envelopment analysis and the analysis of efficiency scores

The data were analysed in a two-stage process. In stage 1, DEA was used to compute the technical efficiency scores of each practice. Efficiency estimates were bootstrapped to improve statistical accuracy in stage 2, where practice characteristics were regressed on the technical efficiency scores obtained in stage 1 using ordinary least squares (OLS) regression to determine whether or not practice efficiency was significantly associated with DCP use and role substitution after controlling for other practice characteristics. What follows is an overview of the methods for the first and second stages. This includes the approach taken to account for scale inefficiencies, how we arrived at a decision to choose an input-orientated model, the methods taken to overcome problems of correlation among the DEA estimates and correlation of the inputs and outputs used in the first stage with second-stage 'environmental' factors.

Data envelopment analysis is a linear programming technique that estimates the relative efficiency of a particular decision unit (i.e. GDS practice) is identified by comparing it to the performance of another decision unit rather than trying to associate performance with statistical averages that may not be applicable. DEA constructs a convex hull around a set of outlaying observations based on best observable practice. The approach is non-parametric because the location and shape of the efficiency frontier is anchored to observations rather than estimated from parameters such as the level of inputs practices have used. The distance of a practice unit from the frontier is a measure of technical inefficiency. Technical efficiency is a comparative measure of how well a decision-making unit deploys inputs to produce outputs, as opposed to the production possibility frontier, which is the maximum output that can be achieved for a particular level and mix of inputs. As the DEA efficiency frontier is based on best observable practice, it is only an approximation of the true unobserved efficiency frontier. In other words, it can tell how efficient a practice is compared with its peers but not compared with a theoretical maximum.

The advantage of DEA is that it can accommodate the multiple inputs and multiple outputs of dental health-care production more easily than regression analysis,<sup>33</sup> where non-biased estimation requires parameters to have specific distributional forms (usually normally distributed) and there is the potential for correlation-based problems such as multicollinearity.<sup>122</sup>

Inputs used for GDS practices were the number of surgeries and staff time worked (GDP sessions with NHS patients, DCP sessions with NHS patients, dental nurse sessions, administrative and managerial staff sessions). Sensitivity analysis was used to test alternative specifications that (1) include consumable materials and laboratory fees as a percentage of total practice revenue as inputs and (2) exclude staff groups that are not involved directly in patient care (administrative staff and nurses).

The output of the practice was measured by UDAs. Two other health-care outcomes were used to assess the sensitivity of results to the choice of the outcome measure. These were the number of treatment plans and the number of unique patients seen. Standard economic techniques were used to calculate missing values of input variables for GDS practices with a missing value (six GDS practices from all samples) with the predicted mean calculated from the values of all other input variables.

The sample efficiency estimates are bootstrapped to correct for bias that exaggerates the level of efficiency within a sample.<sup>123</sup> The bootstrapped DEA is a common approach in the research literature and enables consistent inference within the models explaining the efficiency estimates, while simultaneously producing standard errors and confidence intervals for these estimates.

In the second stage of the analysis, the efficiency scores were regressed onto practice variables. If efficiently operating practices have common characteristics, this allows for identification of possible correlates of inefficiency.

To answer the first three RQs, the variation in dental practice/site use of DCPs was measured in terms of whether or not a practice employs any DCPs for use on NHS patients, the number of sessions worked on

NHS patients in a typical week and, to understand how established DCPs are within the dental team, the relative frequency of DCP sessions on NHS patients compared with GDP sessions on NHS patients.

The following practice characteristics served as control variables in the model: number of sessions per week worked by GDPs on NHS patients (this variable is not present in the model specification when the percentage of DCP sessions to GDP sessions variable is included, to avoid correlation between the terms), the number of support staff sessions per week (managerial and administration staff for GDS practices), the number of surgeries in each practice, the proportion of adult (non-elderly) patients treated (aged 19–59 years) and the proportion of patients who are exempt from NHS payment charges (i.e. exemption for expectant or nursing mothers, income support, job seeker's allowance, working family tax credits and certificates for full or partial help with health-care costs).

# Sensitivity analysis

Sensitivity analysis was conducted with alternative specifications for different input and output variables to explore if findings are sensitive to the choice of these variables.

Alternative outputs of dental health production were considered because the output of health care, which practices focus attention on and arrange their services to achieve (e.g. NHS remuneration), might differ from what patients use to select a dental practice, or what a trust uses to assess practice performance. Other stakeholders (such as the Department of Health) may have other performance indicators, for example patient access to treatment where needed. Patient access per se cannot be measured, but the reach of a practice to patient groups could be proxied by the number of unique patients seen. In comparison to a measure of the number of UDAs or NHS income received, the number of treatment plans (claims) might be a more accurate measure of the volume of care delivered to patient populations because the distribution of efficiency scores will not be skewed by practices that deliver a relatively small (large) number of treatments of relativity high (low) complexity and remuneration.

Sensitivity analysis was also conducted with consumable materials and laboratory fees included as non-labour inputs in production. <sup>124</sup> The sensitivity analysis contributes to understanding whether or not findings are robust to the inclusion of a greater range of inputs. Consumable materials and laboratory fees are measured as a percentage of total practice revenue, with an increase in this ratio resulting in a decrease in efficiency. This was included in the model as part of a sensitivity analysis because the questionnaire asked respondents for approximate values of these ratios, which might have introduced measurement error. However, the ratio of materials or laboratory costs to total revenue can rise or fall without a change in the quantity or cost of materials and laboratory fees when the practice income of delivering a treatment changes independently of material usage. This could have occurred if greater income was received from delivering identical treatments to private rather than NHS patients, with private patients benefiting from a reduction in waiting time in their treatments rather than a change in quality or quantity of the materials or laboratory equipment used. This means that the results from estimation with capital stock variables should be interpreted with care as it is not known whether a change in results is the result of an improvement in information (i.e. using the full range of resources used) or a loss of information from the use of imprecisely measured variables.

Following Scheffler and Kushman,<sup>124</sup> essential inputs are those where output would be zero if they are not present (i.e. surgeries and dental staff); the two non-essential inputs excluded from the analysis were the weekly number of dental nurse sessions and administration and managerial sessions worked. DCP sessions worked on NHS patients are considered an essential input because unlike nurses and administrators, they deliver care that accrues UDAs/income to a practice.

#### Stochastic frontier model approach to estimating efficiency

A limitation of DEA is that it assumes that if a NHS dental practice can produce a certain level of output using specific input levels, another NHS dental practice of equal scale should be capable of doing the same. This might not be the case if, for example, the frontier of most efficient practices is reached because

of unique local contextual circumstances (e.g. a practice has established itself as being able to employ or refer to one of a limited number of specialist GDPs) or because of measurement error in the values of input or output variables.

Stochastic frontier modelling helps overcome this limitation by estimating the frontier with statistical regression methods rather than fixing the frontier from extreme point observations. However, the SFM approach has limitations. In particular, efficiency scores (as measured by regression residuals) reflect a combination of relative efficiency, measurement error in the dependent variable, and statistical noise.<sup>33</sup> Moreover, it does not calculate a productive frontier that corresponds to the theoretical notion of a production function, but instead estimates a fitted 'average' function that provides no direct quantitative information on productive inefficiency in the sample.<sup>33</sup> Nevertheless, SFM was used in this study to test the robustness of the DEA findings for RQ1 and RQ2. A Cobb–Douglas function form was used for the SFM after a more flexible functional form (Translog) was rejected in a joint chi-square test of the coefficients on the additional Translog variables.<sup>33</sup>

# Sample size

The most important sample consideration is the selection of homogeneous sample observations that would achieve a level of efficiency of those practices on the frontier if they adopted the same level and combination of inputs and techniques (e.g. organisational or technological processes of production) for transforming inputs into outputs. For this reason, we limit the main analysis to practices in England operating under the same NHS remuneration arrangements and do not combine samples from different jurisdictions and service types.

The relative nature of DEA (i.e. estimating technical efficiency relative to the best-performing practice on the 'frontier') makes it vulnerable to problems with degrees of freedom. The number of degrees of freedom increases with the number of NHS dental practices in the data set and decreases with the number of input and output variables. Banker *et al.* <sup>126</sup> suggest a rough rule of thumb for acceptable sample size: (1) it should be larger than the product of the number of inputs and number of outputs and (2) it should be at least three times as large as the sum of the number of inputs and the number of outputs. Dyson *et al.* <sup>121</sup> suggested a sample of at least twice the total number of input and output variables. Brown<sup>127</sup> suggested the sample should exceed three times the sum of the number of inputs and the number of outputs. The sample size of DEA models in all jurisdictions and service types estimated in this study comfortably meets all of these criteria.

In this study, conclusions were drawn from the analysis of NHS dental practices located in England (with the remaining analyses for Scotland and Northern Ireland being described in *Appendix 1*). The England sample has over four times as many observations as practices in other jurisdictions. A large number of observations enables a sharper identification of typical relations between inputs and outputs so that more inputs and more outputs can be incorporated into a DEA.<sup>124</sup>

# **Results**

# Sample of NHS dental practices in England

Table 6 compares the sample of 121 practices with data collected from the Business Services Authority for all of England to test how generalisable the study sample was. The mean practice output of UDAs was higher among sample practices than for England (15,444 compared with 10,494), as was the mean number of patients seen (5597 compared with 3670) and treatment plans (7286 compared with 4701). This may be explained by the sample practices being larger (on average 5.7 active GDPs compared with 3.9 in England). Despite this difference in scale, the distribution of claims across pay bands and the percentage of UDAs claimed from treating patients exempt from payment charges in sample practices were both similar to all practices in England. However, the adult share of practice patients was greater for sample practices (mean of 73.6%) than all England practices (mean of 65%).

TABLE 6 Sample practices in England compared with national data

Practice characteristics	England, mean ( <i>n</i> = 8788)	Sample, mean ( <i>n</i> = 121)
Annual number of UDAs	10,494	15,444
Annual number of patients seen	3670	5597
Annual number of treatment plans (claims)	4701	7286
Percentage of UDAs from adults	65	73.6
Percentage of UDAs from patients aged 40 years and over	41	52.7
Percentage of UDA from patients exempt from patient charges	41	37.3
Percentage of claims that are band 1	54	57.3
Percentage of claims that are band 2	30	28.4
Percentage of claims that are band 3	6	4.7
Percentage of claims that are urgent	9	9.6
Active performers at the practice	3.9	5.7

Table 7 presents descriptive statistics of practice characteristics for practices that used DCPs on NHS patients and practices that did not use DCPs. The larger size of practices with DCPs (3.82 dental surgeries compared with 2.8 for practices without DCPs) suggests that DCPs may have been used to increase practice capacity (i.e. to supplement capacity rather than substitute for dentists while maintaining capacity). For all other practice characteristics there is no statistically significant difference between the two practice groups.

Tables 8–10 present the mean values of the variables that were collected from the sample of NHS dental practices in England (corresponding data for Scotland and Northern Ireland practice samples are provided in *Appendix 1*).

#### Data envelopment analysis

Table 11 presents the estimated mean level of practice performance from the sample in England for the main outcome measure (UDAs) and secondary output measures (treatment plans, patients seen). NHS dental practices that perform with full technical efficiency (i.e. producing on the frontier) have an efficiency score of 100. Efficiency scores were estimated after bootstrapping with 1000 replications. The corresponding estimates for Scotland and Northern Ireland are provided in *Appendix 1*.

Practices were operating at a mean level of efficiency of 64%. Practices without role substitution (no DCPS employed) had a mean level of efficiency of 68% (n = 39) and 62% (n = 82) for those with role substitution. The mean efficiency score changes little when practice performance is measured in terms of number of patients or treatment plans. The variation in practice scores around this mean is small, with 95% of scores lying within 5 percentage points of the mean. Subgroup analysis of the efficiency scores revealed the greatest variation in efficiency scores to be among those practices that did not use role substitution (95% of scores lying within 12 percentage points of the mean). This may be an indication of greater variation in operating procedures among practices that do not use role substitution or it could be that operating procedures become more similar by the use of DCPs (e.g. those practices may shift their service mix towards the smaller range of treatments that DCPs are able to perform).

The technical inefficiencies can arise from the ineffective operation of the production process in organising inputs and/or transforming inputs to outputs after controlling for scale effects. Efficiency scores were estimated after bootstrapping with 1000 replications.

TABLE 7 Descriptive statistics of practices that did and did not use DCPs

	Mean number of	Mean number of practices	
Practice characteristics	With no DCPs (n = 39)	With DCPs (n = 82)	p-value of difference in means
Dental health-care outputs			
Annual number of UDAs	15,769	15,289	0.83
Annual number of courses of treatments (claims)	7168	7342	0.86
Annual number of patients seen	5513	5637	0.90
Staff sessions			
Weekly number of GDP sessions with NHS patients	17.7	18.7	0.72
Weekly number of PP sessions with NHS patients	4.7	6.7	0.10
Weekly number of associate GDPs sessions with NHS patients	12.6	10.7	0.42
Weekly number of administrative staff sessions	19.3	23.2	0.19
Practice characteristics			
Number of surgeries	2.8	3.82	< 0.00
Consumable materials as a percentage of practice revenue	12.3	11.5	0.63
Laboratory expenditure as a percentage of practice revenue	9.0	10.0	0.51
Treatment mix			
Percentage of band 1 treatments	57.7	57.1	0.77
Percentage of band 2 treatments	29.8	27.8	0.16
Percentage of band 3 treatments	5.1	4.5	0.38
Percentage of urgent treatments	7.4	10.6	0.12
Patient case mix			
Percentage of adult patients treated	76.2	72.4	0.34
Percentage of copayment exempt patients	42.3	34.9	0.12
Percentage of patients with non-age-related exemption from NHS charges status	25.4	24.2	0.64

TABLE 8 Sample statistics of NHS dental health-care output measures

Measure	Detail	England, mean (95% CI)
UDAs	UDAs accrued in 2013/14 for general dentistry	15,444 (13,433 to 17,455)
Treatment plans	Number of submitted claim forms in 2013/14 for general dentistry	7286 (6367 to 8205)
Patients seen	Number of unique patients seen in 2013/14 for general dentistry	5597 (4905 to 6288)
CI, confidence interva	al.	

**TABLE 9** Sample statistics of staffing variables

Measure	Detail	England, mean (95% Cl)
Weekly GDP sessions with NHS patients	Number of sessions worked by GDPs seeing NHS patients in a typical week	19.4 (17 to 21.9)
Weekly PP sessions with NHS patients	The number of sessions worked by PPs seeing NHS patients in a typical week	6.1 (4.9 to 7.1)
Weekly associate GDP sessions with NHS patients	The number of sessions worked by associate GDPs seeing NHS patients in a typical week	11.3 (9.1 to 13.4)
The percentage of DCP to GDP sessions seeing NHS patients	The percentage of DCP to GDP sessions seeing NHS patients	16.6 (12.6 to 34.6)
Any use of DCPs with NHS patients	Percentage of the whole sample that uses DCPs with NHS patients	68
Number of FTE DCPs employed	The number of WTE DCPs employed (NHS and private work)	1.1 (0.8 to 1.5)
Weekly DCP sessions with NHS patients	Number of sessions worked by DCPs seeing NHS patients in a typical week	3.0 (2.1 to 3.8)
Weekly DH sessions with NHS patients	Number of sessions worked by DHs seeing NHS patients in a typical week	0.6 (0.3 to 0.8)
Weekly DHTs sessions with NHS patients	Number of sessions worked by DHTs seeing NHS patients in a typical week	1.7 (0.9 to 2.5)
Weekly DTs sessions with NHS patients	Number of sessions worked DTs seeing NHS patients in a typical week	0.7 (0.3 to 1.0)
Weekly nurse sessions	Number dental nurse sessions worked in a typical week	47.0 (42.1 to 52.0)
Weekly administration staff sessions	The number of sessions worked by managers and administrative staff in a typical week	20.8 (18.3 to 23.3)
CI, confidence interval; FTE, full-time	e equivalent.	

TABLE 10 Sample statistics of practice/site variables

Management	D. C. II	England, mean
Measure	Detail	(95% CI)
Surgeries	The number of surgeries that are typically in operation (for at least 3 days per week)	3.5 (3.2 to 3.8)
Percentage of child patients	Percentage of patients treated that are children	26.4 (22.7 to 30.1)
Percentage of exempt patients	Percentage of patients that are exempt from patient charges	37.3 (32.9 to 41.7)
Percentage of band 1 treatments	Percentage of UDAs that are band 1	57.3 (55.1 to 59.5)
Percentage of band 2 treatments	Percentage of UDAs that are band 2	28.4 (27.1 to 29.7)
Percentage of band 3 treatments	Percentage of UDAs that are band 3	4.7 (4.1 to 5.3)
Percentage of urgent treatments	Percentage of UDAs that are urgent	9.6 (7.7 to 11.6)
Good Practice Award	Percentage of practices that have been awarded the British Dental Association's Good Practice Award	33
Material costs	Percentage of total practice revenue spent on consumables in 2013/14	11.7 (10.2 to 13.2)
Laboratory costs	Percentage of total practice revenue spent on laboratory fees in 2013/14	9.7 (8.3 to 11.1)
CI, confidence interval.		

TABLE 11 Mean practice efficiency score when health-care output is measured by UDAs, treatment plans and patients seen

Output measure	Mean <sup>a</sup>	95% CI
Annual number of UDAs	0.64	0.60 to 0.69
Annual number of treatment plans (claims)	0.63	0.58 to 0.67
Annual number of patients seen	0.62	0.57 to 0.67

CI, confidence interval.

Research question 1: are NHS dental practices that employ DCPs significantly more technically efficient than practices that do not employ DCPs after controlling for other practice and patient population characteristics?

Table 12 describes the association with efficiency scores of any use of DCPs for different output measures. Any use of DCPs with NHS patients is found to be associated with statistically significant lower efficiency (14 percentage points lower for UDAs, 11 percentage points lower for treatment plans or patients seen) than practices with no DCPs.

Research question 2: among NHS dental practices that employ DCPs, is the level of technical efficiency significantly associated with the level of DCP use after controlling for other practice and patient population characteristics?

The association between efficiency score and the level of use of DCPs with NHS patients among those practices that employ DCPs is presented in *Table 13*. There is no significant association between efficiency and the number of DCP weekly sessions worked with NHS patients after controlling for patient and practice characteristics.

TABLE 12 Association with efficiency scores of any use of DCPs with NHS patients for different health-care output measures

	UDAs		Treatment plans (claims)		Patients seen	
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
Any use of DCPs with NHS patients	-0.14	< 0.00	-0.11	< 0.00	-0.11	< 0.00
Weekly GDP sessions with NHS patients	-0.001	0.76	-0.001	0.54	-0.002	0.21
Weekly support staff sessions	-0.005	< 0.00	-0.006	< 0.00	-0.006	< 0.00
Number of surgeries	-0.04	0.02	-0.05	< 0.00	-0.05	< 0.00
Proportion of adult non-elderly patients	-0.19	0.09	-0.15	0.11	-0.16	0.09
Proportion with non-age-related exemption	-0.13	0.34	-0.38	0.01	-0.29	0.04

a DEA requires the selection of the 'orientation' to be adopted for estimating technical efficiency scores. An input-orientated model estimates the amount by which inputs can be proportionally reduced, without loss of outputs. Output orientation is an estimation of the amount by which outputs can be proportionally expanded, without additional inputs (Thanassoulis<sup>122</sup>). Selection of orientation depends on the precise RQ. In this analysis, outputs were constrained.

TABLE 13 Association with efficiency scores of the level of use of DCPs with NHS patients for different health-care output measures

	UDAs		Treatment plans (claims)		Patients seen	
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
Weekly DCP sessions with NHS patients	-0.004	0.51	-0.005	0.35	-0.0009	0.85
Weekly GDP sessions with NHS patients	0.001	0.73	0.001	0.79	-0.0008	0.68
Weekly support staff sessions	-0.005	< 0.00	-0.006	< 0.00	-0.006	< 0.00
Number of surgeries	-0.04	0.05	-0.04	0.03	-0.05	< 0.00
Proportion of adult non-elderly patients	-0.21	0.11	-0.16	0.16	-0.17	0.12
Proportion with non-age-related exemption	-0.09	0.58	-0.32	0.08	-0.26	0.15

Research question 3: among NHS dental practices that employ DCPs, is the level of technical efficiency significantly associated with the proportion of clinical time provided by DCPs?

The association between efficiency score and the proportion of clinical time provided by DCPs is presented in *Table 14*. The ratio of DCP weekly sessions worked with NHS patients to GDP sessions worked with NHS patients is not significantly associated with any of the three output measures after controlling for patient and practice characteristics.

Research question 4: among NHS dental practices that employ DCPs, is the association between technical efficiency and DCP use dependent on (a) the proportion of the practice population that is made up of adult non-elderly patients (who on average are more likely to have routine dental care needs) and (b) the proportion of the practice population made up of patients with non-age-related exemption from NHS charges status (who on average are more likely to have non-routine dental care needs)?

Table 15 presents the estimated coefficients on the interaction term between the level of DCP use and the characteristics of the patient population. No significant interactions with patient population characteristics were found indicating that the association between DCP use and efficiency is not related to the characteristics of the practice population being served.

**TABLE 14** Associations with efficiency scores of the proportion of clinical time provided for different health-care output measures

	UDAs		Treatment plans (claims)		Patients seen	
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
The percentage of DCP to GDP sessions seeing NHS patients	0.0005	0.09	0.0005	0.07	0.001	0.06
Weekly GDP sessions with NHS patients	-0.005	< 0.00	-0.006	< 0.00	-0.006	< 0.00
Weekly support staff sessions	-0.04	< 0.00	-0.04	< 0.00	-0.05	< 0.00
Number of surgeries	-0.17	0.10	-0.12	0.15	-0.17	0.04
Proportion of adult non-elderly patients	-0.14	0.35	-0.40	0.02	-0.30	0.06
Proportion with non-age-related exemption	0.0005	0.09	0.0005	0.07	0.001	0.06

TABLE 15 Mediating associations with efficiency scores between the level of DCP use and patient case-mix variables

	UDAs		Treatment plans (claims)		Patients seen	
Mediating variable (interaction term)	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
Proportion of adult non-elderly patients	0.08	0.10	0.05	0.24	0.06	0.06
Proportion with non-age-related exemption	0.03	0.21	0.05	0.08	0.05	0.05

Each row is a separate regression that includes the following control variables: weekly NHS GDP sessions, weekly administration staff sessions, number of surgeries, proportion of adult non-elderly patients and proportion of patients with non-age-related exemption from NHS charges status.

Research question 5: among NHS dental practices that employ DCPs, is the proportion of claims for routine services significantly associated with the level of DCP use after controlling for other practice and patient population characteristics? Among practices that employ DCPs, is the proportion of claims for routine services significantly associated with the proportion of clinical time provided by DCPs after controlling for other practice and patient population characteristics?

The association between routine services and the level of use of DCPs and the association between routine services and the proportion of clinical time provided by DCPs are presented in *Table 16*. The only significant association found was between the proportion of preventative treatment and the proportion of clinical time provided by DCPs. One more weekly session of DCP activity for NHS patients per GDP session with NHS patients is associated with a 0.36 percentage point higher proportion of preventative treatments delivered after controlling for practice and patient characteristics. Therefore, practices that employed more DCPs as a proportion of total labour inputs delivered a greater proportion of preventative treatments.

# Sensitivity checks on the estimation of efficiency scores

Table 17 describes the associations of DCP use and efficiency scores with alternative specifications of inputs into the production function. The findings were robust to (1) the exclusion of nurses and administrative staff time and/or (2) the inclusion of material and laboratory expenditure as a percentage of practice revenue. As with the findings reported above, any use of DCPs was negatively associated with efficiency scores.

TABLE 16 Associations between the numbers of routine services delivered in the NHS and the level of DCP use on NHS patients

Outcome variable (routine services)	Coefficient on the level of DCP use on NHS patients	<i>p</i> -value	Coefficient on the proportion of DCP's clinical time relative to GDP's clinical time working with NHS patients	<i>p</i> -value
Proportion of band 1 treatments	-0.0001	0.99	-0.00003	0.86
Proportion of band 2 treatments	0.001	0.51	-0.00001	0.90
Proportion of preventative treatments	2.46	0.10	0.36	0.00

Each row is a separate regression that includes control variables: weekly NHS GDP sessions, weekly administration staff sessions, number of surgeries, proportion of adult non-elderly patients and proportion of patients with non-age-related exemption from NHS charges status.

TABLE 17 Alternative specifications of inputs into the productive efficiency frontier

	Without nur administrati production i	on as a	With materia and laborato expenditure a percentage o revenue as a	ry as a f practice	Without nurses a administration as production input material and labo expenditure as a percentage of pro revenue as an ing	a and with oratory actice
DCP utilisation variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
In the sample of practices that e	employ and do	not employ	DCPs			
Any use of DCPs with NHS patients	-0.12	< 0.00	-0.13	< 0.00	-0.13	< 0.00
Any use of DCPs (on either NHS or private patients)	-0.10	0.03	-0.13	< 0.00	-0.11	0.02
Number of FTE DCPs employed (to work with NHS or private patients)	0.02	0.06	0.001	0.85	0.002	0.84
In the sample of practices that e	employ DCPs wi	ith NHS pat	tients			
Weekly DCP NHS sessions	0.0001	0.97	-0.005	0.37	-0.006	0.30
The percentage of DCP to GDP sessions seeing NHS patients	0.001	0.06	0.0002	0.58	0.0004	0.21

FTE, full-time equivalent.

Sample is GDS practices located in England. Efficiency scores were calculated with an input-orientated approach and with UDAs as the outcome measure. Each row is a separate regression. Each regression includes as control variables the number of NHS GDP sessions, number of support staff sessions, number of surgeries, proportion of the case mix that are adults and proportion of exempt patients. The regression examining the percentage of DCP to GDP sessions seeing NHS patients does not include the number of GDP sessions variable as that is incorporated as part of the percentage of DCP to GDP sessions seeing NHS patients measure.

# Stochastic frontier modelling

Table 18 describes the results of the SFM. Only the number of GDP sessions on NHS patients is statistically significant at a 5% level, with a 10 percentage point increase in the number of GDP sessions on NHS patients being associated with a 7.5 percentage point increase in the level of annual UDAs. The SFM efficiency scores had a lower mean than DEA efficiency scores by 0.06, which is expected because measurement error/statistical noise in the dependent variable is excluded from SFM values.<sup>33</sup> Correlations between efficiency scores estimated by SFM and DEA were greater than 0.6, which supports the robustness of estimated DEA efficiency scores and the internal validity of the approach.<sup>124</sup>

# Impact of different remuneration systems on the technical efficiency of role substitution models

The number of input questionnaires returned from Northern Ireland (n = 29) and Scotland (n = 20) was limited (when compared with England) and so it is difficult to make robust comparisons across the different countries. As a result, the DEA analyses for Northern Ireland and Scotland are presented in Appendix 1. NHS practices in Northern Ireland and Scotland operate within a FFS remuneration system and so, unlike England, there are no constraints on the level of clinical activity (outputs) that can be produced by any one dental team. Although direct comparisons are problematic given the smaller sample sizes, the results suggest that English NHS dental practices operate at a lower level of efficiency (64%) than practices in Northern Ireland (80%) and Scotland (94%). The efficiency scores changed little when practice performance was measured by the effectiveness of the practices in reducing health-care inputs to see a fixed number of patients or treatment plans (output orientation).

TABLE 18 Stochastic frontier modelling approach to estimating efficiency scores

	Output measures							
	Logarithm of annual number of UDAs		Logarithm o number of to courses		Logarithm of annu number of patient seen			
Input variables	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value		
Logarithm of weekly DCP sessions with NHS patients	-0.007	0.36	-0.002	0.76	-0.005	0.54		
Logarithm of weekly GDP sessions with NHS patients	0.75	< 0.00	0.67	< 0.00	0.67	< 0.00		
Logarithm of weekly nurse sessions	0.03	0.57	0.014	0.75	800.0	0.85		
Logarithm of weekly administration staff sessions	0.04	0.26	0.03	0.33	0.03	0.37		
Logarithm of number of surgeries	0.06	0.75	0.13	0.44	0.19	0.28		
Observations	121							

The variation in practice efficiency was also greater in FFS remuneration systems, with 95% of scores lying within 12 (Scotland) and 6 (Northern Ireland) percentage points of the mean, compared with 5 percentage points for practices in England. This difference might have been caused artificially by the unitary measure used for health-care output in Scotland and Northern Ireland, compared with England (pounds sterling vs. UDAs). If pounds are a more sensitive metric of NHS income received, this could explain the greater variation in estimated efficiency. However, the higher mean efficiency scores and greater dispersion of efficiency scores in Northern Ireland and Scotland might be explained by the absence of an ACV (no output constraint). This allows those NHS practices with a given set of inputs to maximise their technical efficiency, expanding their production until their inputs are fully utilised. This may be a particularly effective way to improve efficiency in dental health care in a FFS remuneration system when inputs are relatively fixed or difficult to vary in the short term (e.g. number of surgeries, number of staff). Of note, the number of surgeries was not statistically associated at a 5% level with efficiency scores in Northern Ireland and Scotland, whereas in England it had a negative association (at a 5% level). This suggests that economies of scale are not present in NHS primary dental care in Northern Ireland and Scotland, whereas diseconomies of scale are present in England.

There was no statistically significant association between the use of DCPs and efficiency scores in Northern Ireland and Scotland, whereas in England the association was negative (see *Table 28, Appendix 1*). This might be because practices in Northern Ireland and Scotland are using the full scope of practice of their DCPs. However, the small sample sizes in Northern Ireland and Scotland and the limited use of DCPs (0.2 and 0.3 full-time equivalent DCPs, respectively), compared with 1.1 DCPs per practice in England (see *Table 24, Appendix 1*), would suggest that this is unlikely. The small sample size may also explain why efficiency scores in Scottish practices are not significantly associated with any practice environment characteristics or staffing variables regardless of the model specification of the role substitution measure used (see *Tables 28–38*, in *Appendix 1*).

#### **Discussion**

#### Main findings

The number of studies that have examined technical efficiency in primary dental care is limited. 124,128,129 Gutacker *et al.* 128 found that there were no diseconomies of scale related to capital or labour inputs into the production of efficiency in Australia. A consistent finding from Gutacker *et al.*, 128 Gray 129 and Scheffler and Kushman 124 was that GDPs' own labour input was the most important determinant of their output.

The results from this study show that many NHS dental practices in England were not performing as efficiently as they might (relative to the most efficient practice in the sample). This finding may be influenced by the cost-containment mechanism within the 2006 NHS dental contract in England.<sup>45</sup> Output constraints within this contract place a ceiling on the number of UDAs that can be generated within any one financial year. Given that the number of surgeries and GDPs working in NHS dental practices are relatively inflexible inputs, the inability of PPs to adjust the team size and composition as the contracted deadline for the target output is approached might explain the inefficiency in production. The analyses for the smaller sample from Northern Ireland who were paid on a FFS basis (see *Appendix 1*), and hence faced no output constraints, showed that practice efficiency was higher than for England.

This is the first study to examine the impact of role substitution in primary dental care. Practices with a lower use of DCPs were found to operate at higher levels of technical efficiency, after controlling for staff, practice environment and patient variables. This suggests, that where practices employ DCPs, they do not appear to contribute to higher volumes of NHS dental activity. This finding did not vary with any of the sensitivity tests undertaken. The correlation between the DEA- and the SFM-estimated levels of efficiency was greater than 0.6, which supports the robustness of the estimated DEA efficiency scores and the internal validity of the methodological approach.

The cross-sectional basis of the analysis cannot determine the precise mechanisms underlying the findings or infer causality. Caution should be exercised in the interpretation of these results as the lower efficiency scores among NHS dental practices that utilise DCPs could be caused by the way the DCPs were deployed and structural constraints of the remuneration system. In the current NHS dental contract in England, there is a disincentive to refer work in practices that employ associate GDPs, because the GDP loses income for activities referred on to other providers. This may lead to DCPs not being used to their full scope of practice and being used in ways that do not substitute for more expensive dentist time. As a result, they may represent a relatively costly input for the practice. These factors are explored in *Chapter 5*.

The findings could also be caused by the limited scope of the efficiency measure. This study has examined the efficiency of NHS dental practices, not individual DCPs (as DCPs do not directly contract with the NHS). It could be that DCPs are being used efficiently and in accordance with their full scope of practice, but their use is associated with a reduction in efficiency of GDP time. For example, GDPs may not use the time freed up by DCPs in ways that accrue UDAs for the practice (as noted above, this may be the result of the output targets inherent in the NHS contract). They might instead undertake more private dentistry, such that efficiency gains in the NHS are lost to the private sector. Alternatively, they may increase the length of appointment times, take on new managerial or administrative duties or reduce the effort and intensity of their own work within the NHS.

Ideally, efficiency would incorporate considerations of the quality of products and services in terms of impact on patient outcomes (e.g. oral health). Lower efficiency may be a reflection of higher-quality care that produces better patient outcomes by, for example, taking more time with patients. Further exploration of this line of enquiry would require data on quality of care.

Finally, lower efficiency scores might be explained by other variables that were not measured in the analysis, such as lower skills and experience of dental staff.

# **Study limitations**

This study was based on a sample of 121 general dental practices located in the north of England (with corresponding analyses on smaller samples of dental practices in Scotland and Northern Ireland; see *Appendix 1*). The sample was selected from practices that replied to a postal questionnaire. Although the economic model included control variables for characteristics of the practices and their patient populations, this is not adequate to control for practice site self-selection to the study. Therefore, it is not possible to draw conclusions on the associations between DCP utilisation and technical efficiency in other service settings, especially as the DEA measures efficiency relative to the best practice observed within a study sample.

A limitation of the economic model is that inputs were considered only in terms of hours of work. Low response rates to the initial pilot survey prevented the research team from including many non-labour operating expenditures (establishment costs, marketing expenses, administration and general expenses), rent and interest payments and expenditure on physical capital. The model does not include expenditures on materials, consumables and laboratory supplies among the inputs in the estimation of the production frontier, therefore it is not possible to estimate cost efficiency. However, consumable materials and laboratory costs were included as a proportion of total practice revenue as an input variable to estimate production functions of health-care outputs. We found that there was no change in previously observed associations with efficiency scores. Another limitation is that the model assumes that practices have staff with the same levels of qualifications and experience, etc. However, the quality of labour input may vary depending on individual skills, professional experience within each type of staff. Finally, we observe only the quantity of DCP use, not how DCP time is used in the practice (i.e. according to their full scope of practice or not).

A shortcoming of the DEA approach used is that it assumes that there is no measurement error and, therefore, practices with 'noise' may end up being identified as technically efficient. For example, if one practice understated inputs or overstated outputs, it might be an outlier and hence significantly reduce the estimated (relative) efficiency of other practices. Each unit of data was screened for outliers. Missing data or outliers in data were addressed through a manual check of original records. When a participant had left a question blank, or answered with an uninterpretable, illegible or unlikely (i.e. an 'outlier') answer on the original record of the questionnaire, the practice was contacted to amend or clarify their answer. However, all judgements of what might be measurement error or an outlying observation in the data are necessarily arbitrary and imprecise. For example, a unit value might be within a normal range on a measurement instrument and not be flagged with having measurement error when its true value is very different from the one recorded. We had no opportunity to check the dental practices' output data against the actual claim forms that were submitted.

Finally, quantitative analyses cannot determine what factors lead practices to hire DCPs and delegate activities to them. The potential for practices to meet their NHS remuneration target using fewer resources by delegating more work to DCPs does not appear to have occurred. Reasons for this are explored in the following chapter.

# **Concluding remarks**

The technical efficiency of practices was found to be associated with many factors, including the practice's staffing capacity and ownership structure. We found no evidence of positive efficiency gains from the use of DCPs. However, this might reflect limitations in the way DCPs are currently being used in the NHS, as well as constraints imposed by the use of historically based UDA targets as the basis for contracted activity and a lack of incentive for associate GDPs to utilise DCP skills.

# **Chapter 5** Attitudes of patients and the NHS dental team to role substitution

#### Introduction

Role substitution in dentistry has traditionally lagged behind its use in medicine.<sup>6</sup> This may reflect struggles between NHS GDPs and DCPs regarding role boundaries and jurisdictions.<sup>70</sup> McDonald *et al.*<sup>102</sup> found that the acceptability of role substitution in the medical workforce relies on a number of factors, including distribution of rewards in relation to effort across the team, clarity around roles and responsibilities and personal relationships with colleagues, which raise issues of trust and respect. As highlighted in previous chapters in this report, many DCPs are not being used to their full scope of practice within NHS dentistry. It may also reflect the concerns of NHS dental teams about patients' attitudes to role substitution, although the evidence in the literature is mixed.<sup>93–95</sup> Taken together, there is a need to explore the views of patients and the different members of the dental team on role substitution to examine the formal and informal aspects of organisational life.

The objectives for this part of the research programme were:

- to undertake semistructured interviews with GDPs, DCPs and patients to explore barriers/enablers to the greater use of role substitution
- to undertake semistructured interviews with key policy-makers to determine their perspective on the greater use of role substitution in NHS dentistry.

#### **Methods**

Qualitative methods were used to explore these issues, as we needed to obtain an understanding of relationships and activities within practices and the ways in which these relate to skill mix patterns.

#### Recruitment

The initial stage of recruitment targeted practices that had completed the screening questionnaire (see *Chapter 3*). In addition, NHS GDPs and DCPs were approached via professional networks (e.g. the BSDHT). The second stage recruited participants from England, Scotland and Northern Ireland that had taken part in the first workstream (reported in *Chapter 4*). These were augmented by invitations to DCPs sent via social networks. The third stage used the efficiency data analysed in *Chapter 4* to target those practices that were considered to be the most and least efficient.

# Interview structure and context

The interview questions were designed to explore the potential barriers to, and facilitators of, role substitution in practice and were developed by the research team iteratively. Separate interview schedules were designed a priori for each of the groups (PPs, DCPs, patients). Input was sought from the patient representative with regard to the patient interview schedule.

All face-to-face interviews were held in private rooms within the dental practice, either a staff room or office, or an empty surgery. The interviews varied in length: those with the PPs ranged from 20 to 50 minutes and those with the DCPs ranged from 20 to 40 minutes. Interviews with patients were generally much shorter, lasting around 10–15 minutes each. This short interview time seemed to be caused by the generally low level of awareness patients had regarding the roles and duties of different members of the dental team.

# Data analysis

All data were collected between October 2013 and May 2015. Data collection and analysis were run concurrently. We collected and analysed data using a framework informed by the literature and sought to develop this framework as the findings emerged. This framework was designed to develop an understanding of the practice of role substitution, which was grounded in local and national contexts; to go beyond the particulars of local cases to generate and test theory in order to draw lessons for the NHS more generally; and to inform the international literature on role substitution in dentistry.

All interviews were audiorecorded and transcribed verbatim. Thematic analysis was carried out by group, that is, PPs, DCPs and patients were coded separately. A thematic analysis was undertaken. The transcripts were read by the researchers who undertook the interviews and two of the research team separately. Interpretations were then pooled and edited in the presence of all three researchers to produce the final version of the coding frame, with disputes being resolved using a majority voting system. Data analysis was undertaken concomitantly with the interviews.

#### Results

# Overview of sample

The practices in the north-west of England that were visited by the research team were heterogeneous. Consistent with the findings from *Chapters 3* and *4*, the utilisation of DCPs across the practices was highly varied. Some practices used DCPs to their full scope of practice and were enabling DCPs to undergo further training, whereas, in other practices, DCPs were more limited in the duties they carried out and were used primarily to complete routine scale and polishing.

The patients interviewed ranged in age from  $\geq$  30 to 75 years and were a broadly equal mix in terms of gender. All patients were attending the NHS dental practice for routine appointments either with a GDP or a DCP and were interviewed following their appointment in a private room within the dental practice. Further details of the interviews conducted are provided below (*Table 19*).

TABLE 19 Numbers of participants by type, country, mode of interview and recruitment

Participant type	PP	DCP	Patient
Total (n)	16	17	18
Gender (female : male)	5:11	15:2	_
Site			
England	12	16	18
Scotland	4	1	0
Mode			
Face to face	7	6	18
Telephone	9	11	0
Stage			
From questionnaire	7ª	6	18
Via professional/social networks	4	11	0
From DEA	7ª	0	0

a Two of the PPs were recruited as participants from the questionnaire and also appeared in the 10 top and bottom efficiency groups.

Quotations attributed to PPs or DCPs are referenced as PP or DCP, respectively, and followed by the interviewee number. When PP\_T or DCP\_T is referenced, this indicates that the interview was a telephone interview. When PP\_S or DCP\_S is referenced, this indicates the interview was with a PP or DCP based in Scotland. Patient quotations are referenced by the practice at which they were interviewed as well as interviewee number.

# Results: practice principals

# Perceptions of patient awareness and acceptability of dental care professionals

A number of PPs, mainly those not utilising DCPs to their full scope of practice, reported that they perceived patients as being less willing to see a DCP than a GDP. Some of the PPs referred to this as a reason for not utilising DCPs more in practice.

We did have a therapist who worked with us, but I don't think we got full use out of the therapist, 'cause I don't think that our patients wanted to see a therapist for having fillings done.

PP3

The reason given by this particular PP for patients not wanting to receive dental treatment from DCPs was that his practice was a fully private one; as a result of this, he explained that he did not think his patients would want to pay the same amount for treatment carried out by a less-qualified member of the dental team. Other PPs who were using DCPs in practice tended to refer to being concerned about patients' reactions to being referred to a DCP initially.

They are not so happy about a therapist doing a filling, because they think, I'm paying all this money, why doesn't the dentist do it?

PP5

When the patients have seen me for years and I go, do you mind if [the DCP] does your filling? Oh, I'm paying £100 for that filling. I want you to do it . . . I know you'll do a good job.

PP6

The last quotation from PP6, where he asks the patients 'do you mind?', shows that treatment by a DCP may be represented as inferior by some GDPs.

However, for some PPs, this perception appeared to change over time. PPs reflected on patients' initial hesitation and felt they had been unnecessarily concerned. This change of mind seemed to come about following a positive patient report about seeing a DCP.

Once they'd seen them and realised that they had this extra time, they had more chance to ask questions, whatever else, they felt more comfortable and then people warm to the idea.

PP1

Practice principals who used DCPs in their practices for hygiene or dental work focused on the benefits to the patient. The extra time that DCPs are able to spend with patients was frequently cited as a positive. This extra time was seen as enabling DCPs to provide a more patient-centred, caring, approach than could reasonably be provided by a GDP. PPs reported this to be a particular advantage for anxious patients and for children, noting that seeing a DCP was less intimidating than seeing a GDP.

It's because it's a soft entry . . . I'm going to see the hygienist, feels much like a more acceptable approach to them, because it's friendly, it's preventative, it's not fixing problems, which, even though it may be a check-up, that's the image I think a lot of people have in their minds.

PP7

Some PPs felt they had to pitch the referral in a positive way and explain why certain treatments would be conducted by a DCP. One way to convince the patient and make the referral acceptable was to describe the benefit to patients, in particular the extra time allocated to patients when seeing DCPs.

The patients say they feel very rushed when they see me, I say that's because you're one of 60. If you want to have proper advice and toothbrush instruction, diet, oral health, a therapist is much better at doing that than I am.

PP5

This contrasts with the quote earlier in which the PP describes asking patients 'do you mind if [the DCP] does your filling?', which might imply that the DCP is inferior. Linked to this, some of the PPs interviewed talked of the significance of the way in which they explained the referral to patients. Positive pitches tended to highlight benefits to the patient such as more timely treatment or more specialist treatment. This may be a potential facilitating factor for patient acceptability of seeing DCPs instead of GDPs.

We're going to book them with the hygienist, we'll let them know that, yes, they can do XYZ, they're really good and we're passing you onto her. And now they have got used to that system.

PP7

# Comparison to medicine

When talking about the roles and duties of DCPs in practice, all groups of participants (PPs, DCPs and patients) interviewed tended to liken the model of working to a general medical practice where doctors work alongside nurses and nurse practitioners. This seemed to act as an anchor for understanding how different tasks might be divided among a multiskilled team. This comparison to something that was already working in an 'acceptable way' seemed to add weight to the idea that more role substitution within the dental team could work well.

There'll be less dentists and more therapists and I'm hoping it would become a model like the doctors' where they start off they see the nurse, the nurse does most of the stuff then the therapist does the next step and the dentist does the step after then the endodontist.

PP2

### Attitudes towards the scope of practice of dental care professionals

Practice principals seemed to focus on periodontal work as something that DCPs did and dental work as something that GDPs did, as a result of the focus placed on periodontal work during DCPs' training and their skill level in this area. This thinking appeared to impact on referral patterns as well as on the acceptability of DCPs working within their practices.

I've had a hygienist from the word go ... I think it's very, very important ... it's because I want them to do perio. I think perio is very important and I can use my time better. I do nervous patients, there's loads ...

PP\_T2

Other PPs admitted that the reason they were happy to refer patients for hygiene work is because scale and polishing is not an appealing or enjoyable task.

They do a good job and they take a dirty job off us that we don't normally like doing.

PP\_S3

The scope of practice for DCPs has recently been expanded to include some diagnosis and treatment planning. This seemed to be something that PPs felt unconvinced about and they tended to articulate this as concerns about patient safety. For example, one PP described DCPs as not being able to diagnose, even according to their scope of practice, let alone beyond it.

They're not able to diagnose everything that they're supposed to diagnose.

PP3

If DCPs miss problems that a GDP would have picked up and consequently do not make the appropriate referral, this could lead to worse health outcomes for the patient, although recent evidence confirms that DCPs are able to identify caries, periodontal disease and potentially malignant lesions at a comparable standard to GDPs.<sup>73,131</sup>

Even PPs who used DCPs extensively within their practice showed some reservation about DCPs preforming check-ups, reiterating that, while they could be trained to carry out dental work, diagnosis was something that only a GDP was truly qualified to perform.

The skill is diagnosis, but the actual treatment is the easy part. As long as you're good with the patient and you are competent in skills, you can teach anyone to do dentistry.

PP5

Some PPs argue against role substitution, as they prefer to provide all treatment for patients themselves. They like to be in control of every aspect of care, experience variety in their daily routine and use the full range of skills at their disposal. They also report job satisfaction from seeing diseased patients becoming healthy, which would be lost if they were not involved in the entire care pathway of patients. There is also the opinion that working full time on complex procedures would be tiring; therefore, some PPs prefer to blend complex and straightforward treatments to lighten the workload.

Doing crown and bridge all day or something it just . . . you know, your eyes would get tired. Everything would just . . . yeah, I mean your brain would hurt.

PP S3

The PPs' experience and approach to dentistry are reported as affecting the organisation of the practice and impacting on the rest of the workforce. PPs can be split into two groups: (1) those who like to perform a full range of treatment and actively seek more routine tasks as recovery time; and (2) those who would rather specialise in more complex treatments and refer routine tasks to others. The latter are more likely to see the benefits in adopting role substitution as it allows them to spend less time completing the routine, more straightforward treatments.

It allowed me to maybe concentrate on some of the more complex stuff, giving them the real bread and butter.

PP\_S4

The above PPs run practices that employ therapists and are utilising role substitution wherever possible, whereas the prior set of quotes originates from PPs who either own single-surgery practices or do not employ DCPs for anything more than hygiene work.

In a subgroup of practices, DCPs have been successfully performing hygiene work, but, when PPs attempt to refer patients to DCPs for more complex treatments, such as restorations, they receive a negative response from patients. This seems to be particularly common among private patients, who were reported as expecting restorations to be carried out by the GDP rather than a DCP.

We did have a therapist who worked with us, but I don't think we got full use out of the therapist, because I don't think that our patients wanted to see a therapist for having fillings done . . . It didn't seem to fit in with the, sort of, practice that we are, if you see what I mean . . . Well, they expect to see the dentist, you know. If they're paying private fees.

PP3

Some PPs see role substitution as a threat to their profession. Reasons cited were that DCPs could complete some aspects of the work of GDPs with less training. Another reason given was a concern about oversupply of GDPs and DCPs, particularly in areas with a limited number of patients and a large number of dental practices.

I think the idea was they [DCPs] would be able to do more when there's a shortage of dentists which is great but now there's not a shortage of dentists. So I think it's a worry there's not enough work for a lot of the therapists and the hygienists with all the new dentists coming out . . . Because there are enough dentists about that can treat the population.

PP S1

Practice principals who embrace role substitution have little concern over reduced workloads; instead, they see an opportunity for their practice to progress to a system which allows more flexibility in managing patient care.

Our patients who see the hygienist will see them on a regular basis. Very often they'll be in a couple of times in between visits with myself and that's been treatment planned by the hygienist himself. He will always be in communication with me if there are any issues or any concerns he has but we have got no issues. Even with patients phoning up, registered patients phoning up and saying I'd like to come in and see the hygienist. Fine.

PP S4

#### Financial and payment issues

Financial issues were of key concern to PPs. PPs using DCPs stated that their practices were not necessarily more profitable as a result, which is consistent with the finding from workstream 1.

You look at the profitability of a hygienist compared with a dentist and it made more sense to get another dentist.

PP1

I don't think it's improved the profitability, I'll be honest with you, because there's a lot of down-time . . . and we're still paying the therapist.

PP2

Instead of financial motivations for utilising DCPs in practice, PPs focused on benefits to the practice (e.g. smooth running), to themselves (e.g. more time) and to the patients (e.g. longer appointments with more detailed advice about self-care) as reasons for employing DCPs.

I think part of it is that we'd get a much better job with the patients, it's not just all about the money, we're happy enough that we've got a decent enough UDA value, how many pounds per UDA we get paid, that we don't have to just chase the money and it's about doing good jobs for patients.

PP7

However, it seems that profitability is a complex issue and PPs reported that the use of DCPs can impact on profitability in other ways if used creatively. Some PPs were starting to look at new models of care that expand the role of the DCP towards the top of their scope of practice. In these cases the DCP takes more of a leading role by acting as a triage service prior to a GDP agreeing a treatment plan.

The therapist just has a chat and takes a basic history and then I go in the room, do the charting, tell the therapist what needs doing and then she carries on.

PP2

The current remuneration system was also seen as a barrier to further utilisation by some PPs, who identified the difficulties in claiming UDAs for treatment completed by DCPs.

The entire remuneration system is based on the dentist with the performer number, taking for the UDAs that the treatment incurs in total, and then distributing.

PP7

Some PPs cite the physical space available to the practice to be a limiting factor to utilising more DCPs.

Lack of space. At the end of the day, if I have to do the check-up where do I have the hygienist therapists working? I've got an FD [foundation dentist] 4 days a week. I've got an associate in two days a week and I take a day off. I've got the rooms used to their full extent.

PP4

# Referral patterns

The current remuneration system also had an impact on some of the day-to-day issues experienced by DCPs in practice. In the majority of practices, the workload of DCPs depends on what is referred to them by PPs and associate GDPs. Patterns of referrals tend to vary from practice to practice. For example, in some practices, referrals to DCPs are made only by PPs and not by associate GDPs. In other practices, associate GDPs are incentivised to refer to DCPs. One PP explained the thinking behind an associate GDP choosing not to refer work to DCPs.

I can . . . get three UDAs. That's 60 quid. I can do that now while the patient is here, or I'll pass it over to the therapist and I've got to pay for it to them. Hang on! I'm not getting anything.

PP6

Another PP explained the reasons for the difference in referral patterns between associate GDPs and PPs.

If the associates see a patient who wants one filling doing, he'll do that one filling and I'll pay the associate. If they need half a dozen I'll say, refer it to my therapist, but then I get paid for the UDA, because I'm paying the therapist.

PP5

Practice principals in Scotland reported that they are forced to treat periodontal disease privately because the remuneration system does not suitably reimburse the practice for the time taken during a full periodontal treatment. As a result, patients in Scotland will see DCPs only as private patients and those who cannot afford to pay for private treatment may not see a DCP at all.

A way to overcome this would be to create a system in which the DHT could claim for the work that is required for patients.

Some of the practices that used skill mix appeared to have positive workplace cultures that valued the skill levels of all members of staff and supported continued learning and development. These seemed to be the practices that reported role substitution as working well.

We all help each other out. To be honest, we're a training practice so we also have a foundation dentist. That I think helps galvanise people into helping each other. And our associate was one of our ex foundation dentists as well, so they are all used to working together and helping each other.

PP1

# Benefits to the practice and general dental practitioners

As outlined above, numerous PPs noted the potential for skill mix to free the GDP's time to concentrate on more complex treatments. This was seen as a benefit that meant patients were able to receive their more complex treatment sooner than they would if the GDP was still carrying out a lot of routine work.

If I had somebody in and they were needing some fillings done if the therapist was doing these fillings for me I would be able to have somebody else in my chair at the same time so it was almost as if I was treating two patients at the same time.

PP\_S4

In contrast to this, some PPs were concerned regarding the efficiency of DCPs. Currently, DCPs are often allowed appointment times of around 30 minutes and, therefore, work to a different pace from that of the GDP. So when expanding the roles of DCPs, there may be concerns that DCPs take longer to carry out dental work and that they are not used to working efficiently.

I can do maybe three or four times more patients in an hour than a therapist could do.

PP5

# Results: dental care practitioners

# Patient awareness and acceptability

Dental care practitioners report that there still exists uncertainty from patients regarding what a DCP is actually allowed to do.

Patients are still a bit unsure of what we can and can't do.

DCP2

Dental care professionals were of the view that patients did not really want or need to know about their scope of practice because the majority of patients do not ask questions about the competence of the person providing treatment. This is because, in part at least, to patients trusting the GDP who made the referral or a belief that the system will ensure that everyone has the appropriate training and experience.

They're not interested in our regulations, are they? They just want to know that they're looked after.

DCP\_T3

There is the perception from DCPs that patients prefer seeing them for some of the routine treatments as this is often the impression they get from patients.

I feel a lot more comfortable [with you rather than the dentist], could you do my filling?

DCP\_T5

Approachability is one of the reasons why patients are happy to see DCPs and the reason why some patients prefer to see them rather than a GDP. There is also the point that DCPs generally have the

freedom to spend more time with patients and can therefore be more personable. Furthermore, the DCP has historically not performed surgical treatment, so is perceived by the patient to be gentler than a GDP.

Some of the patients prefer to see the hygienist because they like us more because we're nicer... think we're softer... They think if they see a therapist, oh, it can't be that bad then if a therapist is doing it, so might put my mind at rest a bit.

DCP7

In contrast, there are some DCPs who have experienced the opposite reaction, with patients expecting or demanding to have certain treatments completed by the GDP.

And I suddenly go, I'll give you a filling, you know, they'd be like, hang on a minute are you allowed to do it?

DCP T5

Dental care professionals emphasised that the way that the PP or GDP pitches the referral to the patient has an impact on patient acceptability. Some GDPs spend time explaining the DCPs' training, qualifications and experience. In contrast, others are described by DCPs as just informing the patient that a colleague will complete the treatment, with very little further explanation. Patients are likely to respond differently to different approaches and it may be that there is not one best way to approach this. According to DCPs, one of the most positive approaches is to explain that DCPs are better at some things.

The therapist is doing restorative care day in day out, this is what she does all the time; and, you know, if I had a choice to go . . . my dentist says if I had a choice the therapist would do my teeth.

DCP:

Some GDPs use a more negative approach and provide less detail to the patient. This can lead to patients perceiving that the GDPs believe the care or treatment is inferior.

Generally the dentists . . . I think they tend to say, I'm just referring you to the therapist because it's just a simple filling and I can do the more difficult stuff.

DCP T10

Some practices believe that patients and employees of the practice need to be re-educated in terms of their understanding of who completes which part of the patient care pathway. A way to address this is by adopting a proactive and systematic approach, explaining the roles and responsibilities to all stakeholders.

I think when they join the practice it should be explained how we work. You know, you'll see the dentist and then you'll see the therapist for this treatment . . .

DCP\_T4

#### Referral patterns

The issue of which members of the practice actually make referrals to DCPs appeared to be a barrier to, and facilitator of, role substitution. Some practices try to keep the system uncomplicated by allowing only the PPs to make referrals. PPs often have more of a team-working perspective, with the expectation that the practice team rather than individuals can achieve the UDAs target. As their personal targets and salary are not affected, they are happier to make referrals.

At this practice we've got two owners, so they both refer . . . they both know that all the contracts, everything will be coming back to them; so they refer to me and the money is going into their coffers all the time.

DCP1

The referral pattern becomes more of an issue in practices where associate GDPs are allowed or encouraged to refer. It is reported that many seem threatened or do not want to refer work. This can create a culture in which associate GDPs compete for UDAs, striving to achieve personal UDAs targets, but do not have enough patients to fill their own diary. The result is that they do not refer treatments to DCPs because it will affect their own personal income.

We've got two associates . . . they're doing the work and their books are a little bit gappy so, again, there's absolutely no chance and I don't have a nurse.

DCP T6

The practices that have a team-working approach see the UDAs target as a collective effort and understand that by referring on patients their targets can be achieved more efficiently.

They're happy to refer on to me so that we can get some UDAs sorted out to be honest, because if they're doing the fillings for the three UDAs and I'm doing it then they can be doing a crown prep work and denture work.

DCP\_T2

According to some DCPs, PPs see DCPs as part of a pathway to allow them to focus on more private work and maximise the practice's profit that way.

It's just a case of freeing up their time so that they can do more private work or more advanced treatment themselves.

DCP\_T5

In addition to the discussion on who makes referrals, the interviewees raised points around the type of patients that were referred. Some saw only children.

I [am] predominantly looking after the children especially because for some of them they've not had any experience before of any hands on clinical dentistry.

DCP6

We have some children, but I have more adult patients there in general.

DCP T3

Some GDPs see children as difficult and time-consuming, and they will happily refer children to the DCP so that they can use their time more effectively elsewhere. Other skills of DCPs were reported as being attractive to children as well, namely the 'softer' approach and opportunity to have longer appointments, which we described in the PP section.

Not in physical skills perhaps, but in the way the communication works with children . . . and I've got smaller hands, and being quite a gentle person, I think they see that, and I think their mums really like the preventative element and the way that I used to do that. So I think it comes from the patients as much as anything.

DCP1

This theme of GDPs referring to the DCP what they perceive to be difficult cases does not apply only to children. Difficulty does not necessarily refer to technical challenge, but instead is concerned with other aspects of care, such as managing patients with anxiety and time-consuming patients more generally. For example, this could be patients with learning difficulties, gag reflexes or other issues that mean a longer appointment will be required. Referral to DCPs can be seen in both a positive and negative way by DCPs. Some see it as flattering that a referring GDP values the DCP's skill to be able to provide a caring, patient-centred approach to patients that their pressing diary commitments and UDA targets do not allow.

Other DCPs seem to see this as GDPs being focused on profit maximisation and as soon as difficult patients come along they refer these patients. This is seen with complex cases; for example, a GDP receives the same UDA value for one or five restorations, yet the time required to complete five restorations is considerably higher, so GDPs will refer these patients on. Although it might be expected that the more highly skilled clinician would complete the more complex cases, the UDA system creates incentives for GDPs to pass these patients on to DCPs.

They are often difficult access, anxious patients, one reason or another complicated in handling the patient ones that won't be a quick earner . . . a complex patient, not the complex dentistry.

DCP1

Lack of control over referrals causes increasing frustration among DCPs. Some are left with large gaps in their diaries, which leads to difficult discussions with PPs. This means that DCPs may find it difficult to demonstrate that they are efficient and covering their costs, because they lack the opportunity to do this in the absence of referrals by GDPs.

I'm sitting around, like, doing absolutely nothing and I thought I've got a review next week which, it'll be brought up that, oh you've got gaps in your book, and it'll be . . . I've been told that I have to go ask the dentist to refer more to me.

DCP T8

Some DCPs have tried to address this by appealing to GDPs to refer more patients to them, but are met with a lack of interest and obstructive responses.

I go through phases of trying to argue the point and saying, look don't you want me to do this? But they just sort of shrug and fob me off. And, I've said, you know, I can mop up all your band 2 stuff, you can do all your fancy expensive stuff, you can push your private, but they're just not interested.

DCP\_T6

Other DCPs see blockages elsewhere in the referral pathway and believe that reception staff could do more to promote DHs and DTs.

We have one receptionist, and I think she's a problem in that she doesn't promote me at the desk and encourage people to book in with me. And then I think the dentist forgets to book them in with me.

DCP T8

#### Dental care professional skill level/experience

As with PPs, there are a variety of different sets of beliefs that frame the issue of skill level within the DCP workforce. In part, these may reflect different personalities and experience levels of DCPs, but views are also influenced by their experiences and relationships with their PP.

I do a lot where [the dentists] don't come in, only if it's particularly difficult would I get somebody in. I'd say I'm very confident.

DCP6

This confidence is grounded in the belief that they are the best at what they do and also links to DCPs having greater job satisfaction when they feel they are using their training to its full potential.

The more we can do is always better because it keeps it more interesting, doesn't it?

DCP7

As shown above, some DCPs will challenge a bad situation, whereas others are more willing to accept a poor situation because of a lack of self-confidence or fear of the situation elsewhere being more challenging.

If I went somewhere else, what happens if, because I've de-skilled . . . at the moment I have quite a long . . . I'll have 40 minutes a filling because my book's empty, and so there's no rush for me to do it because my book's quiet and I think I could go somewhere else where the demands are much higher and then it could be a practice that's worse than what I'm in.

DCP T8

Others are becoming so dissatisfied with both their underutilisation and movement towards more private, cosmetic, work that they are exploring the possibility of leaving general dental practice.

I don't feel I am being utilised really to my capabilities; I don't use any of my extended duties that I'm qualified with . . . I've been trying to look in to either getting in to the hospital environment or trying to change my role slightly because I don't really feel that private practice, you know, is for me.

DCP\_T9

Some DCPs show a lack of confidence as a result of what they feel is a lack of experience or opportunity. This often results in DCPs feeling like they are deskilling as a result of not using the skills they were trained in and a lack of trust from their PP. This leads to a lack of job satisfaction and lowering of morale.

They just see me really as a scaling machine, and so I have become deskilled and demotivated because of that.

DCP T8

Some of those who lack confidence are cautious about taking on more responsibility and see the link between experience and confidence leading to an increase in skill level. A key concept is that of a vocational training scheme, which all GDPs receive (as a foundation year), but this is optional for DCPs and geographically dependent. In addition, DCPs suggest that if PPs were exposed to DCPs during a vocational training year then it may raise awareness of DCPs' skill set.

I think I could have done with the VT [vocational training] just to have someone over my shoulder just saying, yeah, that's fine.

DCP T1

The erosion of DCPs' confidence and therefore skill level often occurs at the dental practice as a direct response to the behaviour of colleagues, the environment and the culture of the practice.

[I] Came back from there all motivated, like, you know, I want to do some more, my skills are really up, I'm happy with my local . . . and he just laughed at me. He just stood there and he laughed.

DCP\_T6

Although external factors can be to blame for a lack of confidence, some study participants believe this is related to the individual's characteristics and that some people prefer to follow the lead of others and do not really want the DCP role to develop beyond their comfort zone.

I'm quite an anxious person anyway, so I think for me I like it in black and white that, yeah, you can do a filling, yeah, you can do this.

DCP2

A similar number of interviewees were highly motivated to see change occur in their practices and determined to be allowed to work to their full scope of practice.

As a therapist I do have to prove myself and constantly remind people that I can do those things and that I'm capable of it.

DCP T3

#### Support from nurses

There is concern among the profession that many DCPs are not being fully utilised because they are not given the support of dental nurses in the surgery. Without a dental nurse, many aspects of the role of a DCP are difficult to complete. However, the response from the majority of interviewees was that they have support from nurses.

Of course, always with a nurse, yes. And my nurse would never leave me alone in the surgery.

DCP5

Unfortunately, some DCPs are expected to work without nurses. This often creates major dissatisfaction, and some DCPs raise concerns about the legality of this position. They live under threat of losing their job if they insist on having the support of a dental nurse or by challenging a PP on the issue.

If I give you a nurse just for your hygiene, you know, there might have to be a cut in your pay . . . we don't employ therapists because we don't give you nurses.

DCP T6

#### Workplace culture

Dental care professionals who feel valued, have high job satisfaction and report being well utilised often refer to a pleasant supportive environment within the dental practice. This is often driven by the PP but reinforced by all members of the multidisciplinary team within the practice. This facilitates the learning process for all practitioners and provides the opportunity for all to focus on their greatest assets.

If there's anything I'm unsure about I can ask anybody here and there's literally nobody who you think you can't speak to, which is good because that's what you need, especially when you're new. That's how you learn, isn't it, from people with more experience than you, it's what you've got to feed off, isn't it?

DCP2

This sort of culture and integrated workforce often results in people feeling valued because all members of the team understand what everyone does and views all roles as valuable and productive.

A practice like this where you feel like you're nurtured and you're helped, but you're doing it on your own as well, it's just a very good balance between the team here.

DCP2

At the same time, many DCPs report working in a culture of fear, with no opportunity to seek help and support and employees are undervalued. One of the main factors is a lack of understanding on the part of GDPs about what the DCP can do.

A genuine lack of knowledge and understanding of what we actually trained to do or how we're trained . . . traditional dentists who are thinking how on earth could they possibly diagnose and we're bound to miss things.

DCP T3

Dental care professionals report that often the PP does not value the knowledge and skills of the DCP, in terms of what they refer to them and the amount that practices are willing to pay DCPs.

The practice that I work for don't value me as a therapist, they're just valuing me as a, I guess, saving money for themselves.

DCP\_T4

This can lead to a culture where the DCP is driven out or, at the very least, sidelined.

#### Organisation of role substitution

The type of workforce model in place in the dental practices was explored in all DCP interviews. Two main approaches were identified; one is a traditional model in which DCPs are being used to complete hygiene work only (i.e. scale and polishing, oral health advice and root planning). The other group had embraced more of the ethos of role substitution where PPs and associate GDPs refer patients for restorations, often referred to as therapy work.

We generally get quite a good mix of treatments; I know some practices I think they're used a bit more as hygienists, but here we have a really good mix of perio and restorative as well, a good mix of adult patients and kid patients.

DCP2

There was very limited use of the direct access approach, although some saw the potential in the system and had explored it tentatively within their practices.

I've seen four or five DA patients a week. I then refer them on to the dentist.

DCP T1

Many DCPs continue to be limited to a reduced scope of practice in which they concentrate on oral hygiene advice and maintenance.

I trimmed a filling on Thursday and smoothed over an amalgam and put a tiny bit of composite in, and that's the first time I've done that in 2 months.

DCP\_T1

#### Innovative practices

There is an appetite in some practices, usually those described as having a good workplace culture, to use DCPs more creatively. Initial progress is made by offering the DCP more responsibility and moving them towards diagnosis and treatment planning by encouraging them to make their own decisions.

I can make a decision whether it's appropriate to take an X-ray.

DCP1

This increased level of responsibility was described as providing a foundation for a move towards direct access and potentially DCPs being utilised to diagnose and plan treatment.

We can screen the patients and treatment plan and such doesn't mean they have to be direct access patients, it means we can examine them, gather the information. As long as you know when to refer people, that's the important part.

DCP\_T3

#### Beliefs and attitudes: perceived benefits of role substitution

A key part of the interviews was to understand what the DCPs saw as the benefit to the practice and what benefit would be experienced by the PP and patient as a result of role substitution. The most common benefit reported was that DCPs allow the PP's or referring GDPs' time to be used more effectively.

It's just a case of freeing up their time so that they can do more private work or more advanced treatment themselves.

DCP T5

The benefits to patients that we described in the PP section are also echoed here, particularly the accessibility to treatment.

I've got a bit more time, you'd be waiting weeks to get in with [the dentist], I can see now we'll get it done quicker.

DCP6

#### Payment and its impact

There are different ways of paying DCPs. Many DCPs are self-employed and are paid an hourly rate for the time that they work, irrespective of the number of patients that they see in this time. PPs who run this system generally regard it as their responsibility to ensure that the DCP's time is fully utilised to ensure their employment is profitable and worthwhile. However, as described earlier, this does not always occur and some DCPs are still left with a lack of patients to treat. Another approach is where DCPs are employed by the practice and, therefore, receive a salary. This appears to give the DCP stability and a sense of being valued.

We're salaried, so although we are chasing UDAs, myself and the dentists are on a salary.

DCP T2

If I was salaried . . . it'd be better because then I'd be happy to carry out everything . . . And it wouldn't be that I'm taking UDAs off the dentist.

DCP T4

A third way is that DCPs receive payment for the number of patients they treat. One DCP suggested they would expect their rate of pay to be increased if they were to undertake restorative work, referred to here as 'therapy' work. This raises an interesting point about the value of treatments, which would be less of an issue for DCPs paid an hourly rate or fixed salary.

I'm paid a percentage of what the patient pays so, obviously, that would have to change because if I did more therapy.

DCP\_T7

Dental care professionals were sceptical about whether or not their services added to the overall profitability of the practice. They reported that the hygiene services generally became more cost-effective if undertaken as part of private treatment rather than within the NHS remuneration system. Furthermore, if they were self-employed and paid a fee for services completed, then they were very much aware that they could maximise earnings by focusing on private patients. In Scotland, in particular, it was very evident that completing hygiene work under the NHS remuneration system was far from profitable; in fact, it was virtually impossible for the practice to cover the costs of the treatment.

Therapists don't do therapy because it doesn't pay as well as if you went and did private hygiene work.

DCP T7

One repercussion of DCPs not having a NHS performer number is that they are not NHS employees and lack NHS employment benefits such as job security and NHS pension. This leads to significant resentment within the profession and the feeling of being a less valued member of the workforce.

Any work we do under a dentist prescription with their NHS number it's all allocated to them and they pay us. So that goes to their pension but we've not got a pension but yet we're doing the same work on the NHS.

DCP1

#### Barriers to utilisation of role substitution

Some barriers to greater use of DCPs have been explored elsewhere in this chapter; however, there were many more that DCPs described during the interviews. As with PPs, a common reason for not employing DCPs is a lack of space in the practice and a lack of time.

Just busy with our referrals and the ones that we get in off the dentist. I don't think there'd be time for us to just see people coming in off the streets or anything.

DCP7

A big issue for everyone is the lack of clarity in the legislation around DCPs and whether or not they can perform a more front-line role. Barriers cited are the lack of performer/contract number and (therefore) an inability to make a claim for completed treatment, and the confusion over completing the entire radiographic pathway. DCPs would prefer to be careful in the short term and not broaden their scope until clarity is provided.

You then have to go and do the prescribing radiographs course, which I'm in the middle of but then you still need a dentist to interpret it.

DCP6

I'd need to see in black and white, yes, you can do this; because I wouldn't want to do anything that would jeopardise a patient, or myself.

DCP2

Some DCPs report working in practices where PPs are resistant to change, in a system that works for GDPs and has been successful for years. Even if they can see benefits, there is still not a strong enough reason to make changes while the existing system is working and paying them well. This was reported as being more prevalent in the older generation of GDPs and is linked to concerns that it will result in job losses for GDPs.

[The older generation of dentists] were just not interested, not interested at all. They just saw it as we were trying to steal their jobs.

DCP\_T5

This point of the older generation not wanting to adopt new methods can be linked with teaching and the learning experience of GDPs. According to DCPs, dentists should be educated about the DCP role in dental undergraduate courses. This resonates with evidence demonstrating that courses where GDPs and DCPs work and learn alongside each other lead to a greater understanding of each other's roles and abilities. 132

They don't fully understand our training, like, they haven't been around for that long so it's like, okay can they actually do it.

DCP T4

In the whole 3 years we were there we had one lecture that was integrated with them.

DCP T5

#### Facilitators of greater utilisation of dental care professionals

Even though the introduction of direct access is yet to have an impact on access to DCPs, there is the expectation that the new dental contract will introduce additional incentives to increase their utilisation. To accompany this, a variety of methods to encourage this process were suggested. The most common suggestions cover the education of GDPs and patients, primarily to raise awareness and understanding. In addition to changes in education at dental school discussed above, DCPs suggested other action was needed.

You are going to have to maybe do like CPD [continuing professional development] events, work it into the contracts particularly when this new contract does come out . . . I think if that can outline and show people how it will work and how the financial benefits for the dentists would be, then maybe that might make them change their mind.

DCP6

I think it's showing a financial gain and how more efficient their appointment book will be and how happier the patients would be. It's just educating them.

DCP\_T7

#### Results: patients

#### Awareness of the roles of members of the dental team

In general, patients had low levels of awareness of the specific roles and duties of different members of the dental team. When asked about what they knew about the care provided by DHTs, patients tended to be unsure. A patient from a practice without any DCPs referred to hygiene work only.

Basically just a clean and polish. I don't know if there's anything else they do. Is it just called a clean and polish, or something?

Practice 4, patient 5

A patient from a practice using DCPs as DHs and DTs stated they knew nothing at all about the role.

Nothing at all, because I just come here and get my teeth done, so I wouldn't know anything would I?

Practice 5, patient 3

Similarly, when asked if they knew about what therapists did, another patient stated:

No, not really. No, I'm just happy that they do it and it's comfortable what they do and it's good for you, for your teeth.

Practice 6, patient 1

However, there was a broad understanding that GDPs could do everything but were experts in doing dental work and DHs were experts in hygiene work. Therapists did not fit well into this understanding. One patient expressed a little surprise when it was explained to her that a DHT could carry out a filling.

Right, yeah. I don't know, I just thought, you know, it goes with the title. You know, hygienist, just keeping it clean.

Practice 4, patient 5

This statement highlights the importance of the job title itself and how this helps to form patient understandings of the roles of members of the dental team. Another two patients explained their understanding of the work of GDPs and DCPs.

It's the dentist that I need to see for fillings, etc., and they're just mainly for the hygiene, the cleaning of the teeth.

Practice 6, patient 3

General dental practitioners were seen by some as being necessary for more serious treatments. One patient at a practice that utilised a model whereby low-risk patients attended routine hygiene appointments at 3-month intervals stated:

You do see the dentist afterwards; that's for more important things really.

Practice 6, patient 3

Another patient reported the work that GDPs do as being more 'detailed' and 'severe':

Only in so much that when you see the dentist it's more involved and it's more detailed. The work that the dentist has to do is a bit more severe.

Practice 4, patient 3

Patients also thought of dentists' time as limited and saw referrals to DCPs for hygiene work as a way in which the GDP was able to spread the workload:

They just clean, but more thoroughly than the dentist has got time to do.

Practice 6, patient 1

One patient, however, did have a good understanding of DHTs; this was because a daughter of a family friend was currently training as a DHT. She said:

I think there are some hygienists that go through a diploma course, which is 2 years, and I'm aware now that there's a BSc course for oral hygiene . . . I'm aware that they do a 3-year course and that they are qualified to do some fillings – as I understood it, you could correct me if I'm wrong – on minors, on children.

Practice 3, patient 1

#### Perceived drawbacks

Following some explanation to the patients of role substitution and based on their existing experience of being cared for by DCPs (mostly by DHs rather than by DHTs), some patients expressed concern about the greater utilisation of DHTs. Much of the concern was around DHTs carrying out dental treatments such as fillings. However, there was a number of other perceived drawbacks, including patient preference and convenience.

One such perceived drawback of DCP utilisation was reported by a patient who attended a practice that did not currently employ DCPs. Thinking back to a time when he attended another dental practice, he described the inconvenience of having separate appointments with a GDP and a DH.

They [previous dental practice] had a hygienist which I used to use but I've no need to here because I get it done with my check-up. I didn't have any problems at my check-up and [XXXX] does it all for me. If I have a problem in between I can come and have one [appointment] and so it's much better than booking it separately.

Practice 4, Patient 1

Some patients felt hesitant about seeing someone other than the GDP to whom they had become accustomed. Patients explained that they had good relationships with their GDPs and would not be happy about being cared for on a regular basis by anyone other than their own GDP. This appeared to be regardless of whether the patient was to change from their GDP to a DCP or another GDP. One patient said:

I think I'd prefer to stick with my dentist that I know . . . Just because I don't like trying new dentists. So when you've found one that you really like, I think it's best to stick with them . . . I think it's just that I'm comfortable with what I know.

Practice 6, patient 1

Another patient said that they would reluctantly see someone other than their usual GDP if there was no other choice:

Only if it was OK and there was no choice because my dentist wanted it, you know . . . what's the expression? If it ain't broke, don't . . . you know?

Practice 4, patient 1

Further to these perceived drawbacks, some patients expressed concern over safety issues and the quality of the treatment carried out by DCPs. This tended to relate to thinking along the lines that DHTs' levels of training and skills were lower than those of a GDP. One patient commented on the ability of GDPs to check the health of the whole mouth and spot potentially serious problems.

I'm sure a simple filling is a simple procedure in itself but the dentist isn't just doing a filling, they're looking at other things as well. You know, you might have a bad reaction to a local anaesthetic or it might be another issue. There might be a crack in the tooth that's not been spotted, who knows?

Practice 4, patient 4

Two patients said that they would need a better understanding of the training and the skill levels of DHTs before they would be able to make a decision on whether or not they would accept treatment from a DHT.

I'd be unsure, to be honest. I'd like to know how much they'd done in the past. How many fillings they had done, the training involved, etc. . . . need to understand the level of training and the level of experience that went with it.

Practice 4, patient 4

One patient described that they would not have as much confidence in the abilities of a DHT in comparison to a GDP because of the perceived level of skills and training levels.

I wouldn't have as much faith in a therapist as I do in a dentist because you feel they've got a lot more experience and studied a lot longer.

Practice 6, patient 3

Further to this statement, the patient expressed concern about value for money.

I would think you were getting short changed if you didn't see a dentist really, if you were, you know, being passed on to a hygiene therapist.

Practice 6, patient 3

Another take on this idea of value for money was that some patients might be happier to receive treatment from a DHT if the charge was lower than that of a GDP. One patient at a private practice noted that although this was not the case for her, she could imagine that it might be an attractive prospect for others.

In my case, it wouldn't, but other people might think a lot of it's cheaper for me to have that filling done by a hygienist or somebody with an oral hygiene qualification, and certainly in the economic climate it may be a choice that people feel that they want to make.

Practice 3, patient 1

This patient paid for her dental care privately. She was more aware of the training and qualifications of DHTs than the other patients interviewed; however, she seemed to be sceptical of the quality of the treatment that would be provided by a DHT as opposed to a GDP. She noted that quality was not something she would be prepared to trade off for cost. However, she felt that some people who were more financially restricted than her might be prepared to make this trade-off.

Although this patient was sceptical of the use of DHTs, this seemed to bother her only when thinking about her own treatment. A contrasting view was obtained from another private patient, who expressed concern on a moral level that NHS patients might be 'passed on to a hygiene-therapist'. She felt that the treatment would be of a lesser quality and that this would be a failing of the NHS to patients.

You know, the experience and . . . they've obviously got more qualifications in the long run, the dentist, than the therapist would have, so I would feel, yeah, you'd be a second citizen really. Definitely, it is the dentist that should be doing the actual fillings and the extractions and things like that. It sort of horrifies me to think that you're actually saying this and that they might be considering that for national health patients.

Practice 6, patient 3

#### Perceived benefits

Although some of the patients interviewed highlighted concerns, a large proportion of the patients were able to see the potential benefits of using DHTs in practice. These principally revolved around freeing up GDPs' time to concentrate on more complex treatments. This was thought to be a benefit by some because it could allow for improved rates of access.

If it helps other people out and gets the jobs done quicker and make it more accessible then yeah.

Practice 5, patient 6

Several patients stated that they would be happy to be treated by a DHT instead of a GDP.

Most patients, even those sceptical of DHTs carrying out dental work, thought that use of DHs was important for freeing up GDPs' time.

For [the hygienist] to be able to do that and save [the dentist] the time it is much better, in my opinion, because she's looking at the hygiene . . . it saves [XXXX] time and gives him a chance to concentrate on more complicated issues.

Practice 3, patient 1

In addition to saving time, some patients thought that visiting a DCP rather than the GDP could be an easier option, especially for those with dental anxieties or fears. One patient described how she used to feel nervous about visiting the GDP and how she found it easier to see the DHT.

Yeah, I'd rather go to [hygienist therapist] about anything to be honest with you. I mean ... They've been alright, the dentists in here, but I haven't got the same ... I haven't got the same, what do you call it ... reassurance as I have with her ... I used to dread coming, my nerves would be all over the place before [the hygienist] came but now I look forward to coming, you know, because I've got into that routine of coming every 3 or 4 months. And she's always so nice, asks us about the family and my granddaughter and everything.

Practice 5, patient 1

#### Acceptability

Although there were a number of patients who were unconvinced about the greater utilisation of DCPs in practice, levels of acceptability among the majority of patients were generally high. The majority of the patients were very accepting of DHs, particularly in terms of hygiene work. Some patients even stated that

they would prefer to have hygiene work carried out by a DH than a GDP. This seemed to be because DHs had more time and were experts in hygiene work.

I prefer it because I think that's their field, and they excel at it and she is exceptionally good.

Practice 5, patient 2

Other patients reported that they would be as happy to receive dental treatment from a DHT as they would from a GDP. In some cases, patients used their past experiences to explain their choices.

I trust her, she's been doing it for years and they obviously know everything about it . . . To do my fillings, yeah, fire away. No bother, no. It's my fillings, I can't see any reason why they can't.

Practice 3, patient 2

Across the majority of patients interviewed, it was apparent that there was an implicit trust in health-care professionals and in the health-care system that provided patients with a feeling of safety. Patients trusted that there were sufficient checks and balances within the broad NHS that ensured the appropriate training and vetting of health-care professionals. This trust seemed to reassure patients that standards of care would be maintained.

Patients felt that GDPs would refer only to qualified health-care professionals. In addition, some patients felt satisfied about being treated by a DHT if they could be reassured that they were appropriately qualified to carry out the treatment.

It's not a problem as long as . . . obviously; the person's going to be qualified anyway. They're not going to let an amateur loose on you.

Practice 4, patient 3

Other patients, when asked why they would feel happy to be referred to a DHT for dental treatment, responded with explanations about trusting their GDP or the practice that they attended.

Because I trust them here, they are lovely here.

Practice 5, patient 6

It was apparent that, for some people, their experience of the practice they attended directed their levels of trust in the quality of care provided.

I'd feel fine about it [being referred to a hygienist therapist for a filling] as long as they could do it. Yeah, it doesn't bother me. They've always looked after my teeth well enough here so as long as it gets sorted, it doesn't really matter does it? As long as it's done professionally . . . if they're getting fixed it doesn't really matter. If they are qualified then fair enough.

Practice 5, patient 6

Another patient described how it was important for her to have trust in the individual GDP or DCP treating her on a personal level. She stated that she would prefer her own GDP to carry out even complex treatments rather than be referred to secondary care because she trusted her GDP and knew she would be safe.

[It's more about getting to trust the person rather than] . . . whether it be dentist or a hygienist, basically . . . Yes, I would say that in general. If they trust somebody, you have you have faith in them . . . so I'd rather . . . I'd rather [my dentist] did it than go to the hospital because you know then.

Practice 3, patient 2

#### **Discussion**

The PPs interviewed in this study appeared to be in agreement that, within the current NHS contract, the greater use of DCPs did not result in financial gain. These views support the findings from *Chapter 4*. The current system appears to disincentivise the use of role substitution. Although those PPs who did not use DCPs fully did talk about financial 'disincentives', this was not the primary reason for their lack of use. 31,56,83-90 Instead, the attitudes and beliefs of PPs appeared to be highly influential in terms of explaining the organisation of the practice; that is the type and organisation of the practice culture that cascaded down from the practice owner (PP) was key. This concurs with the findings of Watt *et al.* 101 and McDonald *et al.* 102

Practices with low role substitution utilisation appeared to be led by PPs who were concerned about the abilities of DCPs, in particular how much longer it might take a DCP to carry out a treatment than a GDP.<sup>32</sup> However, it is worth noting that PPs who used DCPs tended to have confidence in the ability of their own DCPs, but were less confident in the ability of DCPs in general. This suggests that experience of working with DCPs may be a factor that influences GDPs' attitudes and beliefs about DCPs.<sup>56,113</sup>

The professional domain of DCPs was also limited by many PPs to the provision of periodontal treatment. This concurs with the work of Godson *et al.*, <sup>92</sup> who found that many DHTs and DTs in England were employed in the NHS as DHs, rather than being utilised across their full range of skills. Those PPs who supported the greater use of DCPs for more interventional treatment (e.g. preparing teeth for restorations) limited their activity to children and anxious patients, on the basis that these types of patients took up more surgery time. As a result, DCPs appeared to be used by many PPs to improve their own efficiency, at the cost of the efficiency of their own DCPs. This may help explain why there was no net effect across the practice. Periodontal treatments are limited to band 1 payments (and some band 2 payments for complex activity), so the number of UDAs produced by DCPs would be reduced.

Overall, financial incentives remained important for PPs, but so were the beliefs and attitudes of PPs to role substitution and DCPs generally.<sup>31,83–86,88–90</sup> It may be the case that these attitudes could change over time if there was greater alignment in the system between financial incentives and the greater use of DCPs. This has been shown to increase utilisation in medicine.<sup>14,102,103</sup>

Among DCPs themselves, a number of day-to-day barriers were found to reduce the efficiency of working in the practice environment. These included not being able to prescribe fluoride varnish and the inability to prescribe local analgesia when they were asked to carry out more interventional treatments. Some practices had developed organisational systems to work around these barriers, such as setting up a 'patient group directive' (an agreement overseen by a pharmacist that allows DCPs to prescribe medicines using their own assessment of patient need). However, many practices did not have such agreements in place and this meant that every time a DCP wished to issue fluoride supplements or provide local analgesia prior to treatment, they had to get approval from the GDP.

Another important related restriction is the inability for DCPs to examine patients or to provide a treatment plan under the NHS dental contract regulations. Although the dental regulator recently changed the scope of practice for DCPs,<sup>58</sup> within the NHS, the GDP still needs to see the patient first. This creates an extra step in the referral pathway and substantial transactional cost, as patients that are seen by GDPs first have to be then 'booked back in' with a DCP if they require treatment. In contrast, GDPs have the freedom to conduct the treatment in the same visit, immediately after the examination, if they have time. Both care pathways generate three UDAs, so the efficiency of the latter is far greater than the former and may contribute to the results found in *Chapter 4*. Furthermore, some of the DCPs interviewed reported that they were not referred enough work to keep them busy all day, a factor that would again reduce their efficiency. This was particularly marked for those DCPs that worked in corporate practices, where they were paid on a FFS basis rather than by an hourly rate.

This reliance on referrals from GDPs appeared to reduce the DCP workload in the practice and a substantial proportion of those interviewed worked across a number of practices (between two and eight). This is consistent with the research that has been conducted in England<sup>92,114</sup> and highlights the precarious nature of professional work for many DCPs.

In general, the majority of patients had a low level of awareness of the roles of different members of the dental team. Most patients interviewed were familiar with GDPs and DHs, but they were not usually aware of DTs. Patients tended to compartmentalise their views on who did what in dental practices; fillings, crowns and bridges were activities undertaken by the GDP, while periodontal treatment ('hygiene work') fell within the expertise of DHs. All patients were happy to have hygiene work carried out by a DH. Although most were open to a DT carrying out simple restorative work, some were adamant that they would only be happy with a GDP carrying out this treatment. These views concur with earlier research. 93–95 However, all patients who had had an experience of being treated by a DT were happy. This may indicate that patients' prior experience of treatment is key. If their only experience was treatment by a GDP, then treatment by a DT may appear daunting. Once they realise that there is no real difference in their management, they appear happy to accept being treated by a DT.

The majority of patients interviewed were simply not interested in the debate about who treated them. Instead, they reported that they just wanted to know that any treatment being carried out was undertaken with a suitably qualified clinician. They all appeared to place a great deal of trust in the NHS, the regulatory system and in the opinion of their GDP. Patients were happy to comply with what was recommended to them. In similarity to the findings by Northcott *et al.*, <sup>133</sup> if the GDP was happy to refer them, they were happy too. <sup>133</sup> Essentially, they felt that the system/regulator/practice/GDP would look after their best interests.

# **Concluding remarks**

In relation to RQ2, the barriers to, and facilitators of, role substitution in NHS dental practices were varied. Financial implications were important, but the organisational culture at the practice was also critical. PPs views of role substitution were varied and key. As PPs own their practices, their views heavily influenced the culture at a practice. DCPs highlighted the inability to prescribe fluoride, local analgesia and radiographs as important barriers to efficient working practice in the NHS. Patients appeared to have an inherent trust in the 'system' and prior experience of role substitution was influential on its acceptability.

# **Chapter 6** The role of financial incentives on NHS dental activity

#### Introduction

Chapters 4 and 5 highlight the variation in efficiency in NHS dental practices and the range of factors that can influence it, including role substitution. This chapter details how the research team used a natural experiment to explore the impact of a change in the remuneration system on the delivery of dental services in a large group of NHS dental practices in Northern Ireland.

In May 2009, the Northern Ireland Health Minister, Michael McGimpsey, announced a plan to improve access across the province to enable 'people to get the health service dental treatment they need'. Rather than paying NHS dental practices via FFS, new contracts were to be based on capitation. Oasis Dental Care was among a number of groups that responded to the tender and, in April 2010, it was successfully awarded funding to provide 38 GDPs working in 14 new practices located in regions where patient access to oral health care was limited. Oasis Dental Care was chosen for their 'extensive expertise and experience in providing dental services to health Trusts across the United Kingdom'.

Contracts were awarded for a period of 3 years and they were to be paid in a block contract payment to register and provide services up to 57,000 patients by 1 August 2012.<sup>131</sup> No additional NHS payments were to be made to Oasis for patient registrations that exceeded this target. Practices were opened at different time points in the first year of this contractual period and each practice was given a target number of registrations to reach within the first year. The Health and Social Care Board in Northern Ireland monitored monthly registrations closely and held regular monitoring meetings with Oasis Dental Care.<sup>131</sup> They considered Oasis to have made 'reasonable efforts' to achieve the target if they operated within a 5% level of the target. Penalties were to be applied if they fell outside this range.

The aim of this study is to examine how specific health-care service indicators differ between the Oasis practices and control practices paid by traditional FFS payments after controlling for other patient and practice characteristics. We addressed the following RQs:

- 1. Patient selection: do Oasis practices serve a different population of patients from control practices?
  - i. Is the proportion of registered patients who are  $\geq$  60 years of age greater in Oasis practices than in control practices?
  - ii. Is the proportion of registered patients who are  $\leq$  18 years of age greater in Oasis practices than in control practices?
  - iii. Is the proportion of registered exempt patients  $\leq$  18–60 years of age greater in Oasis practices than in control practices?
- 2. Receipt of care among registrants: do patients registered to Oasis practices:
  - i. See the GDP less frequently than patients in control practices?
  - ii. Have fewer treatment courses than patients in control practices?
  - iii. Receive less costly treatment than patients in control practices?

- 3. Type and mix of treatment received: are patients seen in Oasis practices:
  - i. More likely to receive an examination than patients in control practices?
  - ii. More likely to receive an extraction than patients in control practices?
  - iii. More likely to receive a filling than in control practices?
  - iv. More likely to receive a scale and polish than in control practices?
  - v. More likely to receive prevention/topical fluoride than in control practices?
- 4. Financial viability:
  - i. Is the payment charge revenue generated per registered patient less among Oasis practices than in control practices?
  - ii. Is the payment charge revenue generated per treatment course less among Oasis practices than in control practices?

#### **Methods**

#### Data collection and matching approach

The data used in this study were based on claim forms that GDPs are required to submit to the Health and Social Care Board as part of their NHS contractual regulations. Although Oasis practices were not remunerated on the basis of services provided, they were still required to submit the same documentation as GDS practices for auditing purposes.

We compared the 14 Oasis practices with a control group of matched GDS practices paid by FFS. Control practices were selected based on practice location. Postcode data of all registered practices with the Health and Social Care Board for Northern Ireland (that hence provide services under a NHS contract) were used to select the five nearest practices to each Oasis practice. Distance from an Oasis practice was chosen as the matching variable as an attempt to control for the oral health-care needs of populations being served by the practices. This provided a control group of 70 practices. After screening the data, 4 of the 70 control practices were removed from the sample because of incomplete claims data. Practices with fewer than 1000 patients registered at the practice for NHS care in any month were removed from the sample because this could have indicated a new practice was not yet operating at a steady state or that a practice was reducing registrations in advance of practice closure. This resulted in removing one Oasis practice and nine practices from the control group, leaving a study sample of 13 Oasis practices and 57 control practices.

The data from the claim forms were pooled over 43 months (from April 2011 to October 2014). Seventy practices formed an unbalanced panel data set of 2971 monthly practice claims (556 Oasis practices and 2415 control practices). A full description of Oasis and control group practices' sample characteristics can be found in *Appendix 3*.

The analysis covers the period from April 2011 to October 2013, in order to ensure a sufficient number of monthly observations for panel regression. The study period started a year after the Oasis contract was signed to accommodate the different starting dates for the Oasis practices and to ensure the policy change in the extension of the patient registration (from 15 months to 2 years) had taken place.

#### Variable description

The analysis is based on routinely collected data on a range of study variables and potential confounders. The outcome measures (or dependent variables) in this study were patient selection, receipt of care among registrants, mix of treatments and financial viability.

The study population was divided into subgroups for analysis of patient selection. Patient selection was defined as the proportion of registered patients aged  $\leq$  18 years, the proportion of registered patients aged  $\geq$  60 years and the proportion of registered patients exempt from payment charges for one of the following reasons: expectant or nursing mothers, income support, job seeker's allowance, working family tax credits, and certificates for full or partial help with health-care costs. These exemption categories were chosen to capture the proportion of the registered population where there may be oral health need. 132

Receipt of care among the registered patient population was defined as the number of unique patients seen (i.e. repeated visits from the same patient was not counted) per 100 patient registrations, the number of treatment courses per 100 registered patients and the monetary value per treatment course. This provided an indication of whether or not patients registered for NHS care received services during a particular month, and the pattern and type of services received.

Treatments provided were categorised into different types to compare the case mix profiles of Oasis and GDS practices. Type and mix of treatment were defined as the prevalence of particular services per 100 unique patients seen. The particular treatments used for this variable were the number of examinations, the number of restorations (fillings), the number of extractions, and the number of scale and polish treatments. As a measure of the prevalence of preventative care, we use the number of fluoride varnishes per 100 children registered at the practice. Routine application of fluoride varnish is recommended for children from the age of 2 years in the NHS in England (*Delivering Better Oral Health*).<sup>134</sup>

Finally, financial viability was defined as the Patient Charge Revenue generated from the NHS services delivered. Reductions in Patient Charge Revenue associated with the frequency and content of services delivered under capitation would increase the net cost of NHS dental care to the board, so is an important consideration for NHS commissioners.

The explanatory variables used in the analysis to control for patient need were the proportion of adult registered patients to registered children, the proportion of exempt patients and the Index of Multiple Deprivation score for the practice location. The capacity of the practice to meet needs was measured by the number of GDPs working at the practice.

#### **Analysis**

To address the RQs, the mean difference between Oasis and control group practices was estimated after controlling for other explanatory variables using an OLS regression model with robust standard errors.

Ordinary least squares is applied to panel data in long panel form (data on n practices, over t monthly periods, for a total of  $n \times t$  observations), with a number of monthly observations for each practice. These repeated observations of an individual practice cause clustering and standard errors to fall as the number of repeated monthly observations increases. These standard errors at practice level were adjusted (robust standard errors) to control for this in a pooled OLS regression. The estimates from the OLS models are presented as associations as there is the possibility that unobservable influences that mediate the relationship between the explanatory variables and the outcome measures invalidate any conclusions of causality.

Panel models provide a way of accounting for unobservable influences, but the data available for this analysis did not favour that approach because of a lack of variation over time.<sup>135</sup> Panel model specifications were estimated, tested and rejected in favour of pooled OLS. The coefficients and statistical significance of exploratory variables on the OLS model were almost identical to panel specifications and a joint Wald test did not reject the null hypothesis that the coefficients on time-varying explanatory variables were identical. This suggests that there was not enough variation across the months within practices to justify a panel approach and we chose pooled OLS with robust standard errors as our main model specification (see *Appendix 4* for a detailed explanation).

The matching variable measured whether an individual practice is part of the Oasis group or control group. The statistical significance and direction of the estimated coefficient indicates if there is a difference between Oasis practices and the control group after controlling for differences in practice and patient characteristics that were (a) related to the outcome measure and (b) unrelated to whether a practice operates on the Oasis contract. In the patient selection model, the explanatory variables measuring patient mix (aged  $\geq$  60 years, children and patient need) were not included because they were not independent of the dependent variable (the proportions of registered patients in particular age groups and in particular exemption categories).

The Ramsey regression equation specification error test (RESET) was used to test for model misspecification; that is whether or not there was a statistically significant improvement in fit with non-linear combinations of explanatory variables.

Data analysis used Stata (StataCorp LP, College Station, TX, USA) software version 11.1. A p-value < 0.05 was used as the threshold for statistical significance and highlighted in the results tables.

#### Results

#### **Descriptive statistics**

Table 20 describes practice characteristics and patient selection variables for Oasis practices in comparison with the control practices. Oasis practices had greater staffing capacity to deliver health care than control practices, with, on average, 0.89 more dentists active per practice per month. The mean number of monthly treatment items delivered was larger among control practices (922 treatment items) than in Oasis practices (811 treatment items). The mean monthly number of treatment plans delivered to patients was larger for an Oasis practice (463 treatment plans) than a control practice (392 treatment plans), while the mean monetary value of the treatment plans delivered was lower in an Oasis practice than for a control practice by £10.83. These differences between the two groups were statistically significant at a 5% level.

TABLE 20 Sample comparison of GDS control practices with the Oasis group for practice characteristics and patient selection variables

	Mean practice va	lue in	Difference in				
Variable name	GDS control practices (n <sup>a</sup> )	Oasis group (nª)	mean practice values	<i>p</i> -value of difference			
Practice characteristics							
Number of dentists	3.21 ( <i>n</i> = 2079)	4.11 (n = 478)	-0.89	< 0.00			
Regional deprivation (Index of Multiple Deprivation score)	21.20 (n = 2415)	19.80 ( <i>n</i> = 556)	1.41	< 0.00			
Number of monthly treatment items	922 $(n = 2079)$	811 (n = 478)	110	< 0.00			
Number of monthly treatment plans (claims)	392 $(n = 2079)$	463 (n = 478)	<del>-</del> 71	< 0.00			
Number of monthly registrations	3359 (n = 2415)	3850 (n = 556)	-490	< 0.00			
Number of monthly patients seen	306 (n = 2415)	320 ( <i>n</i> = 556)	-14.40	0.15			
Patient selection							
Percentage of registered patients aged ≥ 60 years	15.30 (n = 2079)	13.30 (n = 478)	2.01	< 0.00			
Percentage of registered patients that are children	30.70 (n = 2415)	19.90 ( <i>n</i> = 556)	10.70	< 0.00			
Percentage of registered patients exempt from payment charges	43.20 (n = 2415)	36.70 ( <i>n</i> = 556)	6.60	< 0.00			
a One observation is one practice observed in 1 month.							

Table 21 describes receipt of care among registrants, mix of treatments and financial viability variables for Oasis practices in comparison with the control practices. Oasis practices deliver care over a larger patient population. An Oasis practice had, on average, 490 more registered patients and treats an average of 14 more unique patients per month compared with control group practices. This suggests that, although Oasis practices were purposely set up in locations where the government had identified barriers which prevented access to oral health care in the resident population, the practices nonetheless were facilitating access to health-care services to more patients than neighbouring control group practices.

#### Main findings

The estimated coefficients are presented in *Table 22*. Each row in the table is a separate regression, with a different dependent variable to answer a different RQ. Estimated coefficients on the matching variables indicate the difference in the outcome variable between Oasis practices and control practices when the values of other explanatory variables in the equations are held constant. The table shows the matching variable estimate without the use of control variables, with only practice control variables and with practice and case mix control variable. The last is the main model specification. p-Values of the matching variable  $(r^2)$ , the number of practices (n) and the number of practice month observations (N) are described for each regression.

TABLE 21 Sample comparison of national practices with the Oasis group for receipt of care among registrants, mix of treatments and financial viability variables

	Mean practice val	ue in	Difference in			
Variable name	GDS control practices (n°)	Oasis group ( <i>n</i> <sup>a</sup> )	mean practice values	<i>p</i> -value of difference		
Receipt of care among registrants						
Number of monthly patients seen per 100 registrations	8.90 ( <i>n</i> = 2415)	8.40 ( <i>n</i> = 556)	0.54	< 0.00		
Number of monthly courses per 100 registrations	11.70 (n = 2079)	12.50 (n = 478)	-0.80	< 0.00		
Value of treatment per treatment course (£)	33.80 (n = 2079)	22.90 (n = 478)	10.80	< 0.00		
Mix of treatments						
Number of monthly examinations per 100 patients seen	66.70 (n = 2079)	62.40 (n = 478)	4.30	< 0.00		
Number of monthly extractions per 100 patients seen	16.10 (n = 2079)	20.70 (n = 478)	-4.60	< 0.00		
Number of monthly fillings per 100 patients seen	60.30 (n = 2079)	44.30 (n = 478)	15.90	< 0.00		
Number of monthly scale and polishes per 100 patients seen	48.20 (n = 2079)	40 (n = 478)	8.10	< 0.00		
Number of monthly fluoride varnishes per 100 patients seen	0.023 (n = 2079)	0.018 (n = 478)	0.005	0.60		
Number of monthly fluoride varnishes per 100 child registrations	0.008 (n = 2079)	0.007 (n = 478)	0.001	0.83		
Financial viability						
Patient payment charge revenue per registration (£)	11.50 (n = 2079)	15.40 (n = 478)	-3.93	< 0.00		
Patient payment charge revenue per treatment course (£)	9.10 (n = 2079)	10.50 (n = 478)	-1.40	< 0.00		
a One observation is one practice observed in 1 month.						

<sup>©</sup> Queen's Printer and Controller of HMSO 2016. This work was produced by Brocklehurst et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton So16 7NS, UK.

**TABLE 22** Separate regressions showing the difference in the outcome variable between Oasis and comparison practices for each group of RQs

	Matching va	riable	With practice control variables		With practice and case mix control variables		
Dependent variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	n, N, r²
Patient selection							
Percentage of registered patients aged ≥ 60 years	-2.01	0.10	-1.72	0.35	-	-	70, 2557, 0.04
Percentage of registered patient that are children	-10.74	< 0.00	-11.2	< 0.00	-	_	70, 2557, 0.19
Percentage of registered patients exempt from payment charges for reasons associated with high dental care need	-6.55	0.05	-3.81	0.21	-	-	70, 2557, 0.14
Receipt of care among re	gistrants						
Patients seen per 100 registrations	-0.54	0.07	-0.60	0.04	-1.61	< 0.00	70, 2557, 0.11
Treatment courses per 100 registrations	0.78	0.01	0.71	0.02	0.44	0.33	70, 2557, 0.01
Value of treatment per treatment course (£)	-10.83	< 0.00	-10.41	< 0.00	-14.00	< 0.00	70, 2557,0.29
Mix of treatments							
Examination per 100 patients seen	-4.30	0.02	-3.45	0.07	-9.7	< 0.00	70, 2557, 0.05
Extractions per 100 patients seen	4.62	< 0.00	5.12	< 0.00	6.31	< 0.00	70, 2557, 0.18
Fillings per 100 patients seen	-15.97	< 0.00	-15.21	< 0.00	-17.12	< 0.00	70, 2557, 0.22
Scale and polish per 100 patients seen	-8.13	0.01	-9.03	0.01	-11.5	< 0.00	70, 2557, 0.07
Fluoride varnish per 100 patients seen	-0.01	0.80	0.002	0.89	0.03	0.41	70, 2557, 0.19
Fluoride varnish per 100 child registrations	-0.001	0.91	0.001	0.85	0.008	0.33	70, 2557, 0.06
Financial viability							
Patient payment charge revenue per registration (£)	3.93	< 0.00	3.09	0.01	-0.68	0.63	70, 2557, 77
Patient payment charge revenue per treatment course (£)	1.37	0.17	0.76	0.42	-2.91	< 0.00	70, 2989, 0.80

The findings in *Table 22* were robust to model specification. The RESET test was performed for each outcome measure. In each case, the *p*-value of the test was > 0.05, indicating that the null hypothesis (that the model is not misspecified) should not be rejected. In addition, there is little difference in the magnitude and statistical significance of the matching variable when it is estimated by random-effects panel model or panel model and Mundlak correction (see *Table 46*, *Appendix 3*).

#### Patient selection

There is no evidence of differences in patient selection between Oasis and control practices based on patient need. The proportion of registered patients from payment charge-exempt groups with high-expected health-care need was not significantly different between the two practice types. There was mixed evidence of patient selection based on age. Oasis practices had, on average, 11 fewer children per 100 registrations than control practices (p < 0.00), although there was no statistically significant difference in the number of older (aged  $\geq 60$  years) registrants per 100 practice registrants.

#### Receipt of care among registrants

An Oasis practice saw on average 1.6 fewer unique patients per 100 patient registrations per month than a control group practice (p < 0.00). The average value per treatment course was £14.00 lower in an Oasis practice than in a control practice (p < 0.00). This suggests that, over the given time period, Oasis practices saw a smaller proportion of its registered patients and provided courses of a lower monetary value than control practices. There was no statistically significant difference between the practice groups in the average number of monthly claims (a proxy for the number of patient treatment plans) made per registered patient, which suggests that there was no evidence of Oasis practices 'underserving' registered patients in the absence of any treatment plan-based remuneration incentives.

#### Mix of treatments

There was evidence of a difference in treatment prescribing patterns between the two practice groups. An Oasis practice provided an average of 9.7 fewer examinations, 17.2 fewer fillings and 11.5 fewer scale and polish services per 100 unique patients seen per month than control practices (p < 0.00). However, they provided an average of 6.3 more extractions per 100 unique patients per month (p < 0.00). The volume of fluoride varnish applications per patient seen and per 100 child registrations was not significantly different between practice groups.

#### Financial viability

In terms of the financial implications of funding care under capitation (the Oasis contract), an Oasis practice received an average of £2.91 less patient payment charge revenue per treatment course than a control practice (p < 0.00). However, this was mitigated by the lower proportion of children in Oasis practices, meaning a greater proportion of the treatment plans delivered by Oasis practices were subject to patient charge payments. Once we adjust for these 'offsetting' patient mix factors, it resulted in no significant difference in the monthly payment charge revenue per registered patient between Oasis and control practices.

#### **Discussion**

In similarity to earlier findings, financial incentives in the remuneration system did impact on the clinical activity undertaken in the NHS. $^{13,14,135}$  A major consideration in comparing different practice models was whether the different models attract and/or serve populations with different needs. Our analysis indicated that the proportion of registered patients aged  $\leq 18$  years was significantly less in Oasis practices, but no differences were found in the proportions of registered patients aged  $\geq 60$  years, or eligible for exemption from NHS payment charges for reasons associated with higher probability of need. This suggests that there is no evidence of patient selection (or 'cherry picking') associated with need for care among Oasis practices.

After controlling for differences between the two groups in practice and patient characteristics, the number of unique patients seen per 100 patients registered at the practice per month and the average value per treatment course claim were lower in Oasis practices than the control group. This suggests that registered patients at Oasis practices were less likely to visit the GDP than patients in control group practices and received less treatment when they did attend. However, it is important to note that this might be the result of unobserved heterogeneity between practices. As a result, it cannot be concluded that these differences were caused by the different ways the providers were remunerated. For example, Oasis practices were set up where populations were unable to get access to a NHS dentist, hence one could hypothesise that the Oasis practice population might have already had their care needs met by an abundance of provision in the private sector.

It is unlikely that the findings related to the care received among registered patients reflect underlying differences in needs of the patient populations of the two groups of practices. This is because the models included explanatory variables that captured the age mix of the practice population, the prevalence of exemption from NHS payment charges for reasons of higher risks of oral health problems and the level of deprivation in the local community. The findings cannot be explained by Oasis practices having a lower capacity to care for their registered patients because the number of active GDPs in each practice is included as an explanatory variable in the regression model. Although causality is not established by the modelling approach, the findings were consistent with GDPs working under capitation-based remuneration systems seeing patients less frequently and 'doing less' to patients when they do see them.

In terms of the type and mix of services delivered, we found that Oasis practices had higher levels of tooth extractions but lower levels of examinations, scale and polishes and restorations (fillings) than control practices. A factor that is important to consider here is that extractions take less time than other treatment alternatives, that is, clinicians operating within a capitation system could prefer a management option that requires less time.<sup>136</sup> A 'scale and polish' is a preventative procedure aimed at removing calculus and plaque that can cause oral health problems (gum disease) in the future. Equally, it is a procedure that many patients like, because of its aesthetic value (removal of stained deposits). This suggests that Oasis practices were not any more prevention oriented than control practices. However, the application of fluoride varnish is potentially a better marker of prevention orientation and its use on the registered patient population was no different between Oasis practices and the control group. The lower prevalence of examinations and fillings among Oasis practices suggests that patients registered with these practices were not seen as often and had longer recall periods. Although this might reduce the prevalence of unnecessary interventions among patients, equally it may also lead to some oral health problems being missed and interventions being delayed.

Finally, no evidence was found that patient payment charge revenue differed between Oasis practices and those that operated on a FFS basis. Although Oasis practices received less patient payment charge revenue per course of treatment than control practices (explained by the lower value of treatment per course), the revenue per registered patient was not significantly different. However, Oasis practices had a smaller proportion of children who were exempt from NHS patient charges than control practices, which would offset the reduced revenue per patient income and account for the non-significant findings in overall patient payment charge income.

The main limitation of the study concerned the limited amount of information available on the practices and the populations served by the practices as well as the non-random nature of allocating the provider payment methods and allocation of patients. Receipt of oral health care is conditional on registration, therefore the models used in this study were conditional on registrations and hence subject to selection bias. One approach to correct for sample selection is the Heckman model. <sup>137</sup> This incorporates the probability of registration within the population, but as this study is confined to *practice* populations, such methods cannot be used. Instead, an OLS regression model with robust standard errors was used. As highlighted above, the study was designed to minimise the impact of patient and practice heterogeneity between Oasis and control practices. As a result, a 'difference in difference' design could not be used

(there was no 'before-intervention' period for the Oasis practices).<sup>138</sup> Although we were able to identify statistically significant differences in some aspects of patient populations, receipt of care among registrants, service mix and patient payment charge revenue, data were not available on patient outcomes. As such, it is not possible to determine whether the lower prevalence of interventions in the Oasis practices was the result of inappropriate supplier-induced demand among control practices (unnecessary interventions) or the failure to intervene in a timely way among Oasis practices (supervised neglect).

## **Concluding remarks**

The natural experiment described in this chapter explored the change of remuneration in Northern Ireland and enabled the research team to address the third RQ. Capitation influenced the organisation of inputs and outputs in NHS dentistry in Northern Ireland. Practices examined fewer patients and undertook fewer restorations. Capitation remuneration did not improve the level of prevention (fluoride varnish) offered, but did increase the number of extractions performed.

These results must be interpreted with caution. The success of the findings to indicate comparative differences depends on the populations being distinct from each other only in the type of remuneration system. Oasis practices were set up in areas of lowest dental health-care access in Northern Ireland, although, to account for the influence that may have had on interpreting the effect of remuneration incentives, a control variable for regional deprivation was included in the regression analysis and the comparison group chosen from nearby practices. Unfortunately, the circumstances of this natural experiment (there was not a group of Oasis practices remunerated in the same way as GDS practices) prevented an analysis of the remuneration incentives that was dissociated from the management systems and culture. Oasis practices were set up and administered by a corporate provider where the organisational environment (e.g. ways of recruiting, supervising and remunerating the work of staff) might not necessarily match that of the GDS practices.

# **Chapter 7** Conclusions and recommendations

his programme of research sought to address three RQs:

- 1. What is the most efficient mix of GDPs and DCPs for NHS dental practices to achieve UDA targets?
- 2. What are the barriers to, and facilitators of, role substitution in high-street NHS dental practices?
- 3. How do incentives in the remuneration systems in the UK influence the organisation of these inputs and production of outputs in the NHS and population of patients they treat?

From the preceding chapters, it appears that the use of role substitution in NHS dentistry appears to be widespread in some clinical domains (e.g. periodontal treatment), while limited in others (e.g. the provision of restorations) (see *Chapter 3*). The most common form of role substitution found was the use of DHs to undertake hygiene work (see *Chapters 3* to *5*). This may reflect a lack of opportunity for DCPs to exercise their full range of skills within the NHS. Equally, it could reflect some GDPs being unwilling to allow DCPs to encroach on traditional dentist roles. Whether DCPs were hired to replace the clinical activity previously provided by GDPs or simply to meet an increase in demand for routine services is not known. Supporting the latter is the finding (see *Chapter 5*) that not all PPs or associate GDPs were aware of the extended roles of DCPs. Most of the clinical activity that was produced by the practices that had been identified in *Chapter 3* as 'pro'-role substitution was actually generated by more traditional models of service provision, using GDPs rather than DCPs. Overall, the proportion of DCP to GDP sessions seeing NHS patients was relatively low (16.6%) (see *Chapter 4*).

Chapter 4 found that many NHS dental practices in England were not operating at or near the production frontier, when compared with the most efficient group of practices in the sample. This finding may be influenced by the cost-containment mechanism within the 2006 NHS dental contract in England. Output constraints within this contract place a ceiling on the amount of clinical activity that can be generated within any one financial year. Given that the numbers of surgeries and GDPs working in NHS dental practices are relatively inflexible inputs, the inability of PPs to adjust the team size and composition as the contracted deadline for the target output is approached might explain the inefficiency in production. The analyses for the smaller sample from Northern Ireland who were paid on a FFS basis (see Appendix 1), and hence faced no output constraints, showed that their practice efficiency was higher than in England.

The key finding from *Chapter 4* was that NHS dental practices that utilise fewer DCPs capable of role substitution operate at higher levels of technical efficiency, after controlling for other staff inputs, practice environment and patient characteristic variables. This finding from the DEA was robust to different specifications of inputs and outputs and was supported by the SFM.

When UDAs were used as the output measure, NHS dental practices in England operated at a mean level of efficiency of 64%. This changed very little when the outputs were measured in terms of the number of patients seen or the number of treatment plans generated. NHS dental practices that did not use any form of role substitution had a higher mean level of efficiency (68%; n = 39). The variation seen in the full data set in England was small, with 95% of scores lying within 5 percentage points of the mean.

Efficiency in NHS dental practices in Northern Ireland and Scotland was consistently higher (80% and 94%, respectively). Although it is difficult to make robust comparisons across the different countries because of their smaller sample sizes (n = 29 and n = 20, respectively), the results suggest that English NHS dental practices operate at a lower level of efficiency and variance. This may be influenced by the remuneration system employed in England and the limits that are placed on clinical activity (i.e. the output constraint caused by the ACV and the need to keep within a fixed range of UDAs). Without this output constraint (Northern Ireland and Scotland), it appears that many NHS practices maximise their technical efficiency, expanding their production until their inputs are fully utilised.

Caution should be exercised in the interpretation of the efficiency measure. This study examined the efficiency of NHS dental practices, not DCPs (as DCPs do not directly contract with the NHS). It could be that DCPs are being used efficiently but their use is associated with a reduction in efficiency for the practice overall. Equally, efficiency gains in the NHS may be lost to the private sector, if the GDP uses freed-up capacity (by the DCP) to undertake private dentistry. The output measures used, UDAs and number of patient visits, may not reflect the quality of care provided. NHS dental practices may be more efficient in terms of health outcomes, but less efficient in terms of NHS activity, patient throughput and remuneration. Such considerations would require data collection on the quality of the services produced by GDPs and DCPs, which was beyond the scope of this study.

The finding that a greater number of DHs, DTs and DHTs was associated with lower productivity might appear surprising. However, our qualitative investigation highlighted a number of factors which explain why this might be the case. These included attitudes, incentives, structural factors, systems of remuneration and the lack of direct access to DCPs. *Chapter 5* found that there was a disincentive for associate GDPs within practices to refer on to DCPs. Many PPs were also concerned about the financial implications of employing DCPs for activity other than periodontal treatment (hygiene work). The current remuneration system in NHS dentistry does not appear to reward practices for the use of role substitution and a number of PPs were concerned with the perception that DCPs took longer to carry out treatment than a GDP.

General dental practitioners' attitudes and beliefs about DHs', DTs' and DHTs' skill levels also acted as a barrier to them practising to their full scope of practice. GDPs expressed doubts about DCP skills and also reported that in the context of an expansion of supply into the dental market (via an increase in the number of dental schools and the number of overseas GDPs), DCPs capable of role substitution were perceived as a threat to GDPs. Another influential factor in the utilisation of DCPs was the organisation of the practice, (i.e. the type and organisation of the practice culture that cascaded down from the PP; see *Chapter 5*). Some PPs expressed a lack of confidence in the abilities of DCPs generally, although were very happy with the performance of their own DCP. The lack of a contractual relationship between the NHS and DCPs meant that their financial independence was limited and so they were subject to the views of the individual PPs that employed them. This meant that many DCPs had a precarious existence, with multiple jobs at different practices (see *Chapters 3* and *5*).

The need for better alignment between the financial incentives within the NHS dental contract and the use of DCPs is key if the NHS is to address the recommendations of the Independent Review<sup>2</sup> and make best use of the whole dental workforce. Aligning financial incentives to encourage the greater use of role substitution would be an important intermediate step, although the development of a direct DCP NHS contract would offer greater flexibility for commissioners. This is important as oral health continues to improve, if maximising health gain for a given level and mix of resources is an ethical imperative for the NHS. Failure to do so will increasingly mean that the most expensive resource treats those with the least need.

A number of regulatory barriers were also identified by the DCPs that were interviewed. The inability to prescribe fluoride, local analgesia and radiographs (X-rays) were identified as the key barriers to efficient working practice in the NHS (see *Chapter 5*). Some practices had developed organisational systems to work around some of these barriers, such as setting up patient group directives to allow prescription-only medicines to be prescribed. However, many practices did not have such agreements in place and this meant that every time a DCP wished to issue fluoride supplements or provide local analgesia prior to treatment, they had to get approval from the GDP.

Another important, related, restriction is the inability for DCPs to examine patients or to provide a treatment plan under the NHS dental contract regulations. Although the dental regulator recently changed the scope of practice for DCPs, within the NHS, the GDP still needs to see the patient first. This creates an extra step in the referral pathway and substantial transactional cost, as patients that are seen by GDPs first

have to then be 'booked back in' with a DCP if they require treatment. In contrast, GDPs have the freedom to conduct the treatment in the same visit, immediately after the examination, if they have time. Both care pathways generate three UDAs, so the efficiency of the latter is far greater than that of the former.

The majority of patients had a low level of awareness of the roles of different members of the dental team and were not interested in the debate around role substitution within the NHS (see *Chapter 5*). Many exhibited an inherent trust in the system, although prior experience of role substitution was influential in social acceptability.

Drawing conclusions about the relationship between role substitution and efficiency requires an appreciation; therefore, the way in which role substitution is currently being implemented in practice makes it difficult for many practices to reap potential benefits of role substitution. The factors outlined in *Chapter 5* act as powerful barriers to more efficient ways of working, with the result that many DCPs capable of undertaking role substitution feel demotivated and underutilised.

Chapter 6 found that remuneration systems based on capitation did influence the organisation of inputs and outputs in NHS dentistry in Northern Ireland. Practices examined fewer patients and undertook fewer restorations, a finding that was not explained by a difference in patient mix. Lower levels of activity remained when patient selection (for a population with lower levels of dental health-care need) was analysed. Capitation remuneration did not increase the level of prevention offered, but did increase the number of extractions performed.

## **Considerations for future policy**

The growing demand for health care in the UK, recent rises in NHS expenditures and decreasing health-care budgets highlight the importance of paying attention to the organisation and performance of the dental health-care system and of providing policy-makers and practice owners with evidence-based information to better inform their decisions. On account of this background, efficiency analysis and benchmarking has an important role in enabling health policy-makers, practice managers and dental professionals to identify and assess different approaches to organising the delivery of dental health care, including the success of different dental team structures in terms of processes and outcomes. We found a comprehensive analysis of dental health-care efficiency in the UK a challenging task, primarily because data were more readily available on units of health-care output than service quality or patients' oral health gains from treatment. We also found that dental practices were reluctant to provide cost information that could be used to estimate the profitability of their service. In these circumstances, in which data collection is limited, it is possible that findings have been influenced by unobserved determinants of practices' choice of inputs that also determine output, for example past investment in equipment or the attitudes of dentists towards DCPs.

## Strengths and limitations of the research

This was the first study of its kind to empirically examine the efficiency of dental teams in the NHS that make use of role substitution. It was also of methodological interest, given that both models for determining efficiency in health care were utilised and the results were found to be robust and independent of the method of analysis, that there was consistency across both the non-parametric and parametric methods that were utilised. The study was also the first to compare the efficiency of NHS dental teams across different remuneration systems, although the limited sample sizes in Northern Ireland and Scotland prevented a fully robust comparison.

We examined a range of environmental and organisational influences, both internal and external to the dental practice, to establish associations with more efficient production. Ideally we would have applied longitudinal data with frequent waves to account for changes in the dental practice team and changes in care patterns over time. Without longitudinal data it was not possible to identify the extent of change in the efficiency of dental services or determinants of improvements in service performance.

Efficiency studies in NHS dental service provision to date have not adopted a robust empirical design and have used a limited range of input and output measures. A strength of our findings is that they are presented with both activity-based measures (patients seen and treatment plans) and price-based measures (NHS remuneration income). These measures allowed us to explore technical inefficiency (inefficiency caused by the wasteful use of inputs) although limitations in the availability of price and cost data prevented us from exploring allocative inefficiency (inefficiency caused by selection of an inappropriate input mix).

A major limitation of the programme of study was that it was not possible to observe inputs or outputs produced independently by DCPs who were capable of role substitution. Given that the current contractual relationship is between the NHS and the GDP alone, it was not possible to determine the efficiency of DCPs independently. Study findings therefore require careful interpretation, as we only estimate the effect of role substitution on the output of the whole dental team rather than the individual DCPs' output. In addition, we relied on practice-reported input measures recorded via a questionnaire. The pilot questionnaire was tested in depth and a balance had to be found between the inclusion of variables that were considered to be important in the production function and the likelihood of practices participating and returning the questionnaire. Any more detailed analysis would have required a prospective study. The research team also found that many dental practices were wary of sharing too much financial information and had to be reassured that the linkage between input data and the output data kept by the NHS would be anonymised at the source.

Another limitation is that random selection of practices did not take place. This would influence findings if there were factors unobserved to researchers that were related to the incidence of role substitution (e.g. dentists' choice to work with DCPs or limited dentist awareness of DCPs full scope of practice), as this may influence both the choice to hire a DCP and the supervision they receive to produce output. However, a broad range of covariates was entered into the analysis and the results remained robust.

#### **Recommendations for further research**

Further research is needed to fill the gaps in some areas of our productive efficiency model, such as a greater range of capital resources. Use of a health-related quality-of-life measure would provide a measure of the effect that dental health care has on patients' oral health. However, there are difficulties with this approach. Routine dental patients are largely non-symptomatic, so quality-of-life measures in this population would not identify subtle variations in effect and appropriately discriminate.<sup>139,140</sup>

To better inform future planning decisions, it is recommended that future research integrate efficiency analysis with other relevant factors of overall performance in NHS dentistry such as equity, service coverage and access to care, responsiveness to changes in population needs and preferences for services, and accurate alignment at a national level of supply (e.g. the size of the workforce and health-care budget) with population needs for care (driven by, for example, population demographics and oral epidemiology). However, there are challenges to accessing the level of detail required for a full analysis of efficiency at a NHS practice. Prospective studies could offer potential, but do require the full co-operation of the NHS dental practice and our experience to date is that not all appreciate the extent of information required.

# **Concluding remarks**

The use of DCPs capable of role substitution in NHS dental teams is not associated with higher levels of efficiency at the practice level. More attention needs to be paid to identifying the conditions under which DCP use improves practice efficiency and how NHS contracts could be developed in the future to support the efficient use of role substitution within NHS dentistry.

# **Acknowledgements**

#### **Contributions of authors**

**Professor Paul Brocklehurst** (Director of NWORTH Clinical Trials Unit, Professor of Health Services Research and Honorary Consultant in Dental Public Health, Bangor University) led the research team and provided oversight on both main workstreams. He also oversaw the preparation and writing of the report.

**Professor Stephen Birch** (Professor in the Department of Clinical Epidemiology & Biostatistics, McMaster University and Professor of Health Economics, University of Manchester) led the development of the RQs and study objectives for workstream 1. He also oversaw the design of the workstream, the interpretation of the results and the preparation and writing of the report.

**Professor Ruth McDonald** (Professor of Health Services Research and Policy, University of Manchester) led the development of the RQs and study objectives for workstream 2. She also oversaw the design of the workstream, the interpretation of the results and the preparation and writing of the report.

**Dr Harry Hill** (Post-doctoral Research Associate, University of Manchester) undertook the research and analysis for workstream 1. He also wrote the section of the report on the use of technical efficiency (see *Chapter 4*).

**Dr Lucy O'Malley** (Lecturer in Oral Health Research, University of Manchester) undertook the research and analysis for workstream 2. She also wrote the section of the report that examined the experience of DCPs in the workplace (see *Chapter 5*).

**Mr Richard Macey** (Doctoral student, University of Manchester) also undertook the research and analysis for workstream 2. He also wrote the section of the report that examined the experience of DCPs in the workplace (see *Chapter 5*).

**Professor Martin Tickle** (Director of the Institute of Population Health, Professor and Honorary Consultant in Dental Public Health) helped to interpret the findings of both main workstreams and contributed to the writing of the report.

## **Contributions outside the study team**

We would particularly like to thank all those from the NHS and associated organisations who gave so freely of their time during every phase of the project. Particularly, we would like to thank the following:

- Brenda Gannon from the Centre of Health Economics, University of Manchester
- Sandy Fitzpatrick from the Business Services Organisation
- Gillian McCrory from the Business Services Organisation
- Michael Donaldson from the Health and Social Care Board
- Donncha O'Carolan from the Health and Social Care Board
- Jan Clarkson from the Scottish Practice Based Research Network
- Linda Young from the Scottish Practice Based Research Network
- Lorna Barnsley from the Scottish Practice Based Research Network
- the Business Services Authority
- the National Information Centre
- the Privacy Advisory Committee.

# **Publications**

Brocklehurst PR, Birch S, McDonald R, Tickle M. Determining the optimal model for role substitution in NHS dental services in the United Kingdom. *BMC Oral Health* 2013;**13**:46.

# **Data sharing statement**

Data can be obtained from the corresponding author.

# References

- 1. Hollingsworth B. The measurement of efficiency and productivity of health care delivery. *Health Econ* 2008;**17**:1107–28. http://dx.doi.org/10.1002/hec.1391
- 2. Steele J. NHS Dental Services in England. London: Department of Health; 2009.
- 3. Health & Social Care Information Centre. *NHS Dental Statistics for England 2013–14*. URL: www.hscic.gov.uk/article/2021/Website-Search?productid=15294&q=NHS+Dental+Statistics +for+England+-+2013-14&sort=Relevance&size=10&page=1&area=both#top (accessed 23 June 2015).
- Godson JH, Williams SA. Inequalities in health and oral health in the UK. *Dent Update* 2008;35:243–50.
- Milsom KM, Jones C, Kearney-Mitchell P, Tickle M. A comparative needs assessment of the dental health of adults attending dental access centres and general dental practices in Halton & St Helens and Warrington PCTs 2007. Br Dent J 2009;206:257–61. http://dx.doi.org/10.1038/sj.bdj.2009.165
- Gallagher J, Wilson N. The future dental workforce? *Br Dent J* 2009;**206**:195–9. http://dx.doi.org/ 10.1038/sj.bdj.2009.114
- Farrell MJ. The measurement of productive efficiency. J Royal Stat Soc Series 1957;120:253–90. http://dx.doi.org/10.2307/2343100
- 8. Hollingsworth B, Street A. The market for efficiency analysis of health care organisations. *Health Econ* 2006;**15**:1055–9. http://dx.doi.org/10.1002/hec.1169
- 9. Health & Social Care Information Centre (HSCIC). *Adult Dental Health Survey 2009*. Leeds: HSCIC; 2011. URL: www.hscic.gov.uk/pubs/dentalsurveyfullreport09 (accessed 23 June 2015).
- Steele J, Treasure E, O'Sullivan I, Morris J, Murray J. Adult Dental Health Survey 2009: transformations in British oral health 1968–2009. *Br Dent J* 2012;**213**:523–7. http://dx.doi.org/10.1038/sj.bdj.2012.1067
- 11. Health & Social Care Information Centre (HSCIC). *Child Dental Health Survey 2013*. Leeds: HSCIC; 2013. URL: www.hscic.gov.uk/catalogue/PUB17137 (accessed 23 June 2015).
- 12. Brocklehurst PR, Ashley JR, Tickle M. Patient assessment in general dental practice risk assessment or clinical monitoring? *Br Dent J* 2011;**210**:351–4. http://dx.doi.org/10.1038/sj.bdj.2011.284
- 13. Tickle M, McDonald R, Franklin J, Aggarwal VR, Milsom K, Reeves D. Paying for the wrong kind of performance? Financial incentives and behaviour changes in National Health Service dentistry 1992–2009. *Community Dent Oral Epidemiol* 2011;**39**:465–73. http://dx.doi.org/10.1111/j.1600-0528.2011.00622.x
- 14. McDonald R, Cheraghi-Sohi S, Sanders C, Tickle M. Changes to financial incentives in English dentistry 2006–2009: a qualitative study. *Community Dent Oral Epidemiol* 2012;**40**:468–73. http://dx.doi.org/10.1111/j.1600-0528.2012.00687.x
- 15. Chalkley M, Tilley C, Young L, Bonetti D, Clarkson J. Incentives for dentists in public service: evidence from a natural experiment. *J Public Admin Res Theory* 2010;**20**(Suppl. 2):i207–i23. http://dx.doi.org/10.1093/jopart/muq025
- 16. Chalkley M, Tilley C. Treatment intensity and provider remuneration: dentists in the British National Health Service. *Health Econ* 2006;**15**:933–46. http://dx.doi.org/10.1002/hec.1162

- 17. Birch S. The identification of supplier-inducement in a fixed price system of health care provision: the case of dentistry in the United Kingdom. *J Health Econ* 1988;**7**:129–50. http://dx.doi.org/10.1016/0167-6296(88)90012-4
- 18. Grytten J. Models for financing dental services. A review. *Community Dental Health* 2005;**22**:75–85.
- 19. Gosden T, Forland F, Kristiansen I, Sutton M, Leese B, Giuffrida A, *et al.* Capitation, salary, fee-for-service and mixed systems of payment: effects on the behaviour of primary care physicians. *Cochrane Database Syst Rev* 2000;**3**:CD002215. http://dx.doi.org/10.1002/14651858.cd002215
- 20. Ellis RP, McGuire TG. Supply-side and demand-side cost sharing in health care. *J Econ Perspect* 1993;**7**:135–51. http://dx.doi.org/10.1257/jep.7.4.135
- 21. Laurent M, Reeves D, Hermens R, Braspenning J, Grol R, Sibbald B. Substitution of doctors by nurses in primary care. *Cochrane Database Syst Rev* 2009;**4**:CD001271.
- 22. Sibbald B, Laurant M, Reeves D. Advanced nurse roles in UK primary care. *Med J Aust* 2006;**185**:10–12.
- 23. Richardson G. Identifying, evaluating and implementing cost-effective skill mix. *J Nurs Management* 1999;**7**:265–70. http://dx.doi.org/10.1046/j.1365-2834.1999.00137.x
- 24. Wilson A, Pearson D, Hassey A. Barriers to developing the nurse practitioner role in primary care the GP perspective. Fam Practice 2002;**19**:641–6. http://dx.doi.org/10.1093/fampra/19.6.641
- 25. Charles-Jones H, Latimer J, May C. Transforming general practice: the redistribution of medical work in primary care. *Soc Health Illness* 2003;**25**:71–92. http://dx.doi.org/10.1111/1467-9566.t01-1-00325
- 26. Edelstein BL. Examining whether dental therapists constitute a disruptive innovation in US dentistry. *Am J Public Health* 2011;**101**:1831–5. http://dx.doi.org/10.2105/AJPH.2011.300235
- 27. Mertz E, Glassman P. Alternative practice dental hygiene in California: past, present, and future. *J Calif Dent Assoc* 2011;**39**:37–46.
- 28. Glassman P, Subar P. Creating and maintaining oral health for dependent people in institutional settings. *J Public Health Dent* 2010;**70**:S40–S8. http://dx.doi.org/10.1111/j.1752-7325.2010.00174.x
- 29. Edelstein B. The dental safety net, its workforce, and policy recommendations for its enhancement. *J Public Health Dent* 2010;**70**:S32–S9. http://dx.doi.org/10.1111/j.1752-7325.2010.00176.x
- 30. Skillman SM, Doescher MP, Mouradian WE, Brunson DK. The challenge to delivering oral health services in rural America. *J Public Health Dent* 2010;**70**:S49–S57. http://dx.doi.org/10.1111/j.1752-7325.2010.00178.x
- 31. Sun N, Harris RV. Models of practice organisation using dental therapists: English case studies. *Br Dent J* 2011;**211**:E6-E.
- 32. Harris RV, Sun N. Dental practitioner concepts of efficiency related to the use of dental therapists. *Community Dent Oral Epidemiol* 2012;**40**:247–56. http://dx.doi.org/10.1111/j.1600-0528.2012.00670.x
- 33. Greene WH. The Econometric Approach to Efficiency Analysis. In Fried HO, Knox Lovell CA, Schmidt SS, editors. *The Measurement of Productive Efficiency and Productivity Growth*. Oxford: Oxford University Press; 2008. http://dx.doi.org/10.1093/acprof:oso/9780195183528.003.0002
- 34. Wade J, Donovan JL, Athene Lane J, Neal DE, Hamdy FC. It's not just what you say, it's also how you say it: opening the 'black box' of informed consent appointments in randomised controlled trials. Soc Sci Med 2009;**68**:2018–28. http://dx.doi.org/10.1016/j.socscimed.2009.02.023

- 35. Corbin JM, Strauss A. Grounded theory research: procedures, canons, and evaluative criteria. *Qual Soc* 1990;**13**:3–21. http://dx.doi.org/10.1007/BF00988593
- 36. Geertz C. The Interpretation of Cultures: Selected Essays. New York, NY: Basic Books; 1973.
- 37. Great Britain. National Health Service Act 1946. London: Her Majesty's Stationery Office; 1946.
- 38. Great Britain. *National Health Service (Charges for Dental Treatment) Regulations, 1952* (S.I., 1952, Number 1020). London: Her Majesty's Stationery Office; 1952.
- 39. The Health Departments of England, Scotland, Wales and Northern Ireland. *The New Contract An Operating Manual for Dentists*. London: DH; 1990.
- 40. Bloomfield K. Fundamental Review of Dental Remuneration. London: DH; 1992.
- 41. Great Britain. NHS (Primary Care) Act. London: The Stationery Office; 1997.
- 42. Goodwin N, Morris A, Hill K, McLeod H, Burke F, Hall A. National evaluation of personal dental services (PDS) pilots: main findings and policy implications. *Br Dent J* 2003;**195**:640–3. http://dx.doi.org/10.1038/sj.bdj.4810781
- 43. Pitts N. NHS Dentistry: Options for Change in context a personal overview of a landmark document and what it could mean for the future of dental services. *Br Dent J* 2003;**195**:631–5. http://dx.doi.org/10.1038/sj.bdj.4810779
- 44. Department of Health (DH). *Modernising NHS Dentistry Implementing the NHS Plan*. London: DH; 2000.
- 45. Great Britain. *The National Health Service (General Dental Services Contracts) Regulations 2005*. London: The Stationery Office; 2005. URL: www.legislation.gov.uk/uksi/2005/3361/contents/made (accessed 20 February 2016).
- 46. House of Commons Health Select Committee. *Dental Services Fifth Report of Session 2007–08*. London: The Stationery Office; 2008.
- 47. Department of Health. NHS Dental Contract Pilots Learning after First Two Years of Piloting. London: The Stationery Office; 2014. URL: www.gov.uk/government/uploads/system/uploads/attachment\_data/file/282760/Dental\_contract\_pilots\_evidence\_and\_learning\_report.pdf (accessed 20 February 2016).
- 48. Department of Health (DH). *Dental Contract Reform: Prototypes*. London: DH; 2015. URL: www.gov. uk/government/uploads/system/uploads/attachment\_data/file/395384/Reform\_Document.pdf (accessed 23 June 2015).
- 49. NHS England. *Building the NHS of the Five Year Forward View*. Redditch: NHS England; 2015. URL: www.england.nhs.uk/wp-content/uploads/2015/03/business-plan-mar15.pdf (accessed 23 June 2015).
- NHS Education for Scotland. Dental Workforce Report. Edinburgh: NES; 2014.
   URL: www.isdscotland.org/Health-Topics/Dental-Care/Dental-Workforce/2014-11-24-dental-workforce-report-final.pdf?1 (accessed 23 June 2015).
- 51. Birch S. Health human resource planning for the new millennium: inputs in the production of health, illness and recovery in populations. *Can J Res* 2002;**33**:109–14.
- 52. King NC. The new Dentists Act. Br Med J 1921;2:302. http://dx.doi.org/10.1136/bmj.2.3164.302-b
- 53. Great Britain. *Dentistry (Teviot Committee, Final Report)*. Cmnd. 6727. London: Her Majesty's Stationery Office; 1946.
- 54. Great Britain. The Dentists Act 1956. London: The Stationery Office; 1956.

- 55. Great Britain. *Draft Dental Auxiliaries (Amendment) Regulations 2002*. London: The Stationery Office; 2002. URL: www.publications.parliament.uk/pa/cm200102/cmstand/deleg4/st020508/20508s01.htm (accessed 20 February 2016).
- 56. Ross M, Ibbetson R, Turner S. The acceptability of dually-qualified dental hygienist-therapists to general dental practitioners in South-East Scotland. *Br Dent J* 2007;**202**:E8. http://dx.doi.org/10.1038/bdj.2007.45
- 57. General Dental Council (GDC). *Direct Access*. London: GDC; 2013. URL: www.gdc-uk.org/dentalprofessionals/standards/pages/direct-access.aspx. (accessed 20 February 2016).
- 58. General Dental Council (GDC). *Scope of Practice*. London: GDC; 2013. URL: www.gdc-uk.org/ Dentalprofessionals/Standards/Documents/Scope of Practice September 2013 (3).pdf (accessed 23 June 2015).
- 59. The Nuffield Institute. *Nuffield Report: The Education and Training of Personnel Auxiliary to Dentistry*. London: The Nuffield Institute; 1993.
- Laurant M, Harmsen M, Wollersheim H, Grol R, Faber M, Sibbald B. The impact of nonphysician clinicians do they improve the quality and cost-effectiveness of health care services? *Med Care Res Rev* 2009;66:36S–89S. http://dx.doi.org/10.1177/1077558709346277
- 61. Glick M, Monteiro da Silva O, Seeberger GK, Xu T, Pucca G, Williams DM, et al. FDI Vision 2020: shaping the future of oral health. Int Dent J 2012;62:278–91. http://dx.doi.org/10.1111/idj.12009
- 62. White DA, Tsakos G, Pitts NB, Fuller E, Douglas GVA, Murray JJ, et al. Adult Dental Health Survey 2009: common oral health conditions and their impact on the population. *Br Dent J* 2012;**213**:567–72. http://dx.doi.org/10.1038/sj.bdj.2012.1088
- 63. Watt RG, Steele JG, Treasure ET, White DA, Pitts NB, Murray JJ. Adult Dental Health Survey 2009: implications of findings for clinical practice and oral health policy. *Br Dent J* 2013;**214**:71–5. http://dx.doi.org/10.1038/sj.bdj.2013.50
- 64. Age UK. *Agenda for Later Life*. London: Age UK; 2014. URL: www.ageuk.org.uk/Documents/EN-GB/For-professionals/Policy/ID202014\_agenda\_for\_later\_life\_report\_2014.pdf?dtrk=true (accessed 23 June 2015).
- 65. Brocklehurst P, Macey R. Skill-mix in preventive practice will it help in the future? *BMC Oral Health* 2015;**15**(Suppl. 1):S10. http://dx.doi.org/10.1186/1472-6831-15-S1-S10
- 66. Nash DA, Friedman JW, Kardos TB, Kardos RL, Schwarz E, Satur J, et al. Dental therapists: a global perspective. Int Dent J 2008;**58**:61–70. http://dx.doi.org/10.1111/j.1875-595X.2008.tb00177.x
- 67. Johnson PM. International profiles of dental hygiene 1987 to 2006: a 21-nation comparative study. *Int Dent J* 2009;**59**:63–77.
- 68. Friedman JW. The international dental therapist: history and current status. *J Calif Dent Assoc* 2011;**39**:23–9.
- 69. Nash DA, Friedman JW, Mathu-Muju KR, Robinson PG, Satur J, Moffat S, et al. A review of the global literature on dental therapists. *Community Dent Oral Epidemiol* 2014;**42**:1–10. http://dx.doi.org/10.1111/cdoe.12052
- Kitchener M, Mertz E. Professional projects and institutional change in healthcare: the case of American dentistry. Soc Sci Med 2012;74:372–80. http://dx.doi.org/10.1016/ j.socscimed.2010.10.005
- 71. Galloway J, Gorham J, Lambert M, Richards D, Russell D, Russell I, et al. The Professionals Complementary to Dentistry: Systematic Review and Synthesis. York: University of York, Centre for Reviews and Dissemination; 2003.

- 72. Dyer TA, Brocklehurst P, Glenny AM, Davies L, Tickle M, Issac A, *et al.* Dental auxiliaries for dental care traditionally provided by dentists. *Cochrane Database Syst Rev* 2014;**8**:CD010076. http://dx.doi.org/10.1002/14651858.cd010076.pub2
- 73. Macey R, Glenny A, Walsh T, Tickle M, Worthington H, Ashley J, et al. The efficacy of screening for common dental diseases by hygiene-therapists a diagnostic test accuracy study. *J Dent Res* 2015;**94**(Suppl. 3):70S–78S. http://dx.doi.org/10.1177/0022034514567335
- 74. Brocklehurst P, Pemberton M, Macey R, Cotton C, Walsh T, Lewis M. Comparative accuracy of different members of the dental team in detecting malignant and non-malignant oral lesions. Br Dent J 2015;**218**:525–9. http://dx.doi.org/10.1038/sj.bdj.2015.344
- 75. Brocklehurst P, Ashley J, Walsh T, Tickle M. Relative performance of different dental professional groups in screening for occlusal caries. *Community Dent Oral Epidemiol* 2012;**40**:239–46. http://dx.doi.org/10.1111/j.1600-0528.2012.00671.x
- 76. Patel R, Sprod A, Harwood P, Drugan C. The use of dental therapists as examiners in dental epidemiological surveys. *Community Dent Health* 2012;**29**:195–7.
- 77. Kwan SYL, Prendergast MJ. The use of clinical dental auxiliaries as examiners in caries prevalence surveys in the United Kingdom: a feasibility study. *Community Dent Oral Epidemiol* 1998;**26**:194–200. http://dx.doi.org/10.1111/j.1600-0528.1998.tb01949.x
- 78. Wang NJ, Riordan PJ. Recall intervals, dental hygienists and quality in child dental care. *Community Dent Oral Epidemiol* 1995;**23**:8–14. http://dx.doi.org/10.1111/j.1600-0528.1995.tb00190.x
- 79. Wang NJ. Variation in clinical time spent by dentist and dental hygienist in child dental care. *Acta Odontol* 1994;**52**:280–9. http://dx.doi.org/10.3109/00016359409029040
- 80. Brocklehurst P, Mertz B, Jerkovic-Cosic K, Littlewood A, Tickle M. Direct access to midlevel dental providers: an evidence synthesis. *J Pub Health Dent* 2014;**74**:326–35. http://dx.doi.org/10.1111/jphd.12062
- 81. Gallagher J, Wright D. General dental practitioners' knowledge of and attitudes towards the employment of dental therapists in general practice. *Br Dent J* 2003;**194**:37–41. http://dx.doi.org/10.1038/sj.bdj.4802411
- 82. Hay I, Batchelor P. The future role of dental therapists in the UK: a survey of district dental officers and general practitioners in England and Wales. *Br Dent J* 1993;**175**:61–6. http://dx.doi.org/10.1038/sj.bdj.4808225
- 83. Jones G, Devalia R, Hunter L. Attitudes of general dental practitioners in Wales towards employing dental hygienist-therapists. *Br Dent J* 2007;**203**:E19. http://dx.doi.org/10.1038/bdj.2007.890
- 84. Ward P. The changing skill mix experiences on the introduction of the dental therapist into general dental practice. *Br Dent J* 2006;**200**:193–7. http://dx.doi.org/10.1038/sj.bdj.4813251
- 85. Newton J, Gibbons D. Vacant posts reported in dental practice: implications for human resource planning. *Br Dent J* 2002;**192**:37–9. http://dx.doi.org/10.1038/sj.bdj.4801285
- 86. Harris RV, Haycox A. The role of team dentistry in improving access to dental care in the UK. *Br Dent J* 2001;**190**:353–6. http://dx.doi.org/10.1038/sj.bdj.4800971a
- 87. Ireland R. Dental therapists: their future role in the dental team. Dent Update 1997;24:269.
- 88. Jones DE, Gibbons DE, Doughty JF. The worth of a therapist. *Br Dent J* 1981;**151**:127–8. http://dx.doi.org/10.1038/sj.bdj.4804650
- 89. Holt R, Murray J. An evaluation of the role of new cross auxiliaries and of their clinical contribution to the community dental service. part 2: analysis of day books. *Br Dent J* 1980;**149**:259–63. http://dx.doi.org/10.1038/sj.bdj.4804508

- 90. Douglass CW, Lipscomb J. Expanded function dental auxiliaries: potential for the supply of dental services in a national dental program. *J Dent Ed* 1979;**43**:556–67.
- 91. Harris R, Burnside G. The role of dental therapists working in four personal dental service pilots: type of patients seen, work undertaken and cost-effectiveness within the context of the dental practice. *Br Dent J* 2004;**197**:491–6. http://dx.doi.org/10.1038/sj.bdj.4811749
- 92. Godson JH, Williams SA, Csikar JI, Bradley S, Rowbotham JS. Dental therapy in the United Kingdom: part 2. A survey of reported working practices. *Br Dent J* 2009;**207**:417–23. http://dx.doi.org/10.1038/sj.bdj.2009.962
- 93. Dyer TA, Humphris G, Robinson PG. Public awareness and social acceptability of dental therapists. *Br Dent J* 2010;**208**:16–7. http://dx.doi.org/10.1038/sj.bdj.2010.1
- 94. Dyer TA, Robinson PG. Exploring the social acceptability of skill-mix in dentistry. *Int Dent J* 2008;**58**:173–80.
- 95. Sun N, Burnside G, Harris R. Patient satisfaction with care by dental therapists. *Br Dent J* 2010;**208**:E9-E.
- 96. Brocklehurst P, Tickle M. The policy context for skill mix in the National Health Service in the United Kingdom. *Br Dent J* 2011;**211**:265–9. http://dx.doi.org/10.1038/sj.bdj.2011.765
- 97. Adams A, Lugsden E, Chase J, Arber S, Bond S. Skill-mix changes and work intensification in nursing. *Work Employ Soc* 2000;**14**:541–55. http://dx.doi.org/10.1177/09500170022118563
- 98. Burke FJT, Wilson NHF, Christensen GJ, Cheung SW, Brunton PA. Contemporary dental practice in the UK: demographic data and practising arrangements. *Br Dent J* 2005;**198**:39–43. http://dx.doi.org/10.1038/sj.bdj.4811956
- 99. Harris RV, Sun N. Translation of remuneration arrangements into incentives to delegate to English dental therapists. *Health Policy* 2012;**104**:253–9. http://dx.doi.org/10.1016/j.healthpol.2011.11.013
- 100. Doyal L, Cameron A. Reshaping the NHS workforce: necessary changes are constrained by professional structures from the past. *BMJ* 2000;**320**:1023. http://dx.doi.org/10.1136/bmj.320.7241.1023
- 101. Watt R, McGlone P, Evans D, Boulton S, Jacobs J, Graham S, *et al.* The facilitating factors and barriers influencing change in dental practice in a sample of English general dental practitioners. *Br Dent J* 2004;**197**:485–9. http://dx.doi.org/10.1038/sj.bdj.4811748
- 102. McDonald R, Cheraghi-Sohi S, Sanders C, Ashcroft D. Professional status in a changing world: the case of medicines use reviews in English community pharmacy. *Soc Sci Med* 2010;**71**:451–8. http://dx.doi.org/10.1016/j.socscimed.2010.04.021
- 103. Cheraghi-Sohi S, McDonald R, Harrison S, Sanders C. Experience of contractual change in UK general practice: a qualitative study of salaried GPs. *Br J Gen Pract* 2012;**62**:282–7. http://dx.doi.org/10.3399/bjgp12X636128
- 104. Brocklehurst P, Price J, Glenny AM, Tickle M, Birch S, Mertz E, et al. The effect of different methods of remuneration on the behaviour of primary care dentists. *Cochrane Database Syst Rev* 2013;**11**:CD009853. http://dx.doi.org/10.1002/14651858.cd009853.pub2
- 105. Brocklehurst P, Bridgman C, Davies G. A qualitative evaluation of a local professional network programme 'Baby Teeth DO Matter'. *Comm Dent Health* 2013;**30**:241–8.
- 106. Tickle M, Brown P, Blinkhorn A, Jenner T. Comparing the ability of different area measures of socioeconomic status to segment a population according to caries prevalence. *Community Dent Health* 2000;**17**:138.

- 107. Leake J, Birch S. Public policy and the market for dental services. *Community Dent Oral Epidemiol* 2008;**36**:287–95. http://dx.doi.org/10.1111/j.1600-0528.2008.00438.x
- 108. Hart JT. The inverse care law. *Lancet* 1971;**297**:405–12. http://dx.doi.org/10.1016/ S0140-6736(71)92410-X
- 109. Tchicaya A, Lorentz N. Socioeconomic inequalities in the non-use of dental care in Europe. *Int J Equity Health* 2014;**13**:7. http://dx.doi.org/10.1186/1475-9276-13-7
- 110. Whittaker W, Birch S. Provider incentives and access to dental care: evaluating NHS reforms in England. *Soc Sci Med* 2012;**75**:2515–21. http://dx.doi.org/10.1016/j.socscimed.2012.09.035
- 111. Manning WG, Bailit HL, Benjamin B, Newhouse JP. The demand for dental care: evidence from a randomized trial in health insurance. *JADA* 1985;**110**:895–902. http://dx.doi.org/10.14219/jada.archive.1985.0031
- 112. Parkin D, Yule B. Patient charges and the demand for dental care in Scotland, 1962–81. *Applied Econ* 1988;**20**:229–42. http://dx.doi.org/10.1080/00036848800000007
- 113. Ross M, Turner S, Ibbetson R. The impact of teamworking on the knowledge and attitudes of final year dental students. *Br Dent J* 2009;**206**:163–7. http://dx.doi.org/10.1038/sj.bdj.2009.59
- 114. Eaton K, Harris M, Ross M, Arevalo C. A survey of dental hygienists in the United Kingdom in 2011. Part 1–demographics and working patterns as dental hygienists. *Br Dent J* 2012;**213**:E18. http://dx.doi.org/10.1038/sj.bdj.2012.1041
- 115. Jackson RJ, Baird WO, Worthington LS, Robinson PG. A survey to investigate shortfalls in the dental care professional (DCP) workforce in South Yorkshire in 2004. *Prim Dent Care* 2007;**14**:129–35. http://dx.doi.org/10.1308/135576107782144324
- 116. Sprod A, Boyles J. The workforce of professionals complementary to dentistry in the general dental services: a survey of general dental practices in the South West. *Br Dent J* 2003;**194**:389–97. http://dx.doi.org/10.1038/sj.bdj.4810005
- 117. Hillam D. A survey of hygienists qualifying from the Liverpool School of Dental Hygiene. *Br Dent J* 2000;**188**:150–3.
- 118. Hillam D. Career patterns of dental hygienists qualifying from the Liverpool Dental Hospital School of Dental Hygiene. *Br Dent J* 1989;**166**:310–11. http://dx.doi.org/10.1038/sj.bdj.4806812
- 119. Gibbons D, Corrigan M, Newton J. Dental hygienists: a national survey of dental hygienists: working patterns and job satisfaction. *Br Dent J* 2001;**190**:207–10. http://dx.doi.org/10.1038/sj.bdj.4800926
- 120. Turner S, Ross MK, Ibbetson RJ. Job satisfaction among dually qualified dental hygienist-therapists in UK primary care: a structural model. *Br Dent J* 2011;**210**:E5. http://dx.doi.org/10.1038/sj.bdj.2011.50
- 121. Dyson RG, Allen R, Camanho AS, Podinovski CS, Sarrico CS, Shale EA. Pitfalls and protocols in DEA. Eur J Operat Res 2001;**132**:245–259. http://dx.doi.org/10.1016/S0377-2217(00)00149-1
- 122. Thanassoulis E. *Introduction to the theory and application of data envelopment analysis*. New York, NY: Springer; 2001. http://dx.doi.org/10.1007/978-1-4615-1407-7
- 123. Simar L, Wilson PW. Two-stage DEA: caveat emptor. *J Product Anal* 2011;**36**:215–18. http://dx.doi.org/10.1007/s11123-011-0230-6
- 124. Scheffler RM, Kushman JE. A production function for dental services: estimation and economic implications. *South Econ J* 1977;**44**:25–35. http://dx.doi.org/10.2307/1057296

- 125. Ramanathan R. Supplier selection problem: integrating DEA with the approaches of total cost of ownership and AHP. *Supply Chain Manag* 2007;**12**:258–61. http://dx.doi.org/10.1108/13598540710759772
- 126. Banker RD, Charnes A, Cooper WW, Swarts J, Thomas D. An introduction to data envelopment analysis with some of its models and their uses. *Res Gov Non-profit Account* 1989;**5**:125–63.
- 127. Brown R. Mismanagement or mismeasurement? Pitfalls and protocols for DEA studies in the financial services sector. Eur J Operat Res 2006;174:1100–16. http://dx.doi.org/10.1016/j.ejor.2005.03.025
- 128. Gutacker N, Harris A, Brennan D, Hollingsworth B. The determinants of dentists' productivity and the measurement of output. *Soc Sci Med* 2015;**124**:76–84. http://dx.doi.org/10.1016/j.socscimed.2014.11.020
- 129. Gray AM. The production of dental care in the British National Health Service. *Scot J Politic Econ* 1982;**29**:59–74. http://dx.doi.org/10.1111/j.1467-9485.1982.tb00436.x
- 130. Braun V, Clarke V. Using thematic analysis in psychology. *Qual Res Psych* 2006;**3**:77–101. http://dx.doi.org/10.1191/1478088706qp063oa
- 131. Health and Social Care Board. *Proposal for a One Year Extension to the Contract with Oasis Dental Care to Provide Additional Dental Services in Northern Ireland*. URL: www.hscboardhscninet/board/meetings (acessed 23 June 2015).
- 132. Shen J, Wildman J, Steele J. Measuring and decomposing oral health inequalities in an UK population. *Community Dent Oral Epidemiol* 2013;**41**:481–9. http://dx.doi.org/10.1111/cdoe.12071
- 133. Northcott A, Brocklehurst P, Jerkovic-Cosic K, Reinders J, McDermott I, Tickle M. Direct access: lessons learnt from the Netherlands. *Br Dent J* 2013;**215**:607–10. http://dx.doi.org/10.1038/sj.bdj.2013.1193
- 134. Public Health England. Delivering Better Oral Health: An Evidence-Based Toolkit for Prevention. 2014. URL: www.gov.uk/government/publications/delivering-better-oral-health-an-evidence-based-toolkit-for-prevention (accessed 20 February 2016).
- 135. Halaby CN. Panel models in sociological research: theory into practice. *Ann Rev Soc* 2004;**30**:507–44. http://dx.doi.org/10.1146/annurev.soc.30.012703.110629
- 136. Schoen M. Methodology of capitation payment to group dental practice and effects of such payment on care. *Health Serv Rep* 1974;**89**:16–24. http://dx.doi.org/10.2307/4594971
- 137. Heckman JJ. Sample selection bias as a specification error. *Econometrica* 1979;**47**:153–61. http://dx.doi.org/10.2307/1912352
- 138. Lechner M. The estimation of causal effects by difference-in-difference methods. *Found Trends Economet* 2011;**4**:165–224. http://dx.doi.org/10.1561/0800000014
- 139. Locker D, Allen F. What do measures of 'oral health-related quality of life' measure? *Community Dent Oral Epidemiol* 2007;**35**:401–11. http://dx.doi.org/10.1111/j.1600-0528.2007.00418.x
- 140. Locker D, Quiñonezo C. To what extent do oral disorders compromise the quality of life? *Community Dent Oral Epidemiol* 2011;**39**:3–11. http://dx.doi.org/10.1111/j.1600-0528.2010.00597.x
- 141. Hausman JA. Specification tests in econometrics. *Econometrica* 1978;**46**:1251–71. http://dx.doi.org/10.2307/1913827
- 142. Grimshaw JM, Shirran L, Thomas RE, Mowatt G, Fraser C, Bero L, *et al.* Changing provider behaviour: an overview of systematic reviews of interventions. *Med Care* 2001;**39**:112–45. http://dx.doi.org/10.1097/00005650-200108002-00002

- 143. Moore GF, Audrey S, Barker M, Bond L, Bonell C, Hardeman W, *et al.* Process evaluation of complex interventions: Medical Research Council guidance. *BMJ* 2015;**350**:h1258. http://dx.doi.org/10.1136/bmj.h1258
- 144. Murray E, Treweek S, Pope C. Normalisation process theory: a framework for developing, evaluating and implementing complex interventions. *BMC Med* 2010;**8**:63. http://dx.doi.org/10.1186/1741-7015-8-63

# **Appendix 1** Analysis of additional sites

Sites from Scotland, Northern Ireland and from the community dental services were collected to understand the differences that different remuneration systems made to efficiency. Owing to the relatively low numbers collected, the results of the analysis are presented in this appendix, rather than in the main body of the report.

Tables 23 and 24 describe the sample statistics of NHS dental health-care output and input measures to the production of health care. A GDS practice in England had, on average, the most dentist hours on NHS patients (19.4 weekly sessions) followed by Scotland (17.2 weekly sessions), Northern Ireland (13.8 weekly sessions) and community dental services sites (13.3 weekly sessions). Community dental services sites are, however, the most likely to employ DCPs (80% of sample sites) followed by practices in England (68% of sample practices), Scotland (48% of sample practices) and Northern Ireland (34% of sample practices).

Table 25 describes the sample statistics for practice/site characteristics that may influence the production of health care. The number of surgeries is a direct input into the production of health care and is a measure of the capacity/size of each production unit. A community dental services site has on average the largest number of surgeries (3.7), followed by GDS practices in England (3.5), Scotland (2.9) and Northern Ireland (2.5). Patients who are exempt from payment charges are most likely to be found in community dental services sites (75.2% of the patient case mix), followed by Northern Ireland practices (59.9%), Scotland practices (46.4%) and England practices (37.3%). Children are a greater proportion of the patient case mix in community dental services sites (31.1%) than in practices in England (26.4%), Northern Ireland (13.7%) and Scotland (13.1%).

*Table 26* presents efficiency scores estimated by an input orientation model and *Table 27* describes efficiency scores estimated by output orientation.

TABLE 23 Sample statistics of NHS dental health-care output measures

	Sample	Sample							
Measure	England	Northern Ireland	Scotland	Community dental services sites					
UDAs, n (95% CI)	15,444 (13,433 to 17,455)	N/A	N/A	2336 (1761 to 2910)					
Income, £ (95% CI)	N/A	153,224 (102,808 to 203,639)	170,801 (105,848 to 235,754)	N/A					
Treatment plans, n (95% CI)	7286 (6367 to 8205)	3351 (2366 to 4337)	3724 (2516 to 4931)	1278 (975 to 1581)					
Patients seen, n (95% CI)	5597 (4905 to 6288)	2104 (1506 to 2704)	2811 (1992 to 3631)	1211 (827 to 1595)					

CI, confidence interval; N/A, not applicable for this jurisdiction.

**TABLE 24** Sample statistics of staffing measures

	Sample, mean (95%	% CI)		
Measure	England	Northern Ireland	Scotland	Community dental services sites
Number of weekly GDP sessions with NHS patients	19.4 (17 to 21.9)	13.8 (10.4 to 17.2)	17.2 (12.9 to 21.5)	13.3 (9.9 to 16.7)
Number of weekly PP sessions with NHS patients	6.1 (4.9 to 7.1)	6.5 (4.9 to 8.2)	5.6 (3.8 to 7.4)	N/A
Number of weekly associate GDP sessions with NHS patients	11.3 (9.1 to 13.4)	6.2 (3.1 to 9.3)	9.55 (5.2 to 13.8)	N/A
The percentage of DCP to GDP sessions seeing NHS patients	16.6 (12.6 to 34.6)	3.7 (0.3 to 7.1)	3.9 (0.7 to 7.2)	52.1 (34.1 to 70.2)
Any use of DCPs with NHS patients (%)	68	34	48	80
Number of FTE DCPs employed (to work with NHS or private patients)	1.1 (0.8 to 1.5)	0.2 (0.9 to 0.3)	0.3 (0.1 to 0.4)	5.2 (3.7 to 6.7)
Number of weekly DCP sessions with NHS patients	3.0 (2.1 to 3.8)	0.4 (0.02 to 0.9)	0.6 (0.03 to 1.2)	5.2 (3.7 to 6.7)
Number of weekly DH sessions with NHS patients	0.6 (0.3 to 0.8)	0.1 (0.03 to 0.2)	0.4 (0.04 to 0.7)	0.3 (0.01 to 0.62)
Number of weekly DHT sessions with NHS patients	1.7 (0.9 to 2.5)	0.3 (0.02 to 0.7)	0.01 (0.01 to 0.02)	0
Number of weekly DT sessions with NHS patients	0.7 (0.3 to 1.0)	0	0.3 (0.03 to 0.8)	3.6 (2.5 to 4.6)
Number of weekly nurse sessions	47.0 (42.1 to 52.0)	29.8 (23.2 to 36.5)	39.6 (29.8 to 49.4)	40.1 (30.5 to 49.7)
Number of weekly administration staff sessions	20.8 (18.3 to 23.3)	14.1 (10.8 to 17.5)	15.7 (11.0 to 20.3)	26.5 (20.8 to 32.3)

 $\mbox{CI, confidence interval; FTE, full-time equivalent; N/A, not applicable for this programme.}$ 

**TABLE 25** Sample statistics of practice/site measures

	Sample, mean (95% CI)								
Measure	England	Northern Ireland	Scotland	Community dental services sites					
Surgeries (n)	3.5 (3.2 to 3.8)	2.5 (1.9 to 3.0)	2.9 (2.2 to 3.5)	3.7 (2.9 to 4.6)					
Percentage of child patients	26.4 (22.7 to 30.1)	13.7 (7.1 to 20.2)	13.1 (8.7 to 17.6)	31.1 (20.3 to 41.1)					
Percentage of payment-exempt patients	37.3 (32.9 to 41.7)	59.9 (50.6 to 69.2)	46.4 (37.7 to 54.9)	75.2 (71.1 to 79.3)					
Percentage of band 1 treatments	57.3 (55.1 to 59.5)	N/A	N/A	33.2 (29.1 to 38.0)					
Percentage of band 2 treatments	28.4 (27.1 to 29.7)	N/A	N/A	43.2 (39.1 to 47.4)					
Percentage of band 3 treatments	4.7 (4.1 to 5.3)	N/A	N/A	16.1 (12.2 to 20.2)					
Percentage of urgent treatments	9.6 (7.7 to 11.6)	N/A	N/A	7.5 (3.1 to 11.4)					
Good Practice Award (n)	33	3.4	8.7	N/A					
Material costs (£)	11.7 (10.2 to 13.2)	12.9 (9.7 to 16.0)	16.4 (10.7 to 22.1)	N/A					
Laboratory costs (£)	9.7 (8.3 to 11.1)	12.1 (9.4 to 14.9)	11.9 (10.0 to 13.8)	N/A					

CI, confidence interval; N/A, not applicable for this jurisdiction/programme.

TABLE 26 Summary of efficiency scores estimated by input orientation

	Sample, mean efficiency score (range: minimum–maximum)								
Output measure	England ( <i>n</i> = 121)	Northern Ireland (n = 29)	Scotland ( <i>n</i> = 20)	Community dental services sites (n = 40)					
Annual number of UDAs	0.64 (0.19–1.00)	N/A	N/A	0.85 (0.35–1.00)					
Annual gross NHS income	N/A	0.80 (0.34–1.00)	0.94 (0.65–1.00)	N/A					
Annual number of treatment plans (claims)	0.63 (0.19–1.00)	0.81 (0.40–1.00)	0.91 (0.51–1.00)	0.82 (0.36–1.00)					
Annual number of patients seen	0.62 (0.19–1.00)	0.82 (0.40-1.00)	0.91 (0.59–1.00)	0.73 (0.21–1.00)					
N/A, not applicable for this jurisdic	tion/programme.								

TABLE 27 Summary of efficiency scores estimated by output orientation

	Sample, mean efficiency score (range: minimum–maximum)								
Output measure	England ( <i>n</i> = 121)	Northern Ireland (n = 29)	Scotland ( <i>n</i> = 20)	Community dental services sites (n = 40)					
Annual number of UDAs	0.48 (0.08–1.00)	N/A	N/A	0.82 (0.32–1.00)					
Annual gross NHS income	N/A	0.68 (0.22–1.00)	0.90 (0.46–1.00)	N/A					
Annual number of treatment plans (claims)	0.46 (0.09–1.00)	0.71 (0.21–1.00)	0.88 (0.51–1.00)	0.81 (0.43–1.00)					
Annual number of patients seen	0.45 (0.10–1.00)	0.74 (0.22–1.00)	0.89 (0.54–1.00)	0.68 (0.22–1.00)					
N/A, not applicable for this jurisdic	tion/programme.								

© Queen's Printer and Controller of HMSO 2016. This work was produced by Brocklehurst et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Practices are operating at a mean level of efficiency of 64% (England), 80% (Northern Ireland), 94% (Scotland) and 85% (community dental services sites) with input orientation. The mean efficiency score changes little when practice performance is measured by the effectiveness of the practices in reducing health-care inputs to see a fixed number of patients or treatment plans (output orientation).

The associations with efficiency scores of any use of DCPs with NHS patients are calculated for the sample of practices in each jurisdiction and service type in *Table 28* (UDAs as the health-care output), *Table 29* (treatment plans as the health-care output) and *Table 30* (patients seen as the health-care output). *Table 31* describes the findings for the sample of practices in each jurisdiction and service type when the measure is use of DCPs including sessions worked with all patients (NHS or private). This broadening of the measure of DCP use does not change the direction of the statistical associations identified above and the estimated coefficients on the use of DCPs remain similar. Any use of DCPs with NHS patients was found to be associated with 14% lower efficiency when the output measure was UDAs and 11% lower efficiency when the output measure was treatment plans or patients seen. All of these associations are statistically significant. The only other statistically significant association between efficiency and use of DCPs with NHS patients is found in the sample of community dental services sites in which use of DCPs is associated with a 21% lower efficiency score when health-care output is measured by the number of patients seen.

TABLE 28 Association with efficiency scores of any use of DCPs with NHS patients when the health-care output was UDAs/gross NHS income

	Sample								
	England	England N		Northern Ireland			Community dental services sites		
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	
Any use of DCPs with NHS patients	-0.140	< 0.00	-0.07	0.38	-0.0500	0.32	-0.130	0.16	
Weekly GDP sessions with NHS patients	-0.001	0.76	-0.01	< 0.00	0.0040	0.26	0.008	0.26	
Weekly support staff sessions	-0.005	< 0.00	-0.02	< 0.00	0.0050	0.13	-0.004	0.09	
Number of surgeries	-0.040	0.02	0.06	0.07	-0.0500	0.10	-0.020	0.51	
Proportion of adult non-elderly patients	-0.190	0.09	-0.68	0.02	-0.2800	0.11	-0.230	0.12	
Proportion with non-age-related exemption from NHS charges status	-0.130	0.34	0.22	0.31	-0.0001	0.97	0.150	0.64	

Note

Efficiency scores are calculated from an input orientation.

TABLE 29 Association with efficiency scores of any use of DCPs with NHS patients when the health-care output was treatment plans (claims)

	Sample									
	England I		Northern Ire	Northern Ireland		Scotland		dental s		
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value		
Any use of DCPs with NHS patients	-0.110	< 0.00	-0.04	0.65	-0.020	0.83	-0.110	0.21		
Weekly GDP sessions with NHS patients	-0.001	0.54	-0.02	< 0.00	-0.001	0.79	0.007	0.33		
Weekly support staff sessions	-0.006	< 0.00	-0.02	0.01	-0.004	0.47	-0.005	< 0.02		
Number of surgeries	-0.050	< 0.00	0.07	0.05	-0.020	0.72	-0.010	0.63		
Proportion of adult non-elderly patients	-0.150	0.11	1.10	0.09	0.180	0.49	-0.180	0.20		
Proportion with non-age-related exemption from NHS charges status	-0.380	0.01	0.20	0.27	-0.510	0.38	-0.110	0.70		

Note

Efficiency scores are calculated from an input orientation.

TABLE 30 Association with efficiency scores of any use of DCPs with NHS patients when the health-care output was patients seen

	Sample							
	England N		Northern Ire	Northern Ireland			Community dental services sites	
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
Any use of DCPs with NHS patients	-0.110	< 0.00	-0.03	0.70	-0.030	0.69	-0.210	0.02
Weekly GDP sessions with NHS patients	-0.002	0.21	-0.02	0.01	-0.001	0.77	-0.010	0.13
Weekly support staff sessions	-0.006	< 0.00	-0.01	0.02	-0.004	0.42	-0.001	0.50
Number of surgeries	-0.050	< 0.00	0.09	0.02	0.004	0.93	-0.007	0.78
Proportion of adult non-elderly patients	-0.160	0.09	0.50	0.12	0.090	0.70	0.180	0.22
Proportion with non-age-related exemption from NHS charges status	-0.290	0.04	0.10	0.66	-0.260	0.62	-0.090	0.76

Note

Efficiency scores are calculated from an input orientation.

TABLE 31 Associations with efficiency scores of the utilisation of DCPs with either NHS or private patients

		Sample						
	Utilisation of DCPs with either NHS or private	England		Northern Ireland		Scotland		Community
Health-care output		Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	dental services
UDAs/income	Any use of DCPs (on either NHS or private patients)	-0.14	< 0.00	-0.080	0.30	-0.03	0.66	N/A
Treatment plans	Any use of DCPs (on either NHS or private patients)	-0.12	< 0.00	-0.070	0.38	-0.02	0.82	N/A
Patients seen	Any use of DCPs (on either NHS or private patients)	-0.13	< 0.00	-0.080	0.31	-0.08	0.39	N/A
UDAs/income	Number of FTE DCPs employed (to work with NHS or private patients)	0.02	0.10	-0.007	0.58	-0.11	0.49	N/A
Treatment plans	Number of FTE DCPs employed (to work with NHS or private patients)	0.01	0.43	-0.010	0.30	-0.08	0.73	N/A
Patients seen	Number of FTE DCPs employed (to work with NHS or private patients)	0.01	0.32	-0.010	0.25	-0.33	0.15	N/A

FTE, full-time equivalent; N/A, not applicable for community dental services sites as DCP use is with NHS patients only.

Each row is a separate regression that includes control variables: weekly NHS GDP sessions, weekly administration staff sessions, Number of surgeries, proportion of adult non-elderly patients and proportion of patients with non-age-related exemption from NHS charges status.

The associations with efficiency scores of the level of use of DCPs with NHS patients is calculated for the sample of practices in each jurisdiction and service type in *Table 32* (UDAs as the health-care output), *Table 33* (treatment plans as the health-care output) and *Table 34* (patients seen as the health-care output).

There is no difference in the associations identified for the England sample except in the sample case study of community dental services sites. A site with an additional weekly session of DCP use in community dental services sites is associated with a 5% lower efficiency score (p < 0.05) when UDAs is the output measure and a 4% lower efficiency score (p < 0.05) when the number of treatment plans is the output measure, although there is no significant difference in efficiency scores among community dental services sites when the output measure is patients seen.

The association between efficiency scores and the proportion of clinical time provided by DCPs in practices/sites that employ DCPs is presented for each jurisdiction and service type in *Table 35* (UDAs as the health-care output), *Table 36* (treatment plans as the health-care output) and *Table 37* (patients seen as the health-care output). There is no difference in the case study samples in the non-statistically significant difference identified in the England sample. In all jurisdictions and service types (practices or community dental services sites), the ratio of DCP weekly sessions worked with NHS patients to GDPs' weekly sessions worked with NHS patients is not significantly associated with any measure of output after controlling for patient and practice characteristics.

TABLE 32 Association with efficiency scores of the level of use of DCPs with NHS patients when the health-care output was UDAs/gross NHS income

	Sample							
	England	England I		Northern Ireland		Scotland		dental s
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
Weekly DCP sessions with NHS patients	-0.004	0.51	0.020	0.43	-0.020	0.63	-0.05	< 0.00
Weekly GDP sessions with NHS patients	0.001	0.73	-0.010	0.56	0.007	0.35	0.02	0.04
Weekly support staff sessions	-0.005	< 0.00	-0.010	0.17	0.007	0.50	-0.005	0.01
Number of surgeries	-0.040	0.05	-0.002	0.98	-0.080	0.34	0.03	0.16
Proportion of adult non-elderly patients	-0.210	0.11	-2.030	0.12	-0.620	0.31	-0.38	0.01
Proportion with non-age-related exemption from NHS charges status	-0.090	0.58	0.710	0.11	0.005	0.53	0.24	0.43

Note

Efficiency scores are calculated from an input orientation.

**TABLE 33** Association with efficiency scores of the level of use of DCPs with NHS patients when the health-care output was treatment plans (claims)

	Sample									
	England !		Northern Ire	Northern Ireland		Scotland		Community dental services sites		
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value		
Weekly DCP sessions with NHS patients	-0.005	0.35	0.008	0.89	0.010	0.81	-0.040	< 0.00		
Weekly GDP sessions with NHS patients	0.001	0.79	-0.020	0.16	0.004	0.52	0.010	0.11		
Weekly support staff sessions	-0.006	< 0.00	-0.010	0.53	-0.006	0.58	-0.006	0.01		
Number of surgeries	-0.040	0.03	-0.010	0.95	-0.100	0.23	0.040	0.17		
Proportion of adult non-elderly patients	-0.160	0.16	2.500	0.36	0.470	0.46	-0.300	0.04		
Proportion with non-age-related exemption from NHS charges status	-0.320	0.08	0.670	0.30	-0.780	0.50	-0.090	0.77		

Note

Efficiency scores are calculated from an input orientation.

**TABLE 34** Association with efficiency scores of the level of use of DCPs with NHS patients when the health-care output was patients seen

	Sample							
	England		Northern Ireland Scotland		Scotland		Community services site	
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
Weekly DCP sessions with NHS patients	-0.0009	0.85	-0.020	0.77	-0.020	0.77	-0.020	0.10
Weekly GDP sessions with NHS patients	-0.0008	0.68	-0.010	0.68	0.002	0.79	-0.008	0.39
Weekly support staff sessions	-0.0060	< 0.00	-0.020	0.35	-0.001	0.91	-0.003	0.30
Number of surgeries	-0.0500	< 0.00	0.140	0.45	-0.020	0.80	0.020	0.54
Proportion of adult non-elderly patients	-0.1700	0.12	-0.400	0.86	0.080	0.92	0.120	0.49
Proportion with non-age-related exemption from NHS charges status	-0.2600	0.15	0.310	0.76	-0.140	0.91	0.010	0.98

Notes

Efficiency scores are calculated from an input orientation.

Regression was with GDS practices/community dental services sites that employed some level of use of DCPs with NHS patients.

TABLE 35 Association with efficiency scores of the proportion of clinical time provided by DCPs when the health-care output was UDAs/gross income

	Sample							
	England		Northern Ireland Scotland		Community dental services sites			
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
The percentage of DCP to GDP sessions seeing NHS patients	0.0005	0.09	-0.003	0.54	-0.003	0.72	-0.0500	0.67
Weekly support staff sessions	-0.0050	< 0.00	-0.020	0.11	0.006	0.58	-0.0030	0.27
Number of surgeries	-0.0400	< 0.00	0.060	0.47	-0.050	0.47	0.0002	0.99
Proportion of adult non-elderly patients	-0.1700	0.10	-2.690	0.01	-0.500	0.37	-0.1900	0.32
Proportion with non-age-related exemption from NHS charges status	-0.1400	0.35	0.560	0.10	0.004	0.64	-0.0400	0.91

#### Notes

Efficiency scores are calculated from an input orientation.

Regression was with GDS practices/community dental services sites that employed some level of use of DCPs with NHS patients.

**TABLE 36** Association with efficiency scores of the proportion of clinical time provided by DCPs when the health-care output was treatment courses (claims)

	Sample	Sample							
	England		Northern Ireland Scotland			Community services site			
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	
The percentage of DCP to GDP sessions seeing NHS patients	0.001	0.07	0.02	0.46	0.003	0.69	-0.060	0.62	
Weekly support staff sessions	-0.006	< 0.00	0.02	0.64	-0.006	0.55	-0.005	80.0	
Number of surgeries	-0.040	< 0.00	-0.30	0.44	-0.070	0.25	0.010	0.78	
Proportion of adult non-elderly patients	-0.120	0.15	2.85	0.51	0.510	0.35	-0.170	0.36	
Proportion with non-age-related exemption from NHS charges status	-0.400	0.02	1.05	0.20	-0.990	0.42	-0.320	0.36	

### Notes

Efficiency scores are calculated from an input orientation.

Regression was with GDS practices/community dental services sites that employed some level of use of DCPs with NHS patients.

TABLE 37 Association with efficiency scores of the proportion of clinical time provided by DCPs when the health-care output was patients seen

	Sample							
	England		Northern Ireland Scotland		Scotland		Community dental services sites	
Variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
The percentage of DCP to GDP sessions seeing NHS patients	0.001	0.06	-0.03	0.14	0.0001	0.99	0.200	0.12
Weekly support staff sessions	-0.006	< 0.00	-0.06	80.0	-0.002	0.89	-0.001	0.81
Number of surgeries	-0.050	< 0.00	0.41	0.11	-0.010	0.89	-0.060	0.10
Proportion of adult non-elderly patients	-0.170	0.04	-3.34	0.07	0.240	0.71	0.270	0.15
Proportion with non-age-related exemption from NHS charges status	-0.300	0.06	0.02	0.98	-0.160	0.90	0.160	0.64

#### Notes

Efficiency scores are calculated from an input orientation.

Regression was with GDS practices/community dental services sites that employed some level of use of DCPs with NHS patients.

Table 38 presents the estimated coefficients on the interaction term between the level of DCP use and the characteristics of the patient population for all jurisdictions and service types. There is no difference in the case study samples in the non-statistically significant interaction terms identified in the England sample. No significant interactions with patient population characteristics were found, indicating that the association between DCP use and efficiency is not related to the type of population being served.

The associations between routine services and the level of use of DCPs, and the associations between routine services and the proportion of clinical time provided by DCPs are presented in *Tables 39* and *40*, respectively.

Among the case study samples, the only significant association between routine services and the level of use of DCPs was found in the community dental services sample, in which the proportion of band 2 treatments was larger (p < 0.05). One more weekly session of DCP use for NHS patients is associated with a 1% higher proportion of band 2 treatments delivered after controlling for practice and/or patient characteristics.

DOI: 10.3310/hsdr04220

TABLE 38 Mediating associations with efficiency scores between the level of DCP use and patient case-mix variables

		Sample							
		England		Northern Ireland		Scotland		Community dental services sites	
Output measure	Mediating variable	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value	Coefficient	<i>p</i> -value
UDA/income	Proportion of adult non-elderly patients	0.08	0.10	-2.46	0.31	-0.33	0.80	0.02	0.57
	Proportion with non-age-related exemption from NHS charges status	0.03	0.21	-0.58	0.49	0.02	0.25	-0.04	0.66
Claims	Proportion of adult non-elderly patients	0.05	0.24	0.04	0.99	0.02	0.98	0.04	0.31
	Proportion with non-age-related exemption from NHS charges status	0.05	0.08	-4.30	0.06	2.64	0.29	-0.12	0.21
Patients seen	Proportion of adult non-elderly patients	0.06	0.09	4.52	0.67	-0.43	0.59	0.004	0.93
	Proportion with non-age-related exemption from NHS charges status (95% CI)	0.05 (-0.01 to 0.10)	0.08	-3.44 (-6.72 to -0.16)	0.05	5.09	0.18	0.042	0.72

CI, confidence interval.

TABLE 39 Associations between the numbers of routine services delivered in the NHS and the level of DCP use on NHS patients

	Sample	Sample							
	England		Northern Ire	land	Scotland		Community dental services sites		
Outcome variable (routine services)	Coefficient: number of DCP NHS sessions	<i>p</i> -value							
Proportion of band 1 treatments	-0.0001	0.99	N/A	N/A	N/A	N/A	0.004	0.67	
Proportion of band 2 treatments	0.0010	0.51	N/A	N/A	N/A	N/A	0.01	0.01	
Percentage of preventative treatments (scale and polish, fluoride varnish)	2.46	0.10	-0.31	0.86	0.54	0.78	0.67	0.61	
Proportion of examinations	N/A	N/A	0.69	0.64	-1.10	0.06	1.32	0.42	
Proportion of fillings	N/A	N/A	-3.75	0.61	-3.24	0.07	0.04	0.93	

N/A, not applicable for this jurisdiction/programme.

#### Note

Each row is a separate regression that includes control variables: weekly NHS GDP sessions, weekly administration staff sessions, number of surgeries, proportion of adult non-elderly patients and proportion of patients with non-age-related exemption from NHS charges status.

TABLE 40 Associations between routine services delivered in the NHS and the proportion of DCPs' clinical time relative to GDPs' clinical time working with NHS patients

	Sample							
	England		Northern Ire	land	Scotland		Community services site	
Outcome variable (routine services)	Coefficient: number of DCP NHS sessions	<i>p</i> -value						
Proportion of band 1 treatments	-0.00003	0.86	N/A	N/A	N/A	N/A	0.05	0.50
Proportion of band 2 treatments	-0.00001	0.90	N/A	N/A	N/A	N/A	-0.09	0.06
Percentage of preventative treatments (scale and polish, fluoride varnish)	0.36	0.00	0.11	0.65	-0.02	0.96	2.37	0.84
Proportion of examinations	N/A	N/A	-0.31	0.06	-0.19	0.11	0.63	0.96
Proportion of fillings	N/A	N/A	-0.50	0.15	-0.52	0.18	-1.36	0.70

N/A, not applicable for this jurisdiction/programme.

### Note

Each row is a separate regression that includes control variables: weekly NHS GDP sessions, weekly administration staff sessions, number of surgeries, proportion of adult non-elderly patients and proportion of patients with non-age-related exemption from NHS charges status.

# **Appendix 2** Dental team questionnaire

ID:	

### DENTAL TEAM QUESTIONNAIRE

This is a questionnaire that has been designed to help us understand how dental teams work and the types of activity that they undertake. The questionnaire has been piloted with fellow practitioners to ensure that it is fit-for-purpose, whilst being sensitive to the commercial interests of the practice. It has received full ethics approval (University of Manchester Research Ethics Committee reference number 14028). Given the type of questions that are asked, either the practice owner or the practice manager should complete the questionnaire.

It will not be possible for the research team to identify persons who complete this questionnaire. All the data recorded on the form will be *anonymised independently* from the research team so it will not be possible to identify the practice to which the data belongs:

- The answers to the questionnaire will be entered onto a password protected Excel file, totally independent to the research team. Upon receipt of the questionnaire you will be sent a £75 voucher to thank you for your participation;
- Your contact details will then be sent to the relevant NHS Information Centre to link the Excel file to your activity data, again independent to the research team;
- Your contact details will then be removed from the Excel file and destroyed, before the NHS Information Centre sends the linked, anonymised data to the research team;
- All of the data collected in the questionnaire will be held in compliance with the Data Protection Act 1998 and will be processed in accordance with the rights under this Act;
- The data will only be used for the analyses that have been approved by the Ethics Committee;

Please complete the questionnaire and return it in the envelope provided. You will be sent a £75 voucher for your practice *when the fully completed* questionnaire is returned (this includes the full completion of the consent form).

## **SECTION 1**

Please answer all of the following questions:	
1. Is your practice a dental body corporate? Y	N (please circle)
2. How many surgeries are typically in operation	on (for at least 3 days / week)?
<ul><li>3. Has your practice been awarded the British I</li><li>/ N (please circle)</li></ul>	Dental Association's Good Practice Award? <b>Y</b>
• If yes, what is your most recent award	status: gold, silver, bronze?
4. Please describe your practice by detailing th have (you'll get to describe the number of clini	
• Nurses	
• Practice managers	
Other administrative staff	
5. How long (in minutes) do you tend to book	for a routine examination?
• For a regular patient:	
• For a new patient (non-emergency):	

6. As a	6. As a proportion of total annual income, what do you typically spend on:							
•	• Consumables and materials per year?%							
•	Laboratory fees per year?		%					
	7. What would be a typical remuneration rate for a session for the following members of staff (either that you actually pay or what you'd be prepared to pay)?							
•	Hygienist		£ per session					
•	Hygiene-therapist		£ per session					
•	Therapist		£ per session					

**SECTION 2** 

Please complete the table below, with each row representing a separate **dentist** at your practice:

Dentist	Please indicate	Please indicate	Please indicate	Please indicate
number	whether each	the dentist's	how many	the
	dentist is a	<b>UDA</b> target for	sessions* the	approximate
	principal (P), an	the year	dentist works	proportion of
	associate (A), a	2013/14?	in the practice	this time that is
	salaried dentist		in a typical	spent on
	(S), assistant (As)		week?	treating NHS
	or Dental			patients?
	Foundation			
	Trainee (DF)?			
1				%
2				%
3				%
4				%
5				%
6				%
7				%
8				%

<sup>\*</sup>session = half-day

Please complete the table below, with each row representing either a hygienist, hygiene-therapist or therapist that you employ in your practice (**we do not require information on dental nurses**). Pease leave the table blank if you employ none of these types of staff.

DCP	Please indicate the	Please indicate	Please indicate
number	type of DCP that	how many	the approximate
	works at your	sessions* the	proportion of
	practice: hygienist	DCP works in	this time that is
	(H), hygiene-	the practice in a	spent on treating
	therapist (H-T)	typical week?	NHS patients?
	and therapist (T)?		
1			%
2			%
3			%
4			%
5			%
6			%
7			%
8			%
9			%
10			%

\*session = half-day

### THANK YOU FOR YOUR TIME, IT IS VERY MUCH APPRECIATED

PLEASE RETURN THIS QUESTIONNAIRE IN THE ENCLOSED ENVELOPE

THE DATA RECORDED ABOVE REPRESENTS IMPORTANT INFORMATION THAT WILL BE TREATED ANONYMOUSLY AND WILL HELP US UNDERSTAND HOW DIFFERENT DENTAL TEAMS ARE UTILISED.

WE'D ALSO LIKE TO COLLECT RELEVANT AND ANONYMISED ACTIVITY
DATA FOR THE PRACTICE. THIS WILL BE SOURCED FROM THE RELEVANT
NHS INFORMATION CENTRE FOR A SINGLE ANNUAL PERIOD.

ON RECEIPT OF THE FULLY COMPLETED QUESTIONNAIRE (including the consent form), YOU WILL RECEIVE YOUR £75 VOUCHER BY POST

# **Appendix 3** Analysis of Oasis data

his appendix contains the descriptive statistics of the Oasis and control group.

Table 41 describes practice characteristics and patient selection variables for Oasis practices in comparison with the control practices. Oasis practices had greater staffing capacity to deliver health care than control practices, with, on average, 0.89 more GDPs active per practice per month. The mean number of monthly treatment items delivered was larger among control practices (922 treatments items) than among Oasis practices (811 treatment items) and this difference between the two groups was statistically significant at a 5% level. The mean monthly number of treatment plans delivered to patients was larger for an Oasis practice (463 treatment plans) than a control practice (392 treatment plans), whereas the mean value of the treatment plans delivered was lower in Oasis practices than in control practices by £10.83.

On average, in control group practices, 2.01% more of the practice population was made up of patients aged  $\geq$  60 years; in Oasis practices, 6.55% more of patients seen were exempt from payment charges for reasons associated with high dental care need. However, Oasis practices are, on average, located in areas of lower deprivation than control practices.

TABLE 41 Sample comparison of Oasis practices with control practices for practice characteristics and patient selection variables

	Average for						
	Control	Control		Oasis		Difference	
Variable name	Mean (n)	95% CI	Mean (n)	95% CI	Mean	<i>p</i> -value	
Practice characteristics							
Number of GDPs	3.21 (2079)	3.15 to 3.27	4.11 (478)	3.91 to 4.31	-0.89	< 0.00	
Regional deprivation	21.2 (2415)	21 to 21.47	19.8 (556)	19.4 to 20.3	1.41	< 0.00	
Monthly number of treatment items	922 (2079)	895 to 948	811 (478)	766 to 857	110	< 0.00	
Monthly number of treatment plans (claims)	392 (2079)	381 to 402	463 (478)	435 to 492	-71.41	< 0.00	
Monthly number of registrations	3359 (2415)	3282 to 3437	3850 (556)	3665 to 4034	-490	< 0.00	
Monthly number of patients seen	306 (2415)	297 to 314	320 (556)	302 to 339	-14.4	0.15	
Patient selection							
Percentage of registered patients aged ≥ 60 years	15.3 (2079)	15.1 to 15.6	13.3 (478)	13 to 13.6	2.01	< 0.00	
Percentage of registered patients that are children	30.7 (2415)	30.3 to 31.1	19.9 (556)	19.7 to 20.2	10.7	< 0.00	
Percentage of registered patients exempt from payment charges	43.2 (2415)	42.4 to 44.0	36.7 (556)	35.9 to 37.3	6.6	< 0.00	
CI, confidence interval.							

© Queen's Printer and Controller of HMSO 2016. This work was produced by Brocklehurst et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton S016 7NS, UK.

Table 42 describes receipt of care among registrants, mix of treatments and financial viability variables for Oasis practices in comparison with the control practices. Oasis practices deliver care over a larger patient population. An Oasis practice has, on average, 490 more registered patients and treats an average of 14 more unique patients per month than control group practices. This suggests that, although Oasis practices were purposely set up in locations where the government had identified barriers which prevented access to oral health care in the resident population, the practices nonetheless facilitated access to health-care services to more patients than neighbouring control group practices.

TABLE 42 Sample comparison of Oasis practices with control practices for receipt of care among registrants, mix of treatments and financial viability variables

	Average for practice					
	Control		Control		Difference	
Variable name	Mean (nª)	95% CI	Mean ( <i>n</i> ª)	95% CI	Mean difference	<i>p</i> -value
Receipt of care among regis	strants					
Monthly number of patients seen per 100 registrations	8.9 (2415)	8.8 to 9	8.4 (556)	8.1 to 8.6	0.54	< 0.00
Monthly number of courses per 100 registrations	11.7 (2079)	11.6 to 11.9	12.5 (478)	12.1 to 12.9	-0.8	< 0.00
Value of treatment per treatment course (f)	33.8 (2079)	33.4 to 34.2	22.9 (478)	22.4 to 23.5	10.8	< 0.00
Mix of treatments						
Monthly number of examinations per 100 patients seen	66.7 (2079)	66.2 to 67.1	62.4 (478)	61.6 to 63.2	4.30	< 0.00
Monthly number of extractions per 100 patients seen	16.1 (2079)	15.8 to 16.4	20.7 (478)	20 to 21.4	-4.6	< 0.00
Monthly number of fillings per 100 patients seen	60.3 (2079)	59.6 to 60.9	44.3 (478)	43.3 to 45.4	15.9	< 0.00
Monthly number of scale and polishes per 100 patients seen	48.2 (2079)	47.6 to 48.7	40 (478)	39 to 41.1	8.1	< 0.00
Monthly number of fluoride varnishes per 100 patients seen	0.023 (2079)	0.015 to 0.032	0.018 (478)	0.008 to 0.029	0.005	0.60
Monthly number of fluoride varnishes per 100 child registrations	0.008 (2079)	0.005 to 0.01	0.007 (478)	0.003 to 0.01	0.0007	0.83
Financial viability						
Patient payment charge revenue per registration (£)	11.5 (2079)	11.1 to 11.8	15.4 (478)	15.1 to 15.7	-3.93	< 0.00
Patient payment charge revenue treatment course (£)	9.1 (2079)	8.8 to 9.41	10.5 (478)	10.2 to 10.8	-1.4	< 0.00
CI, confidence interval.						

a One observation is one practice observed in 1 month.

The revenues accrued to the practice from patient payment charges per registration and per patient seen are both higher for an Oasis practice by £3.93 and £1.37, respectively. Oasis practices deliver an average of 4.6 more extractions per 100 patients seen than control practices. Other types of treatments are less frequently delivered in Oasis practices. These differences are statistically significant, which indicates different service mixes between Oasis and control practices.

A limitation of the pooled OLS approach is the potential bias for unobserved individual heterogeneity on estimators. Each dental practice has its own unique individual characteristics, not all of which are observed in the data but may influence or bias the explanatory variables. For example, the tenure of GDPs could influence the amount of health care they provide, the number of nearby dental practices could influence the amount of patient registrations, business expertise may influence the size of the practice; use of DCPs could influence the treatment service mix. Not controlling for unobserved practice specific effects leads to potential bias in the resulting estimates. In panel estimation, the influence of time-invariant practice characteristics is assumed to be captured by the correlation between the error term for the practice and explanatory variables.

Fixed- and random-effects estimators are alternative estimators for the panel model. Each has advantages and disadvantages. However, in this study random effects is appropriate because calculating an estimate of the matching variable (the estimated coefficient of matching variable indicates the effect on an outcome of belonging to either the Oasis or control group) is not possible with a fixed-effects estimator as that variable is time invariant. We use the Hausman test<sup>141</sup> to test the null hypothesis that the coefficients estimated by a more efficient model (random effects) are the same as the ones estimated by a less efficient but consistent model (fixed effects). At a 5% significance level the null hypothesis was accepted, which means that random effects are preferable to fixed effects.

A problem with the random-effects model is that it assumes that the practice-specific effect is a random variable that is uncorrelated with the explanatory variables of all past, current and future time periods of the same practice. This assumption is false if, for example, the ability of a practice to deliver oral health care is learnt from their experience delivering care to patients. Then the practice-specific effect would not be random but related to types of patients they have seen (patient case-mix explanatory variables). The assumption would also be false if the ability of a practice to deliver health care arises from its management structure, organisational culture and workplace environment. Those are in part constituted by the number and type of staff members that the practice has employed, so the practice-specific effect would be correlated with the explanatory variables measuring staffing levels.

To relax this assumption, we estimate the random-effects model with a Mundlak correction. This approach is often used with random-effects models and is conducted by adding group means of the time-varying explanatory variables to the model. The results of the estimators are now divided between the current effect and the long-term effect of the explanatory variables on the outcome variable. The coefficient on the group means of the time-varying explanatory variables is the longer-term effect of those variables on the current outcome, while the coefficient on the time-varying explanatory variables is the current effect of those variables on current outcome. The current effect of the time-varying explanatory variables should be less than the 'total' (current and long-term) effect if the model holds true. Hence, if the correct specification is the Mundlak-corrected random-effects model, the coefficients on its time-varying explanatory variables (current effect) should be of smaller magnitude than the coefficients in the OLS model and random-effects model on the time-varying explanatory variables ('total' effect). Table 43 shows the coefficients and statistical significance of exploratory variables to be almost identical across the three model specifications. A joint Wald test did not reject the null hypothesis that the coefficients on time-varying explanatory variables are identical. This suggests that there is not enough variation across the months within practices to justify a panel approach and we chose pooled OLS with robust standard errors as our main model specification.

TABLE 43 Separate regressions showing the differences in the outcome variables between Oasis and control practices for each group of RQs when estimated using OLS with robust standard errors, random effects and random effects with the Mundlak correction

	OLS		Random ef	fects	Random ef Mundlak co	
Dependent variable (i.e. outcome measure)	Matching variable estimate	<i>p</i> -value	Matching variable estimate	<i>p</i> -value	Matching variable estimate	<i>p</i> -value
Patient selection						
Percentage of registered patients aged ≥ 60 years	-1.72	0.35	-2.28	0.20	-1.73	0.35
Percentage of registered patient that are children	-11.2	< 0.00	-11.04	< 0.00	-11.16	< 0.00
Percentage of registered patients exempt from payment charges for reasons associated with high dental care need	-3.81	0.21	-3.78	0.50	-3.56	0.53
Receipt of care among registrants						
Patients seen per 100 registrations	-1.61	< 0.00	-1.85	< 0.00	-1.51	< 0.00
Treatment courses per 100 registrations	0.69	0.25	-0.22	0.70	0.69	0.25
Value of treatment per claim (£)	-14.29	< 0.00	-14.44	< 0.00	-14.29	< 0.00
Mix of treatments						
Examination per 100 patients seen	-9.76	< 0.00	-9.50	< 0.00	-9.76	< 0.00
Extractions per 100 patients seen	6.20	< 0.00	7.18	< 0.00	6.20	< 0.00
Fillings per 100 patients seen	-17.89	< 0.00	-16.90	< 0.00	-17.89	< 0.00
Scale and polish per 100 patients seen	-11.47	< 0.00	-14.45	< 0.00	-11.47	< 0.00
Fluoride varnish per 100 patients seen	0.034	0.41	0.017	0.66	0.034	0.41
Fluoride varnish per 100 child registrations	0.008	0.53	0.005	0.65	0.008	0.53
Financial viability						
Patient payment charge revenue per registration (£)	-0.53	0.63	-1.42	0.14	-0.53	0.63
Patient payment charge revenue per treatment course (£)	-2.85	< 0.00	-3.02	< 0.00	-2.85	< 0.00

# **Appendix 4** Topic guide and coding frame

Theoretical frameworks are increasingly used for process evaluation and to determine the barriers to, and enablers of, implementation.<sup>142–144</sup> As a result, normalisation process theory was used as a means of framing the development of the topic guides to understand the barriers to, and enablers of, role substitution and how such practices could become normalised.

The following domains were considered:144

- 1. Coherence (i.e. meaning and sense-making by participants)
  - Is the intervention easy to describe?
  - Is it clearly distinct from other interventions?
  - Does it have a clear purpose for all relevant participants?
  - Do participants have a shared sense of its purpose?
  - What benefits will the intervention bring and to whom?
  - Are these benefits likely to be valued by potential participants?
  - Will it fit with the overall goals and activity of the organisation?
- 2. Cognitive participation (i.e. commitment and engagement by participants)
  - Are target user groups likely to think it is a good idea?
  - Will they see the point of the intervention easily?
  - Will they be prepared to invest time, energy and work in it?
- 3. Collective action (i.e. the work participants do to make the intervention function)
  - How compatible is it with existing work practices?
  - What impact will it have on division of labour, resources, power and responsibility between different professional groups?
  - Will it fit with the overall goals and activity of the organisation?
- 4. Reflexive monitoring (i.e. participants reflect on or appraise the intervention)
  - How are users likely to perceive the intervention once it has been in use for a while?
  - Is it likely to be perceived as advantageous for patients or staff?
  - Can users/staff contribute feedback about the intervention once it is in use?
  - Can the intervention be adapted or improved on the basis of experience?

### **Topic guide: general dental practitioners**

- 1. What do you see as the advantages and disadvantages of using DCPs at the practice?
- 2. What model of DCPs and dentists do you currently have in your practice?
  - Why is this?
  - How well do you think this model works?

- 3. What do you perceive to be the optimal combination of DCPs and dentists in your practice?
- 4. Could DCPs be utilised more in practice?
  - In what way?
  - What are the potential barriers?
- 5. Have your views on the use of DCPs in the practice team changed over time?
  - How?
  - Why do you think this is?
  - Have the views of your colleagues changed?
  - What about new associates?
- 6. What do you think about the duties that DCPs can carry out under their current scope of practice?
  - Should they be doing more/less?
  - Does it work well in practice?
- 7. What do you think about the recent expansion in the scope of practice of DCPs?
  - Has it changed the way you have chosen to utilise DCPs at this practice?
  - What are your dentist colleagues' views about the change?
  - What are your DCPs' views about the change?
  - What are your views about direct access?
- 8. How do patients respond when you refer them to DCPs?
  - For treatment and for hygiene/advice?
  - Are patients aware of the role of DCPs?
  - Which way is best to communicate this to patients?
  - What about for more complex treatments?
- 9. How does the way that dentists get paid in the NHS affect the dental team?
  - Does it affect the types of tasks that you refer to DCPs?
- 10. Has using DCPs freed up your time to concentrate on more complex cases?
  - Does this happen often at the practice?
  - If so, how often?
  - What kind of tasks?
  - Are you happy with seeing more complex cases rather than the routine care?
- 11. What would prompt you to use DCPs more at the practice?
- 12. How do you think that your use of DCPs over the last few years has affected the profitability of this practice?
- 13. Where do you think we will be in ten years' time in respect to the DCPs' role in delivering dental health care?
- 14. How do you think the changing oral health needs of the population will affect the way you organise your practice in the future?

## **Topic guide: dental care professionals**

- 1. How are DCPs used in your practice at the moment?
  - Who completes which tasks?
  - How do you refer to the dentist?
  - How do they refer to you?
  - Associates? Do they refer to you?
- 2. What do you think about the recent expansion in the scope of practice of DCPs?
  - Has it changed your practice?
  - What are your colleague's views about the change?
- 3. Could DCPs be utilised more in practice?
  - Could you do more?
  - Would you want to do more?
- 4. What are your views about direct access?
  - How successful has this been in your practice?
  - Do you think it has changed the way that care is delivered?
  - What are your colleagues' views about the change?
  - Do you think this will change over time?
- 5. What response do you get from patients that come to see you?
  - Do they accept receiving care from the whole dental team, rather than just the dentist?
- 6. Do you think they are aware of the different types of roles that DCPs can undertake?
  - How does your practice communicate this to patients?
- 7. Would changes to the way that the NHS pays the dental team affect the types of tasks that you undertake?
- 8. How happy are you with the level of support you have in the surgery?
  - Dental nurse?
- 9. Does using DCPs help to free up time for dentists to concentrate on more complex cases?
  - Does this happen in practice?
- 10. How confident are you about performing dental procedures without oversight from a dentist?
  - Which tasks?
- 11. What would prompt dentists to use DCPs more at the practice?
- 12. Where do you think we will be in ten years' time in terms of the role of DCPs?

## **Topic guide: patients**

To open: how long have you been coming to this practice?

- 1. When you last saw or were treated by a DHT what treatment did you receive?
  - How did you feel about receiving this treatment from your DHT rather than from your dentist?
  - Was it different seeing a DHT rather than a dentist?
  - Were you happy with the service you received?
- 2. How much do you know about the roles and duties of a DHT?
  - Are there any tasks that only a dentist can do?
- 3. If, in the future, your dentist referred you to a DHT for a filling how would you feel?
- 4. How do you think other patients at this practice would feel about seeing DHTs instead of dentists?
- 5. What things could your practice do to help patients manage the change from seeing the dentist regularly to seeing the DHT?
- 6. How much information do you think patients would want or need about the role of DHTs?
- 7. What do you think is the best way to communicate this type of information to patients?

**TABLE 44** Coding frame: GDPs

Theme	Codes	Example quote
Attitudes and beliefs	Benefits to the practice and GDPs	The biggest advantage is it frees up more of my time to more involved treatments, you know, endos, crowns, cosmetic treatment  PP2
	Perceptions of patient awareness	We actually had more people come to us via the direct access hygiene route, than we did via the traditional check-up and referral route  PPT_7
	Perceptions of patient acceptability	Yes, I mean they seem absolutely fine to see the therapist or hygienist for hygiene treatment because it seems fine, you say, oh, I want you to see the hygienist, absolutely fine. You say I want you to see a therapist for fillings, oh no, I want you to do it. So there's some sort of block that patients have got but they do understand the value of the hygienist for some reason  PP2
	GDP pitch of referral	We tend to say, I think your teeth need cleaning and at the same time of being cleaned the therapist can do the filling for you. They say, well I need to come back one more time for a clean up and a filling. That's how I sell it  PP5
	Leadership	I think it's both. I think basically we tried to explain to patients why we had the therapist here, for what reason, and I think once they'd seen them and realised that they had this extra time, they had more chance to ask questions, whatever else, they felt more comfortable and then people warm to the idea
		PP1
	Low awareness of DCP roles and skills	I don't know how it works. Are they limited to NHS? How does it all work? I know like they're not allowed to do a check up. A dentist has to do a check up
		PP4
	Value of DCPs	Only benefit for us might be that we could reduce the prices for patients a bit, because the hourly rate would be different. But otherwise, we just change them an hourly rate anyway, so as long as patients are prepared to pay us our hourly rate, we'll do the fillings for them  PP3

TABLE 44 Coding frame: GDPs (continued)

Theme	Codes	Example quote
	Patient safety – DCPs may miss something	The skill is diagnosis, but the actual treatment is the easy part. As long as you're good with the patient and you are competent in skills, you can teach anyone to do dentistry, but diagnosis is the hardest part because that comes with experience  PP5
	Professional threat from DCPs	And I think lots of dentists feel threatened  PP4
Structural issues	Financial and payment issues	There are problems right now, from what I understand. The entire remuneration system is based on the dentist with the performer number, taking for the UDAs that the treatment incurs in total, and then distributing, in inverted commas, fairly, to the team members as appropriate
	Referral patterns	It makes therapists, hygienists, look silly to patients. This way of working makes them look silly. Oh, I can't carry on with your treatment because I have to go and get a prescription to give your local anaesthetic. It hardly flows  PPT_7

**TABLE 45** Coding frame: DCPs

Theme	Codes	Example quote
Structural/ organisational barriers	Facilities and nurse support in clinics	I've always got a dental nurse with me and the dentists are all very happy to (look at referred patients), if one is busy I can look to another DCP1
	Scope of practice – lack of clarity/way it works at present	I was made redundant from a therapy job in a corporate earlier in the year because of the fact that I can't apply fluoride every 3 months without a prescription, and it just didn't create that workflow that we needed DCPT_1
	Financial – payment and patterns of referral	All of them are supposed to refer to me but some are really good at it, some aren't very good at it relying on them to refer patients to me DCPT_4
	Professional culture/ identities	Anyone over a certain age would just go in, a dental what? You know, and then they were just not interested, not interested at all. They just saw it as we were trying to steal their jobs and that was it  DCPT_5
	Skill level and experience/confidence	And I don't know how hand on heart, how I would feel completely in a practice on my own. I think I'm probably not the right person to be in practice completely on their own  DCP3
	Workplace culture/PP leadership	Yeah. So it's the team approach. And I don't know how they do it in many practices and how they keep an accurate check on it, and how they do a lot of monitoring at this practice because we have a deputy manager on reception. Our receptionists have to be – and this is another DCP progression route – our receptionists are all dental nurses, or have been dental nurses, experienced dental nurses, so they know what it's like to be in the surgery and they can have a fair judgement of the performance of each clinician; and they know the treatment times for each clinician as well, so they can measure one against the other
		DCP1

TABLE 45 Coding frame: DCPs (continued)

Theme	Codes	Example quote
Social/patient acceptability	Compared to general med	Because we have nurse practitioners now in our doctors' surgery that people know they can do some things but they can't do everything doctors do. Most people have to go and see a nurse practitioner. Most people go to the diabetic nurse, they're not bothered about seeing the doctor nurse and then there's the practice nurse. Do you know what I mean? As long as people understand and are told where they're going and what they're going for to it all
	Patient awareness	I think some of the patients may not be aware of exactly what I can do, and perhaps I need to address that part, I think I've got to sign up that's more clearly
		DCPT_2
	Patient acceptability (trust in GDP/NHS)	And maybe they don't understand what the therapist exactly is, but then you come through and you do the work for them, and they know that [dentist's name] who they trust, has passed you so that's okay. But if you ask somebody on the street what's a dental therapist they think that you put teeth on a couch and talk to them  DCP5
	Perceptions of patient awareness/ acceptability (GDP pitch)	When the dentist sends them to us, they explain fully who they're going to see, why they're going to see myself or one of the other girls if it is just for hygiene or the oral health educator or the fluoride girls. They know and they understand that and then that's absolutely fine
		I think they're just saying, you can go to my colleagues and have it done, I don't think they're saying what I am DCPT_4
Beliefs and attitudes to role substitution/ direct access	Benefits to dentist	Before when you're explaining to some of the patients it's a simple appointment, the dentist can concentrate on more complex treatment, so again it's directing the needs to the appropriate place. If it's a simple occlusal filling, I can do that, the dentist doesn't need to do that, so because I've freed up 20–30 minutes for them, they can then be doing something that's a bit more complex
	Perceived benefits to patient/practice	And then they know that their work is going to last longer because their oral hygiene is a lot better, so they know overall as well they're not going to have the same problem constantly because we're hopefully dealing with part of the problem
		DCP2
	Dentists awareness	Education of dentists. Them realising that we are not the enemy. We don't want to be dentists  DCP5
		Who's read the scope of practice? Probably a third of the room put their hand up, and even then it's only been a brief read probably of their own rather than everybody else's
		DCPT_1

**TABLE 46** Coding frame: patients

Theme	Codes	Example quote
Patient awareness of roles	Good knowledge about roles	I'm aware that they do a three year course and that they are qualified to do some fillings – as I understood it, you could correct me if I'm wrong – on minors, on children
Toles		Practice 3, patient 1
	Confused about or uninterested in roles	Probably that I'm sure it would be, like would be qualified to do it, so I don't really understand the difference who's qualified to do a filling and the dentist. What's the difference there?
A ++:+	Danieli ta matianta	Practice 4, patient 5
Attitudes	Benefit to patients (population access)	The hygienist probably saved time for the dentist and that's why they had probably more patients than what they do in todays  Practice 4, patient 5
	Free up GDP time	I personally think if they've seen the dentist and then the next time they say, next time you come in for your check-up it will be a dental hygienist. If there's any problems that they can't deal with, they'll refer you to me. Then that gives the dentist, the dentistry staff, the more professional dentists, time to deal with other more serious items
		Practice 4, patient 3
	Lower-quality care	I think everybody should see a dentist not a therapist and be a second-class citizen really, even if you're on benefits To me a dentist is a dentist, you know; a hygienist therapist will not have the experience or the qualifications of what a dentist will have
		Practice 6, patient 3
	Continuity of care	It's just you go with yeah. I think also seeing the same dentist is they're familiar with if you've got problems or
		Practice 4, patient 5
Acceptability	GDP pitch	A person I trust trusts somebody else, that's what's important  Practice 6, patient 4
	GDP–patient relations	Yeah, I've always found that when I've come here they're very, very patient because I'm not the best person in the world
	t Property	Practice 4, patient 3
	Implicate trust in system/qualifications/ training	They're obviously qualified otherwise the dentists wouldn't be recommending them. It comes with professional practice and you just automatically know you're going to get professional treatment  Practice 4, patient 2
	Untrusting of system/qualifications, etc.	I'd be unsure, to be honest. I'd like to know how much they'd done in the past. How many fillings they had done, the training involved, etc.  Practice 4, patient 4
	Preference for status quo	I think I'd prefer to stick with my dentist that I know Just because I don't like trying new dentists. So when you've found one that you really like, I think it's best to stick with them I think it's just that I'm comfortable with what I know
		Practice 6, patient 1
	Comparison to general medicine to help understanding	You go to see the practice nurse who obviously, you're not going to go to a doctor to have your blood pressure taken. The practising nurse is quite capable of doing that and referring the readings to the doctor if they're not satisfactory, and I expect the same thing of a dental hygienist. If they feel they can't actually carry out the work, they don't want to mess things up or if they've got the confidence or whatever, or the skill, then they refer you to the dentist and you make an appointment to see the dentist for that work to be done
		Practice 4, patient 3
		continued

© Queen's Printer and Controller of HMSO 2016. This work was produced by Brocklehurst et al. under the terms of a commissioning contract issued by the Secretary of State for Health. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

TABLE 46 Coding frame: patients (continued)

Theme	Codes	Example quote
	Past DCP experience	Yeah, I'd rather go to [hygienist] about anything to be honest with you. I mean They've been alright, the dentists in here, but I haven't got the same I haven't got the same, what do you call it reassurance as I have with her  Practice 5, patient 1
	Patient inconvenience	The next one I went to was private and they had a hygienist which I used to use but I've no need to here because I get it done with my check-up. I didn't have any problems at my check-up and [XXXX] does it for me. If I have a problem in between I can come and have one then so it's much better than booking it separately  Practice 4, patient 1
	Patient perception of care quality	Yeah, because I think it's the full treatment and I think in the long run it serves a better cause because as you're getting older your gums need a bit more attention. So I think it's saving you from extra dental work really  Practice 5, patient 3
	Social acceptability	But I just feel like the more it's publicised, the more people get used to the idea, and then they accept the idea
		Practice 4, patient 3

## **Appendix 5** Patient and public involvement

Patient and public involvement (PPI) was an important part of the research process and the research team were keen to have PPI embedded within the main workstreams of the programme.

The grant application had been reviewed by PRIMER (www.population-health.manchester.ac.uk/primer/ about/), which is an experienced group of users and user researchers at the University of Manchester. This led to the addition of interviews with patients to the qualitative workstream. The PRIMER group was also invited onto the Research Steering Group, alongside a patient representative (who was also a coapplicant).

The patient representative reviewed the initial screening questionnaire and the main questionnaire to record the inputs for the DEA and SFM. They also reviewed the questions for the qualitative interviews and wrote the interview schedule for the patients.

The intention was to involve the patient representative in the analysis of the qualitative data but, unfortunately, the patient representative secured new employment outside the research programme, which meant that she had little time to be involved further. As a result, a new patient representative was appointed, but this was after the analysis was complete.

However, the research team had also developed a PPI group to get an idea of the types of issues that arise in role substitution. These views were used alongside the analyses for both the quantitative and the qualitative workstreams to ensure that patient perspective remained central. Key themes that arose from engagement with this group were around the awareness of the roles of members of the dental team, perceived drawbacks and benefits. In addition, social acceptability was key. These became the central elements of the analysis for the patient interviews. The PPI group were also recorded to aid recollection and the key views expressed are summarised below.

I don't think there's much difference. She seems to be more, how can I say, logical in the way she expresses what she's doing. She seems to . . . she tells you what's going to happen and what have you. I mean I know Sam does it the same, but she seems to be more, how can I say, knowledgeable in what she was doing.

You don't realise that they're capable to do the same job. But at the end of the day if they can do it then it's taking work off the dentist, obviously they'll see more patients, isn't it, if both of them can do it.

I'm happy to see the therapist as well as the dentist. Because I mean obviously they'll probably have the same ideas, both of them.

But my son, he goes to a different dentist but he sees a therapist and hygienist a lot, so I know how good she was with him. So maybe it's part of that. And I know that he doesn't have the dentist doing the job, he has her because she's small, child friendly, etc., talks to him better.

A bit more relaxed, yes, because you think well, this isn't the dentist who's going to drill. It's a bit more, yeah, at ease but obviously she's fully capable.

I just think two minds are better than one, aren't they? Just teamwork, isn't it, at the end of the day. I saw it as that really rather than oh, she's not capable.

But it's like I shop at Aldi, but I would the same costs for a dentist or therapist. Whereas someone that might not, just wants to pay top dollar for the dentist, they shop at M&S. Different classes, isn't it, really, how people think and what they pay for.

Yes possibly, I think it's more down . . . for me, it's more down to the fact that I think I would probably expect to pay less if I was getting less of a service, not less of a qualified person but if they're deemed to be qualified enough to do the check-ups and make decisions then I don't see any issue with paying any different.

I imagine the whole reason for this is down to the sort of availability of dentist and the fact that people are struggling on waiting list and if you got a therapist that come through that can do the work then it's going to be easing sort of, you know, the backlog of the patients.

I've seen three different people now on all three times I've come. I don't know who I expected to see really today, but no, it doesn't make me any worse or better. Just as soon as I went in I said look, I'm anxious, I'm pregnant, da-da-da, I'm mad and they were fine with me. So it doesn't make me feel uncomfortable.

I think having treatment from the therapist here in the past, I know that the dentist is coming in . . . where there's been more trickier work, the dentist has been in and just double checked and there has never been any issues. I may have had some reservations maybe before I'd seen the therapist, but have been very happy with.

I think I would probably be just as comfortable, which is probably the big thing really because, like you said, they're not the full dentist, they're a therapist, but having had the experience of having work done by them I more than confident to see them instead of the dentist.

Maybe the impression if they were a hygienist then they're just there to do stuff like plaque removal and the deep clean of people's teeth, whereas if you say a therapist then it covers a bit more of broader scope.

The research team intend to use the patient representative and the PPI group in the dissemination of the results. Presentations and papers produced, as outputs from the research programme, will be reviewed prior to submission. This was the process for the presentation of the initial research findings at the International Association for Dental Research meeting in Boston in 2015 and for the submission of the protocol to *BMC Oral Health*.

The *Plain English summary* was also cowritten by the patient representative and patient-facing summaries will be produced.

# EME HS&DR HTA PGfAR PHR

Part of the NIHR Journals Library www.journalslibrary.nihr.ac.uk

This report presents independent research funded by the National Institute for Health Research (NIHR). The views expressed are those of the author(s) and not necessarily those of the NHS, the NIHR or the Department of Health